

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Obling, Anne Roelsgaard

Doctoral Thesis

Management of Emotions in Accelerated Medical Relationships

PhD Series, No. 8.2012

Provided in Cooperation with:

Copenhagen Business School (CBS)

Suggested Citation: Obling, Anne Roelsgaard (2012): Management of Emotions in Accelerated Medical Relationships, PhD Series, No. 8.2012, ISBN 9788792842435, Copenhagen Business School (CBS), Frederiksberg, https://hdl.handle.net/10398/8419

This Version is available at: https://hdl.handle.net/10419/208805

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



NC ND https://creativecommons.org/licenses/by-nc-nd/3.0/



COPENHAGEN BUSINESS SCHOOL

HANDELSHØJSKOLEN

SOLBJERG PLADS 3 DK-2000 FREDERIKSBERG

DANMARK

www.cbs.dk

Copenhagen Business School HANDELSHØISKOLEN

Management of Emotions in Accelerated Medical Relationships

Anne Roelsgaard Obling

Management of Emotions in Accelerated Medical Relationships

Doctoral School of Organisation and Management Studies

Management of Emotions in Accelerated Medical Relationships

Anne Roelsgaard Obling

Copenhagen Business School

Doctoral School of Organisation and Management Studies

February 2012

Anne Roelsgaard Obling Management of Emotions in Accelerated Medical Relationships

1st edition 2012 PhD Series 8.2012

© The Author

ISSN 0906-6934

Print ISBN: 978-87-92842-42-8 Online ISBN: 978-87-92842-43-5

The Doctoral School of Organisation and Management Studies (OMS) is an interdisciplinary research environment at Copenhagen Business School for PhD students working on theoretical and empirical themes related to the organisation and management of private, public and voluntary organizations.

All rights reserved.

No parts of this book may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or by any information storage or retrieval system, without permission in writing from the publisher.

CONTENTS

Acknowledgments	6
Preface	8
Chapter 1: Introduction	10
Why emotions and why now?	13
The importance of analysing practice	17
Structure of the thesis	19
Notes on the format of the analyses and their audiences	23
Chapter 2: Organising cancer illnesses in the Danish health care system	25
Introduction	25
From melancholy to present forms of organising cancer illnesses and their treatment	26
Introducing the terms 'packaged care' and 'accelerated cancer pathways'	29
Introducing the 'soft' dimensions of accelerated cancer pathways	34
Chapter 3: Theories about emotions in organisations	38
Introduction	38
Emotions and epochal battles in social sciences	39
Weber and emotions in bureaucracy	42
Parsons and the conduct code of 'affective neutrality'	45
Elias and historical processes of 'social restraint towards self-restraint'	48

Goffman and the staging of a cynical performer
Hochschild and the search for an unmanaged heart
Strauss and invisible, informal emotion work
Literature on emotions in health care
Bridging the divide: Emotion and rationality in health care
Working within the divide: The colonising of 'sincere' emotions in health care 73
A combined theoretical framework to the study of emotions in organisations 77
Chapter 4: To study emotions - Focusing, gathering and writing up material on
emotions
Introduction
The situatedness of the PhD and my access to a cancer clinic and its occupants 83
Defining my object of inquiry: From foci to a developing focus on emotions 87
Gathering material on emotions
Problems of involvement and detachment in the study of emotions
Writing on emotions
Chapter 5: The 'compassionate' doctor - Emotional injunctions to medical
professionals in accelerated medicine
Introduction
The approach to the management of emotions in work and organisational settings
Method
Accelerated cancer pathways and the conceptualisation of the 'stand up patient' 114

Emotional injunctions to doctors in accelerated cancer pathways116
Individual professionals and 'one shot available' for managing patients' emotions
Discussion
Concluding thoughts
Chapter 6: Training of controlled empathy in accelerated cancer care
Introduction
Theoretical framework
Material and methods
A presentation of a training workshop
The situatedness of the workshop
Invitation to join the workshop
The training sessions
Staging the role-playing scenarios
First stage performance: A patient who is emotionally cold and distant 145
Second stage performance: The relative who is too emotionally involved 148
Discussion
Conclusion
Chapter 7: Doctors' emotional experience and challenges in accelerated medical
work
Introduction 157

	Theoretical framework	160
	Method	163
	Procedure	163
	Emotions in accelerated cancer care	165
	The professional role of a doctor	165
	Emotional experience and challenges	168
	Techniques to manage the emotions	171
	Emotions in standardised work procedures	174
	Conclusion	177
	Chapter 8: The negotiation of the sick role - general practitioners'	
	classification of patients with medically unexplained symptoms	181
	Introduction	181
	Classification of illness	184
	Theoretical framework	186
	Data and methods	188
	The sick role and social problems	191
	The sick role and problematic personality	195
	Discussion	197
(Chapter 9: Conclusion	202
	Theoretical intent and contribution	204
	Practical concerns	211

Future research	213
English summary	216
Resume af afhandling	222
Appendix A: Interviews in the cancer clinic and its surroundings	227
Appendix B: Observations in the cancer clinic and its surroundings	228
References	229

Acknowledgments

Most theses owe their existence to a lot of different people. This thesis is one of them.

I would like to thank the Centre of Head and Orthopaedics, Rigshospitalet, Copenhagen University Hospital, for the financial support that enabled me to begin this Ph.D. project. Special thanks to Tom Pedersen and Ester Lind to whom I owe a great deal of my knowledge about management and administration of public hospitals. Many years ago, Pedersen introduced me to evidence based medicine and its mediation in clinical practice; a topic we since have discussed at lengths.

Many thanks to the cancer clinic and its employees for providing me access to their everyday activities. Without them, this thesis could not have been written. I am especially grateful to the doctors for teaching me about clinical medicine and cancer illnesses, and for answering difficult questions about what a doctor 'feels'. Especially thanks to Niels Kroman for many helpful discussions.

I thank wholeheartedly Monica Greco for allowing me the opportunity to spend a longer visit in spring 2010 as research fellow at the Department of Sociology, Goldsmiths, University of London. She compelled me to think thoughtfully about what I was working on and more specifically she pointed me in the direction of rereading classic sociologists with the aim to understand 'questions of emotion' in specific socio-historical contexts. Greco must also be thanked for reading larger parts of the thesis and offering sharp criticisms.

I am also deeply grateful to my supervisor Signe Vikkelsø. Through the process of writing this thesis, she became my personal 'Howard Becker'. She has taught me to write like a professional social scientist and she has helped me develop sensitivity towards academic writing style. She also took her own role very seriously and helped me re-writing drafts of the thesis. Through the years, I have learned to

follow her advices. Her critical insights have been an invaluable help in the often difficult process of writing this thesis.

I owe a particular debt of gratitude to Nanna Mik- Meyer who plays an important role in my process of becoming an academic kind of person. She has often found my ways of doing things hopelessly amateurish, but has taught me a few tricks on how to balance work tasks and deadlines.

Thanks to the people at the Department of Organisation, Copenhagen Business School, who have contributed to my work. Especially thanks to the members of 'Center for Health Management' for providing many helpful discussions.

I also want to thank friends who have spent time listening to many confusing moments and thoughts. I would especially like to thank Anja Svejgaard Pors, Kirstine Zinck Pedersen and Morten Wilhelm Scholz. Because of them, the task of writing this thesis has not only been hard work but also good fun and enjoyable conversations.

Special thanks to Line Hasle who has proof-read the manuscript and suggested changes to improve the finished work. Also thanks to Cecilie Roelsgaard Obling who has paid attention to references and reference style throughout the thesis chapters.

Preface

This thesis is the result of an ethnographic fieldwork at a major university hospital in Denmark that I undertook between June 2009 and January 2011. I was an 'embedded' observer in a cancer clinic and entirely dependent on the staff – administrative and clinical – for access to facilities, people and diseases. That said, I was never asked to modify my writings in any way or to show the content of my field notes or tape recordings. Neither does the hospital have any formal share in the overall thesis. The responsibility for the final outcome is on my shoulders alone. As an embedded observer I was to handle personally sensitive data, such as specific details in patient records, with confidentiality. There is no information in my writings which can be traced – directly or indirectly – back to individual patients or relatives at the hospital and hence disclose their identity. My observations lasted anywhere from 20 minutes (the length of a typical staff meeting) to five working days in a row. During a day of observation, I followed doctors from they arrived in the early mornings; when they attended the morning conferences, until they left the hospital in the late afternoon after hours of clinical work in the outpatient clinic. I also followed them in their offices and in the operation theatres. Many tableaux from the thesis you are reading now were recorded in my notebook and then reconstructed in the later writing. Wherever possible, I have used my free access to the hospital to check the accuracy of my writing, for example by procuring typical situations more than once or by going through precarious details with involved staff members. Statements that appear in quotation marks ('...') were recorded directly on my tape recorder or in my notebook while the person was speaking, or immediately hereafter. Through the process I have shared my ideas with the staff members involved to make sure that they understood the purpose of my work and also in order for them to have a chance to feel comfortable with my presence. Throughout the thesis, I have shortened quotes from documents and interviews in order to make the text more readable.

In addition to my fieldwork at the hospital, I have worked with the sociologist Nanna Mik-Meyer. In her work, Mik-Meyer has focused on general practitioners and their preoccupation with patients who attend the consultancy with medically unexplained symptoms. Parts of the raw data material from some of her previous studies became the basis of a co-authored article, which is included in this thesis. Utterances from individuals described in this article are directly quoted from a larger quantity of interviews with general practitioners in primary care medicine.

Chapter 1: Introduction

'Continuous reflection, foresight, and calculation, self-control, precise and articulate regulation of one's own affects, knowledge of the whole terrain, human and non-human, in which one acts, become more and more indispensable preconditions of social success'

(Norbert Elias, The Civilizing Process)¹

Picture the scene. It is eight o'clock Thursday morning and the professional residents of a breast cancer clinic are gathered in the staff room. The setting is a prototypical meeting room for staff members; it can be located on every floor and in every department at the hospital. The room is narrow and opens up in the back towards windows with a vista to a large green leisure area flanking the side of the hospital. The meeting table is ellipse-shaped solid Scandinavian design, white as the walls and the colour of the staff members' crisp uniforms. Every second Thursday, doctors, nurses, students, patient pathway coordinator, secretaries and heads of department meet in the conference room for an interdisciplinary staff meeting. The agenda of the meetings vary between introduction of new treatment methods, changes in IT systems, rescheduling of tables in the operation theatres, problems with documentation of certain performances – for example nutrition screenings of hospitalised patients, presentation of new research and so on. Today, the issue to be presented and discussed in the plenum is a new model program developed by some of the nurses in the clinic. The model program addresses different work tasks and focus areas of nursing in a typical breast cancer patient pathway. In other words, it addresses what the nurses are supposed to do in different work phases of the pathway.

¹ Elias, N. (2000 [1939]) The Civilizing Process, Malden, MA: Blackwell Publishing, p. 398.

The staff room is already crowded because on Thursdays, as every other weekday, the doctors — including head of department, senior consultants, junior doctors and medicine students — begin their workday with mandatory participation in a morning conference, scheduled to start at a quarter to eight. The doctors occupy all the chairs around the table and the only chairs available for the rest of the staff are chairs located against the wall or near the very entrance of the room — with very limited view of the event. Hence, half of the attending staff participates in the meeting standing on their feet.

One of the nurses raises her voice and calls for attention. Prior to this, she has connected a laptop to a projector with help from one of the juniors. She is ready for her presentation of the model program, which she now introduces in front of a white-board:

'We [the nurses in the clinic] have now rethought and developed our model program for patients diagnosed with tumour mammae [breast cancer] so it fits with how we [the clinic] have reorganised our accelerated, optimised patient pathways'

She shows her colleagues a PowerPoint slide of the detailed nursery tasks which are to be conducted when a patient visits the clinic after surgery to be informed by a doctor of the results of biopsies of her breast(s) and prospective post-surgery treatment. The PowerPoint slide is a one-to-one copy of a page in the model program. The page is divided into two columns: one named DOCTOR in capital letters and one named NURSE also in capital letters. The column of nursing is ten times the size of the column of doctoring. The few tasks described in the DOCTOR column mainly point out what kind of information the doctor must hand over to the patient, for example it says, 'Providing information about definitive microscopic analysis answers' and,

'Providing information about possible adjuvant treatment'. The written tasks of the nurses are different from those of the doctors. They are described as 'showing care and psychological support and being sensitive to reactions/frustration and taking action upon those feelings' and, 'talk with the patient about expressing thoughts and feelings in regard to surgery result/new surgery'. The presenting nurse patiently works her way down the two columns of specialised work tasks; reading out loud every written point on the slide and going through the material as going through a grocery list on a refrigerator door.

A senior doctor starts giggling and the audience of the meeting turns around to face her. She explains herself:

'I find it rather funny that this model program shows an almost urgent need to record that you nurses constantly conduct care work and psychosocial support to the patients. It's almost like it isn't one of the most profound tasks of doctoring to practice care and empathy in our everyday work. Perhaps, compared to you, we just don't need to record it every time we truly engage in care work. Don't you think that we [the doctors in the clinic] are empathic towards our patients? Don't we display empathic responses? Or is the problem that we, according to your standards, are not empathic enough in the multiple encounters we have with our patients?'

The senior doctor continues shaking her head while commenting on the PowerPoint slides, asking questions to the progressing presentation, while the nurse in front of the white board starts looking pretty annoyed and slightly resigned, as if she knew it would come to this. She has lost the attention of the audience and during the rest of the presentation, the doctors discuss in smaller groups how absurd it is for the nurses to proclaim the right to do 'sincere' care work and record proceeding acts of empa-

thy and emotions, while taking care of patients. The model program, which, apart from scheduling and outlining tasks of nursing in a typical breast cancer patient track, also addresses tasks of empathy work, is nevertheless fully demonstrated and later backed up by heads of department as a piece of 'solid documentation' of the accomplished work carried out in the clinic.

Why emotions and why now?

This thesis is about 'empathy', cancer illnesses, doctors and a increasingly persistent request for emotions in accelerated medical relationships. It is concerned with exploring the relationship between emotions in a cancer clinic and forms of professional conduct therein. A recurring line of argument in the thesis is that 'questions of emotion' offer ways through which we may come to understand how activities of individual human beings are interlinked with forms of social life. This means ways to consider the active, shaping influences between individuals and social, medico-scientific and political worlds. Furthermore, the attention to these questions – or more specifically to behaviour, feelings, gestures, impressions, sentiments, manners, affects, expressions, and etcetera, and the codes, norms and values directing and regulating them – can help us understand how individuals emotionally express and experience themselves in specific historical moments, organisational contexts and social situations. The case in question, the cancer clinic and its inhabitants, seems like a particularly apt case for studying empirically these themes, because of the recent introduction of so-called accelerated cancer pathways in the Danish health care system. And why is that?

The treatment of cancer illnesses is increasingly organised in terms of accelerated cancer pathways or 'cancer packages' (The National Board of Health, Denmark, 2008a). These so-called packages are part of a larger strategy to improve the overall health outcome by offering patients fast, effective and efficient care. The packages are characterised by streamlined patient flows; extremely standardised ac-

tivities; better techniques of micro-surgery and medication; early hospital discharge of patients; a growth of registration and documentation practices; a renewed focus on patient information and communication and implementation of waiting time policies. In addition, I will argue, they entail an increased regulation and training of the 'soft' side of the professional-patient relationship, e.g. 'empathic communication' and 'interpersonal respect'.

In an atmosphere of rationalised and standardised medical work processes and performance measurement technologies, I want to suggest that the development of cancer treatment regimes has created some challenges in relation to the clinical encounter and questions of emotion therein. Moreover, I want to argue that the reason for this lies at the heart of what is known as contemporary modes of governance, where key characteristics are distributed accountability, audibility and efficiency (Power, 1997; Miller and Rose, 2008). This, I want to argue through the thesis, is inextricably linked with concerns of framing the direction and content of the clinical encounter, and hence, concerns about regulating the emotional engagement between doctors and patients.

Medicine is generally a field in which emotions are contested, negotiated, managed and produced, and the importance of emotions in health care has long been recognised by medical practitioners. However, its role in accelerated medical relationships is less straightforward. A natural consequence of today's attempts of rationalising medical practices may be conceived as pushing doctors in a direction of less emotional engagement and more objective detachment from the clinical encounter. The rearrangements of the pathways leave little room for the provision of emotion management, and often leave us with the impression of a 'cold' doctor. What patients yearn for is empathic, engaged and responsive doctors who understand and attend to their personal needs (see, for example, The Danish Survey of Patient Experiences, 2010). Patients, relatives and lay people in general complain about a lack of 'empa-

thy', 'responsiveness' and 'compassion' in clinical encounters. They request engagement, communication and extra-personal attention and empathy from the practitioners. This call for more 'warmth' in medicine has not been overheard. Both medical practitioners and scholars within the disciplines of medical humanities or narrative medicine agree that there is a need for recognising and offering 'psychosocial support' and 'empathy'. Notions of the 'empathic healer' (Bennett, 2001); the 'empathic practitioner' (More and Milligan, 1994) or the 'compassionate practitioner' (Gilbert, 2009) have emerged. Consequently, medical professionals are called upon – not only to be efficient and effective and to cure diseases and save lives, but also to perform 'authentic' or 'deeply felt' emotional engagement in interactions with patients

Perhaps symptomatic, political debates regarding the rearrangements of cancer illnesses and their treatment mainly focus on the managerial aspects of streamlining the pathways such as the coordination of diagnostics and treatment processes across occupations, units, hospitals and sectors. One example is 'Governance of cancer pathways' (Dansk Sundhedsinstitut, 2008), which focus on the importance of 'well-defined standards for appropriate waiting times and quality; systematic monitoring and consequences of dismissing targets' (2008: 5). However, as is evident from the complaints from patients and practitioners, there is a concurrent need for better managing also the implications of the rationalisation and standardisation of clinical pathways on the way patients and professionals feel and cope during the treatment processes. I conceptualise this as a call for management of emotions in doctor-patient relationships. So far, this theme has predominantly been addressed in relation to the psychological and social comfort work done by nurses. Nursing practice is a textbook example of how professionals manage emotions in health care organisations, and, as a particular field of study, this has been covered at lengths. Far less so is this the case when we focus on doctors. Therefore, the focus of this thesis is to zoom

in on emotions in medical care, especially in relation to the conduct of doctors in a contemporary organisational framing of less time and calls for more empathy.

What is striking in the previously described morning meeting in the cancer clinic is the straightforward articulation of emotions in a context of a presentation of the recently rearranged pathways for cancer patients. Concurrently with the rationalisation of the pathways, which is manifest in the presented model program, the emotional aspects of the pathways emerge in the contexts as significant objects of interest. In the pictured example, we see how the presenting nurse tries to purify the conception of empathy, understood as empathic responses and empathic behaviour, from other work tasks in clinical practice. She furthermore monopolises it as a property of the nursing profession. However, her presentation is aggressively obstructed by a doctor who claims that empathy is already comprised in medical care. The doctor calls empathy a 'profound task of doctoring' and she claims that empathy cannot be separated from other skilful competences of a doctor, such as clinical reasoning or decision-making. The situation turns our attention to the possible connections between forms of empathy, rationalisation efforts – such as attempts of speed and standardisation – divisions of labour and professional conduct.

In trying to address this relationship, I use a simple analytical principle as my guide: to avoid the injunction of a priori distinctions between 'emotion' and 'reason'; 'emotionality' and 'rationality'; 'passion' and 'logic', and so forth. In the history of Western thought there is a long tradition of separating emotion from reason, seeing them as opposites and as different kinds. Emotions are often taken into account as something irrational that is perverting or disrupting reason. A classic example is Descartes' placing of emotions in the corrupt body and rationality or reason at the centre of human identity. It is the emotions, passions and sentiments that are 'the harm of this life', because they are not an integral part of the mind and cannot in any way be absorbed into strict science (Rorty, 1986: 533). However, as argued by Jaggar (1989), the relationship between emotion and rationality has never been categorical

and the split between them not absolute. The metaphor of a charioteer steering a wild horse, originated by Plato in *Phaedrus*, is an enduring characterisation of this relationship between emotions and reason. Reason is the charioteer; emotions are the horses he tries to steer. In this model, emotions must be managed and directed by reason. This is not the same as stating that emotions need to be suppressed. Only an idiot would not express anxiety in precarious, frightening situations. However, even lifethreatening situations are thought possible to stabilise through the channelling of appropriate emotions. Here, according to the interpretations of Plato, emotions have a function in the establishment of social order, which is also captured in the metaphor of a charioteer and his horses: imagine the function of the former without the powers of the later

To avoid taking over the 'work of purification' (Latour, 1993: 31) in this study of emotions and, for instance, cementing problematic distinctions between emotion and reason, I will instead try to explore how such distinctions arise and become stable in various organisational contexts. Adopting such an approach also implies that it becomes impossible to separate emotion and reason from the specific historical moments and social situations in which they emerge. Hence, attention is paid to techniques and practices of individuals and how individuals use forms of emotionality, in what situations and for what purposes, instead of focusing on some restricted dichotomies that circumscribe how emotion or rationality might be described and understood on an abstract, metaphysical level.

This theoretical attention leads to an interest in exploring the relationship between emotions in a cancer clinic and forms of professional conduct therein. More specifically, I centre on the following research question: *how is emotion framed, trained and performed by doctors in accelerated medical relationships?*

The importance of analysing practice

The thesis addresses the everyday work of doctors in a cancer clinic. What I will present is an account of how doctors perform medical care in a tense climate of rationali-

sation and standardisation processes at a public hospital in Denmark. I argue that health care organisations are governed as profound territories of different and diverse concerns, and, in addition to this, I intend to show how changes of organisational contexts have massive implications for both professionals and patients — with an empirical focus on the former. My concern is political in the way that it addresses the concerns of practitioners in a concrete work setting. By political, I refer to an argument put forth by Clegg. He writes: 'there is an ethical dimension to [...] organisation studies [...] the organisation analyst has a responsibility towards the subjects of that science. When we investigate organisations we also address the impact of major structures of society on the lives of ordinary people' (Clegg, 2002: xxvi).

The thesis focuses on the changing conditions under which doctors are made to work. As the analyses, developed in later chapters of this thesis, will show, traditional ideals of medicine (such as objectivity and affective neutrality) will be called into question when attempts of transforming the health care sector anew necessarily include its professions and the conduct of its practitioners. As such, the thesis must be considered a work of critique, even though the criticism is not directed towards the medical profession per se. Instead, the thesis intends to lend support to clinical medicine. In her ethnography of atherosclerosis, Mol (2002) highlights in a similar way that her work 'lends support to clinical medicine' (2002: 183). She exemplifies this attitude with the statement that 'surgeons only open up bodies if they think they can gain a curing result by doing it' (ibid). The position of Mol is nonsentimental because, in her work, doctors' conduct is investigated by looking at the practices – or in Mol's own terms – the enactments constituting the medical activities in questions. Inspired by this attitude, I intend to go along with the key objective of doctors in the cancer clinic, which is to diagnose and prospectively cure people suffering from malign cancer illnesses, and engage directly with the individuals who undertake important and often lifesaving medical care. Hence, it may be possible to

support the ideals of clinical medicine and address the critique towards the changing techniques and technologies of contemporary health care practice.

I have no intention to produce a comprehensive account of medicine in general; that is, of how doctors think about and carry out medicine. Rather, I concentrate on just one element of medical care, namely how emotion is framed, trained and performed in a cancer clinic. The academic interest in emotion and the management of emotions is by no means new. Neither is the interest in the issue set in health care. There is a large and ever-growing literature on emotions and health and emotions and organisations — also literature that use hospital settings as research sites. Nevertheless, today's strong focus on effectiveness and efficiency in Western health care services and a pressure on public organisations, such as hospitals, to rationalise, evidence base and optimise their performances, reinstate the academic interest and posses new questions concerning emotion. And yet, among the things I hope to bring to this rising interest in emotion — in healthcare as well as in organisation studies — is a caution against sustaining the dichotomies being produced in, or as a response to, such rationalisation attempts. Instead, I will argue, there is need for empirical studies of the way emotions and rationalisation are co-developing in different forms.

Structure of the thesis

The questions of emotion are guiding the thesis and framing the individual chapters. The thesis is divided into two parts – a frame (Chapter 1-4 + 9) and four articles (Chapter 5-8). The frame begins with a description of the organisation of cancer illnesses and their treatment in the Danish health care system. Accordingly, Chapter 2 introduces the concept of 'accelerated medicine' and it explains the development of the so-called 'cancer treatment packages'. It further introduces the issue of emotions in accelerated cancer care and it pays attention to how questions of emotion arise from the empirical context.

Moving on from here, Chapter 3 deals with theories of emotions in organisations. In this chapter both classic sociological literature and more recent theories of emotions, social order and organisations are addressed. In addition to this, I review texts that focus especially on emotions in health care practice and discuss how some of these texts fall short in their approach to emotions because they tend to elevate emotions to the disregard of rationality and rationalistic activities. The chapter then moves towards a preliminary draft of my own theoretical approach, which is further developed in the remaining chapters of the thesis.

Chapter 4 explains a number of methodological challenges which accompany my qualitative study of emotions in a cancer clinic and its surroundings. It describes the situatedness of my PhD project and my access to the clinic and its occupants, and it explains how my object of inquiry developed over time – from foci to a developing focus on emotions. It also contains a relatively clean description of the conducted fieldwork; of the qualitative methods and the amount of material used for this research, and it describes how each analysis in the thesis takes smaller parts of the empirical material as its point of departure. Finally, it discusses problems of 'involvement' and 'detachment' (Elias, 1987) in the study of emotions and it pays attention to a kind of 'methodological vocation' which is described as the vocation of researchers to approach emotions without sentimentality.

In summary, Chapter 2-4 move steadily towards the analyses of the relationship between emotions in a cancer clinic and forms of professional conduct therein. The analyses are structured into the headlines 'framing', 'training' and 'performance' of emotions in order to analyse in detail the frames, techniques and practices which direct and regulate the conduct of doctors in everyday, accelerated medical relationships. I will now briefly address the aim of the four analyses. Afterwards, I will make a few comments on the relationship between them.

Chapter 5 explores how emotions are framed in medical relationships and it addresses how this framing equips doctors to act in certain ways in particular situations. Using the development of accelerated cancer pathways and a wide range of health care reform documents as an empirical case, the chapter explores how doctors are encouraged to become more emotionally available to patients. As the chapter demonstrates, this call for emotionality is accompanied by increasing demands on doctors to manage the emotions of others in recordable, measurable and standardised ways. To be a compassionate doctor involves an exhibition of a particular emotional behaviour in medical relationships, what Elias calls a 'controlled de-controlling of emotions' (Elias and Dunning, 1986: 44), namely that of a 'sincere' or 'authentic' behaviour which doctors are required to enact and to be committed towards. What this points to, I argue, is a set-up where health care reform documents encourage room for personal and unique concerns in medical relationships while expecting maximum acceleration and standardisation of these relationships.

Chapter 6 continues the investigation of questions of emotions through a discussion of how empathy and responsiveness, as specific techniques of emotion management, arise as the outcome of well-structured forms of training and practice in the cancer clinic. The chapter focuses on a training workshop in 'empathic communication' through which doctors from the clinic learn to recognise and control the emotional frame of doctor-patient interactions. It addresses how the training and practice of communicating empathically in these interactions rely on standardised scripts, which in turn direct and cultivate the conduct of the doctors involved. In line with the argument in the previous chapter, it concludes that humanistic values increasingly become the target for techniques of micro-management such as qualitative measurement and performance audit. In other words, the chapter argues that attempts to improve 'soft' dimensions of medical services entail a further standardisation of these aspects.

Chapter 7 addresses the question of how doctors relate emotions to their understanding of professionalism and principles of standardisation and speed in the treatment of cancer illnesses. In a present work environment of reforming and rationalisation drives, the chapter directly asks 'how doctors feel'. This question is explored through doctors' personal biographies of emotional experience and challenges in relation to their routine and everyday situations. The chapter explains how emotions and emotional display are frequently performed in a rationalised way to help doctors ensure the progress and efficiency of the accelerated treatment regimes. However, the chapter scrutinises how the rearrangements of cancer illnesses and their treatment into 'accelerated packages', and attempts to separate social and psychological dimensions of medical services from technical dimensions of these services, complicate doctors' ability to incorporate emotions into their practices in a 'proper' way. What this points to, is the need for researchers to investigate how different forms of organising technical tasks within health care, entail different ways of organising emotions and emotional practices within this field, and how this generates organisational problems of all kinds for both patients and health care workers.

Chapter 8 completes the shift from the 'framing' and 'training' of emotions to concrete 'performances' of emotions in medical relationships. It makes this shift in a relatively radical way because it addresses questions of emotion through a study of how general practitioners (GPs) approach patients with medically unexplained symptoms (MUS) in primary care. Concerns about the role and function of emotions in the core of the medical enterprise come under closer scrutiny here, because the chapter explores how GPs are diagnosing with feelings. Accordingly, the chapter investigates how GPs diagnose MUS patients as legitimately sick patients. What makes this classification procedure difficult is that MUS patients fall inside a particular residual category in the eyes of the doctors. This defies the doctors' capacity to decipher the patients' intentions and motives, making it hard to empathise with them. To confirm

the subjective complaints of patients, the GPs must rely on their personal opinions and evaluations of a patient's particular circumstances when deciding whether the patient is legitimately sick or not. These different strategies for managing the complaints of patients are connected to ways of dividing individuals into certain categories of persons characterised by deviant features such as unpleasant personalities and manipulative appearances. The chapter emphasises how the affective connection between the patient and the doctor becomes paramount to the outcome – the diagnosis – of the clinical encounter.

This occupation with frames, techniques and practices through which doctors are equipped with the capacity to act as certain sorts of people, and how this particular performance takes form in practice, is manifest throughout the chapters. The connection between the 'framing', 'training' and 'performance' of emotions in medical relationships is investigated in detailed empirical analyses of how notions of 'compassion' and 'authenticity' emerge in the heart of accelerated, standardised medical services, and moreover, how doctors get to know about and manage peoples' feelings and make themselves emotionally available to others in social interaction.

Notes on the format of the analyses and their audiences

The PhD thesis is an article-based thesis. This means that it consists of a 'frame' and four separate articles. The articles appear in different stages 'outside' of this thesis: A translated and shortened version of Chapter 5 'The compassionate doctor' has been accepted for publication in Järvinen, M. and Mik-Meyer, N. (eds) *What makes a professional*? (In Danish: Hvad skaber en professionel?'), Copenhagen, Hans Reitzels Forlag (June, 2012).

Chapter 6 'Training of controlled empathy' has been accepted for publication in a special issue of the Danish journal, *Forskning i Sygdom og Samfund*, (September, 2012).

Chapter 7 'Doctors' emotional experience and challenges' has been accepted for publication in *Journal of Health Organization and Management*.

Chapter 8 'The negotiation of the sick role' has been accepted for publication in *Sociology of Health and Illness*, Vol. 34 (7), (September, 2012).

Chapter 2: Organising cancer illnesses in the Danish health care system

Introduction

A cancer clinic at one of the most specialised university hospitals in Denmark makes up the empirical foundation of this inquiry. The clinic is a surgical unit, specialised in breast cancer diagnostic and treatment. It is located on the 10th floor of a tall building in the centre of Copenhagen. The clinic comprises consultancy rooms, ward, doctors' and nurses' rooms, common staff meeting rooms, cleaning rooms, kitchen, storing rooms and so on. The operation theatre is located on the third floor and so are most of the doctors' offices. The clinic is staffed by 15 doctors, including senior consultants, assistants and juniors. In addition to this, nurses, secretaries and coordinating staff are employed in the clinic. Recently, the clinic has been merged with another surgical unit and is now part of a huge clinic with newly appointed heads of department. One of the senior consultants, interviewed as part of this research, described this merging as being 'a matter of administrative cuts'. He furthermore explained that 'the merging with the other clinical unit has yet no experienced effects when it comes to our daily work, namely the diagnostic and treatment of women with breast cancer'. His words justify why I have chosen to describe the cancer clinic as an autonomic clinic throughout the thesis, hereby leaving the merging of the two clinics aside when it comes to empirical attention.

The cancer clinic is a typical example of a sub-specialised unit where most of the patients are diagnosed with the same illness, namely that of breast cancer. Annually approximately 12.500 (2009 numbers) patients attend the clinic, out of which 860 are diagnosed with cancer and thus undergo surgical treatment in the clinic. The homogeneity of the clinic's patient sample – in regard to its unifying illness label – makes it possible to describe the patient trajectories of the clinic in terms of one single standardised patient trajectory, which every patient attending the clinic goes through. I term this patient trajectory 'accelerated cancer pathway' and in the

next section I will describe in detail what the term comprises. The aim of the description is not only to give the readers a general familiarity with the term, but also to set down some of the basic premises on which the thesis' arguments rest. By talking about these premises, the reader may come to think of the subject matter presented here as a mapping of current forms of organising cancer illnesses and their treatment. This mapping is only partial but nevertheless adequate for the purpose. It provides a backdrop for my selection of facts, people and stories from the cancer field and helps to define the purposes that the rest of the thesis will have to satisfy.

From melancholy to present forms of organising cancer illnesses and their treatment

A very brief medical history of breast cancer illness goes back to Claudius Galen, a Greek doctor and biologist living in the second century who believed that cancer in the breast region was caused by excess of black bile, which is the Greek word for melancholia. Galen's observation inserts an early historical interest in the relationship between emotion and cancer, because melancholy was used to label someone who was both depressed and emotionally reserved and non-assertive (Giese-Davis and Speigel, 2003). The therapeutic consequence of this belief was that patients were offered purgation and bloodletting services. Those patients who died because of the treatment were either patients who were not sufficiently convinced in Galen's methods or patients who lacked the constitutional vigour to sustain prolonged bloodletting (Baum, 2007).

It was not until the second half of the nineteenth century, that breast cancer became accepted as a disease of cellular pathology within the breast and it was recognised as a disease which spread along the lymphatic system to other areas of the body. From 1894 up to the last half of the previous century there was only one standard procedure in the treatment of breast cancer: the routine use of radical mastectomy (removal of the breast). The treatment can be traced back to Dr. William Stuart

Halsted, who was professor in surgery in the beginning of the 20th century. Halsted's hypothesis was that breast cancer is a locoregional disease which spreads from its origin. The way to cure cancer patients was to eradicate the localised disease. In the 1950's, the mastectomy method was supplemented by radiotherapy and it was now possible to perform a less mutilating mastectomy. In the beginning of the 1970's, adjuvant treatment (medical oncology) was developed. Around the same time, research made it possible to divide patients into high-risk or low-risk groups based on the number of positive lymph nodes found in the patient's body. These developments changed the treatment of breast cancer into a systematic approach where treatment was directed towards the spread of the disease. During the last two to three decades new medical technologies have been developed such as steadily more intensive adjuvant treatment; sentinel lymph nodes biopsies and minimally invasive surgery. Generally, more gentle treatment methods have thus been invented to both meet increased demands for quality and to make interventions safer (Kehlet & Wilmore, 2002).

The Danish Breast Cancer Cooperation Group (DBCG) was established in 1977. The objective of the DBCG project was built on an aphorism by the famous surgeon – and founder of the Mayo Clinic – Dr. Charles H. Mayo, whom in 1916 proclaimed: 'The keynote of progress in the 20th century is system and organisation' (Blichert-Toft et al., 2008). The purpose of the DBCG was to standardise the treatment of breast cancer with reference to novel therapeutic principles of diagnostic and treatment. The establishment of the group changed the ways in which surgical and non-surgical partners collaborated in a systematic way. As a multidisciplinary professional group, DBCG developed – and continues to develop - guidelines for databases, translational research, platforms and other infrastructure (Overgaard, 2008).

The guidelines developed by DBCG follow diverse quality assessment programmes of the Breast International Group (BIG), and are consistent with international standards for breast cancer treatment. Today, nearly all treatment units in Denmark have applied the DBCG guidelines for diagnostic procedures, surgery, ra-

diotherapy, systematic therapy, and follow-up for early-stage breast cancer. The clinical guidelines are used by the doctors in their daily medical work and they guide clinical judgment and medical decision making processes in the treatment of patients. These guidelines became fundamental building blocks for the cancer treatment packages to which I now turn.

In 2000 the clinical focus on cancer illnesses and their treatment was directly coupled to economics, and political and customer concerns. In 2000, The National Board of Health, Denmark (NBH), introduced the national: 'Cancer plan: Status and suggestions for initiatives concerning the treatment of cancer' (NBH, 2000). The cancer plan report was initiated by The Danish Ministry of Health as a consequence of a public debate concerning the overall quality of cancer services in Denmark compared with other Nordic countries. An epidemiologic task force found that Danish cancer patients had a lower change of surviving their cancer illnesses viewed in a five years perspective compared to patients in their neighbouring countries (2000: 37–46). The report concluded, among other things, that new ways of organising cancer treatment – including 'the organisation of work and work methods' – may raise the quality of cancer services and thus lead to a reduction in mortality (2000: 13-18). Furthermore, the report stated that accelerated diagnostic procedures and surgical interventions are significant means to improve these services and thus improve their outcome. The report also recommended the implementation of breast cancer screening programs. Between 2000 and 2005, the screening programs were implemented nation-wide. In addition, in 2001, the Danish Government introduced a 'waiting time guarantee' assuring every patient with suspicion of cancer a right to get diagnosed by a specialist within two weeks. And from the day of diagnosis to the day of treatment, the waiting times were limited to maximum two weeks. The waiting times were politically defined and covered a diverse range of cancer illnesses (also cardiovascular illnesses were covered by the guarantee).

However, the implementation of various means to improve the outcome of the services provided was not without problems. The introduction of politically defined waiting times led to the problem of keeping the waiting time promises. And the screening programs led to an increased amount of women diagnosed with early stages of breast cancer which raised the total healthcare expenditure on cancer treatment services. Also demographic changes, such as an ageing population in need of cancer treatment brought about by increasing life expectancy, had a financial impact on the health care system.

Generally however, these organisational challenges were not met with a huge supply of financial resources to the cancer field. Instead, they were met with concrete attempts of organisational change, as I will turn to next.

Introducing the terms 'packaged care' and 'accelerated cancer pathways'

In 2005, in continuation of the first cancer plan, the NBH published a follow-up plan with the name 'Cancer Plan II'. It specified a new organisation of cancer care in the Danish health system and described how concrete organisational actions were connected to centralised monitoring systems in order to systematise cancer care services across hospitals departments, sectors and individual practitioners. An important aim of this second report was to 'provide a well-organised packaged trajectory to cancer patients on the highest possible international quality level without unnecessary waiting time' (NBH, 2005a: 10). The report understands the term 'well-organised' in relation to activities which are planned and performed by health professionals in a focused manner and with a high degree of application. This entails that necessary resources must be assembled. The report wrote that 'taking departure from a range of well-organised work tasks and patterns of co-operation makes it possible to establish a patient trajectory where the individual patient – without unnecessary delays and efficient allocation of resources – is offered an uninterrupted series of treatment from general practise to involved hospital departments' (2005a: 7). This series of diagnos-

tic and treatment procedures, the report recommends, must be organised in 'packages': standardised, accelerated pathways. In this way, the term 'cancer packages' (in Danish: 'Kræftpakker') is introduced as standardised trajectories to patients who are suffering from cancer illnesses and which ensure that the majority of the patients diagnosed (or with suspicion of a diagnosis) within specific cancer categories (e.g. lung cancer; breast cancer or gastric cancer) can be offered the same 'package' of therapeutic processes, and can expect a minimum of delays at hospitals and across primary and secondary health care sector. The organisational design of 'cancer packages' was intimately associated with another widespread concept of 'accelerated medicine', which also took form in these years. Both concepts emerge from an ambition concerning the establishment of economic, effective and efficient patient trajectories in the health delivery system. The overall focus of the concept of accelerated medicine is an optimisation of the organisation of professional work processes through increased interdisciplinary teamwork, implementation of technological interventions (for example micro-surgery) and formalisation of communication and patient information. Through these means, the concept aims to reduce the patients' length of hospital stay and hence release resources and capacity to be used elsewhere in the public health care sector (The Secretary of Ministers, 2007). I use the term 'accelerated medicine' as an assemblage term to capture core connotations of the concept, such as 'packaged diagnostic and treatment'; 'optimised care'; 'accelerated trajectory', 'fast track surgery' and 'early recovery'. I also intend to use the terms 'accelerated clinical pathways', 'optimised clinical pathways' or 'treatment packages' more or less synonymously through the text.

Under guidance of NBH, task forces were appointed to develop the concept of 'cancer packages' within specific illnesses. In 2008, a 'Breast cancer package' report saw the light of the day (NBH, 2008a). The report described a standard trajectory for breast cancer patients – from they visit the general practitioner in primary care, get diagnosed at the hospital, undergo surgery and adjuvant treatment to control

programs and palliation. Processes of nursing care, communication and patient information are also specified in the material. In the trajectory description, psychosocial support procedures are explained as essential for the outcome of the trajectories. These kinds of procedures are defined in a separate document as 'the health professionals' specific efforts in relation to psychological, emotional and existential dimensions of the entire trajectory' (NBH, 2008c: 5). The efforts are integrated with the clinical actions of the trajectory as concrete work activities which must be performed by the doctors in relation to every patient undergoing diagnostic and treatment in breast cancer clinics. The trajectory description, which determines activities of accelerated clinical cancer pathways, is evidence based, national and nation-wide guidelines and recommendations, or so the report says (NBH, 2008a). In cases where no evidence is available, the trajectory descriptions are based on 'best available practice'. In the particular case of breast cancer illness, it is the previously mentioned DBCG guidelines that support most of the actions in the trajectories. This model explains the treatment phase of the trajectory and its different work activities (clinical, logistic, information, specialty, monitoring):

Clinical action	Logistic ac-	Patient infor-	Medical spe-	Monitor-
	tion	mation	cialty	ing
Pre-examination in the out-patients clinic	Booking: surgery	 Delivery of diagnosis Plan for further action Informed content Summoning: surgery 	Surgeon	Monitoring: in- formed content
Hospitalisation			Surgeon	Monitor- ing: treat- ment start

Surgery			Surgeon	
Histology result			Pathologist	
• Chemotherapy, hormonal treatment and radiotherapy	Booking: consultation in the out-patients clinic	Summoning: Consultation in the outpatient clinic	Multidiscipli- nary team	
Consultation in the out-patient clinic	Booking: • Oncological preexamination • GP: Epicrisis (case summary)		Surgeon	Monitor- ing: surgi- cal treat- ment fin- ished

Model 1: A trajectory model, NBH, 2009a: 22

The rubric 'patient information' describes in bio-medical terms some concerns about the diagnostic moment and judgments about treatment plan. The rubric 'clinical actions' describes the various forms of clinical activities in objective pathological categories.

In the breast cancer trajectory model two formalisation aspects are present: detailed and precise specification of tasks and responsibilities of medical specialties and individual practitioners within an ordered process line of medical work. Every performance of the trajectory is specified and the performances are routinely monitored. Furthermore, it is co-ordinated and integrated via a complex IT-based system that makes it possible to measure the output of the cancer programs on a central-

ised level. This is to measure the development in productivity rates and to measure 'the time span from hospital referral to treatment' (NBH, 2011). The treatment phase of the program and a specification of time standards coupled to the different work tasks can be presented like this:

- 0 Day. Managing patient referral; booking of patient-doctor consultations and examinations
- 3 Day. Preliminary examination (including clinical examination); patient information; patient interview (nurse, anaesthetics, physiotherapist); eventual supplementing picture diagnostic tests
- 6. Day. Patient time to use for reflexion
- 6. Day. Valuation of co-morbidity
- 7. Day. Surgery (hospitalisation and (optimal) discharge of patient)
- 8. Day. Eventual discharge of patient

Model 2: A treatment phase model, NBH, 2009a: 13-15.

This is a standard example of a trajectory in a cancer clinic, running from the moment a patient is examined for breast cancer, receives a cancer diagnosis, undergoes surgery and is discharged from the hospital. This leaves the doctors in the clinic with eight days to make an exact diagnosis; to make preparations; provide emotional or psychosocial support; remove the cancer tumour, stitch up the patient and discharge the person from the clinic. The trajectory description codifies general processes of a breast cancer trajectory, as opposed to clinical guidelines which function more as specific tools or practices in a medical line of work. So-called 'time of sequences' of the trajectories are written in the left column of model 2. These time estimates are

also measuring points from where the clinic is hold responsible for meeting or failing to meet the standards.

The cancer clinic in my study was well-prepared for the introduction of these 'cancer packages'. From programs of accelerated medicine, manifest in the clinic by concepts of fast track surgery and early recovery (Kehlet and Dahl, 2003), the clinic had already developed local initiatives for accelerated procedures which included standardised drug modules (to prevent post-operative pain and nausea), technical interventions (early removal of drainage tube) and changes in patient passages between department units – for example, most patients are removed directly to the clinic ward instead of trespassing through the hospital's recovery ward (Gärtner, 2010; Mertz et al., 2009). Procedures of extended information to patients had also been implemented. The nurses for instance now call the patients after their discharge to make sure that newly operated patients are generally well-functioning according to standards of physical and emotionally recovery.

It is not hard to detect how the cancer clinic and the conducts of its doctors and nurses follow strictly formalised rules in the accelerated cancer pathways. In addition to this well-ordered line of activities, another important dimension in the cancer clinic has been standardised and accelerated: something which in the trajectory descriptions goes under the names 'psychosocial support procedures', 'communication' and 'patient information'. I will now turn to this issue.

Introducing the 'soft' dimensions of accelerated cancer pathways

Simultaneously with the introduction of the two national cancer plans, a report issued by the Danish Ministry of Interior and Health was published under the headline 'The patient's encounter with the health care system' (The Ministry of Health and Interior, 2003). It says in the report that it focuses on 'so-called human relations, which are often referred to as "the softer values" of health care services' (2003: 3). Furthermore, it explains that it aims to improve these soft dimensions of medical care

that hospital departments promote an organisation culture where professionals possess competences to recognise the 'values, feelings, assumptions and thoughts of the patients' (2003: 8). To be able to promote and frame such a culture in the clinic, the professionals need to undergo 'supervision, training and education in a variety of communication techniques' (2003: ibid). The report recommends that professionals 'display sincerity, responsiveness, trust, engagement' in the practical execution of diagnostic and treatment procedures (2003: 9). It states that the individual professional must provide the way for patients to 'display their emotions in a spontaneous way' in their encounters with the health system (ibid). Professional involvement in the provision of cancer services implies 'combinations of emotional and practical conduct' (2003: 32), the report explains.

This kind of professional conduct is likewise promoted in a report from The Danish Cancer Society (KB), simply covering 'The world of the cancer patient' (KB, 2006). Based on an extended patient satisfaction survey, the report articulates cancer patients' need for an improvement of health services: 'Cancer patients far from receive the optimal treatment they need. Especially their needs as individuals and human beings are often let down or left unrecognised' by the health care system (2006: 2). This call for improvement of cancer services thus includes an improvement of the health care workers' psychosocial or therapeutic competences. The workers need to understand the individual patient as 'a unique person instead of merely a number in the line of patients in the trajectory' (2006: 17). As described in the report, one way to meet the patients' needs is through the improvement of communication or communication skills. The report especially appraises 'good communication which includes empathy, respect, responsiveness, and time and engaged presence of professionals' (2006: 16) as a significant element in the organisation of accelerated cancer pathways. Here, communication is understood both as a practical skill – for example in relation to the delivery of information – and as a therapeutic skill which may help

to improve 'cancer patients' psychological, emotional and spiritual health' (2006: 15). Through the medium of communication, doctors can become emotionally available to cancer patients.

The 'National Cancer Plan II' (NBH, 2005a) refers explicitly to the recommendations of these reports, and the above mentioned 'soft' dimensions of medical care are written into the cancer plan's appendix. In this way, the recommendations are also represented in the breast cancer trajectory description under the headlines 'patient information', 'communication' and 'psychosocial support to patients and relatives' (NBH, 2009a). In the description, this inclusion of these dimensions are furthermore emphasised as proper work. The definition suggests that it takes continuous efforts and time to include these dimensions as 'an integrated element of the entire achievement which is – and should be – offered to patients in every phase of the illness trajectory' (2009a: 28).

In the local cancer clinic I have researched, it is widely acknowledged among the doctors and nurses that for instance communication and information play a significant role in the accelerated treatment pathways. These activities are therefore not only evident in the generic trajectory descriptions. They are also written into the clinic's local procedure plans (see for example The Cancer Clinic, 2009). To be able to meet the aims of early recovery and discharge in the pathways, the patients are required to understand the various steps of the trajectory in detail. This includes that they are prepared to make an informal approval of the services offered, to undergo treatment and leave the hospital shortly afterwards. The clinic's procedure plans describe in what way the patients who undergo accelerated procedures of diagnostic and treatment will be prepared by doctors and nurses in the clinic for the coming interventions. This preparation work is explained on a practical level ('inform the patient about diagnosis and treatment') and on an emotional level ('communicate with the patient about the need to express feelings and thought in relation to the cancer illness'). As also emphasised in the trajectory descriptions, the delivery of information

and psycho-social support to patients is monitored alongside, for instance, a recording of clinical activities. The clinic's local monitoring process involves that these elements are written into a patient record system in a standardised language in order to be externally extracted – for example when the clinic receives audit visits from the hospital's department of quality or international accreditation agencies.

In 2012 the cancer clinic and the surrounding hospital will be accredited by the Danish Quality Model (IKAS, 2009). The cancer clinic has been accredited four times since 2000. In relation to previous audit rounds, the delivery of standardised forms of psycho-social support and communication will, according to centrally placed audit officers at the hospital, be a new, important topic of interest for the accreditation agency. Central to this procedure is the quantitative measurement of doctors' manners and behaviour within medical relationships.

The rest of this thesis will mainly be devoted to examining these so-called 'soft' dimensions of medical care. My argument will be that the development of accelerated cancer pathways involves some means of adjusting the conduct of doctors. These means may insert new emotional injunctions to doctors, which involve that certain emotions are explicitly framed, trained, performed and monitored in the clinic for the purpose of improving the overall service outcome of the cancer trajectories.

Chapter 3: Theories about emotions in organisations

Introduction

Emotions in organisations have taken up considerable research interest in recent decades adding new perspectives to the classic sociological literature on the topic. However, a number of analytical challenges seem to accompany this interest. First, there is a lurking tendency to separate emotions as distinct from 'rationality'. Second, there is a related tendency to elevate emotions as more 'authentic' or 'sincere' than rationality. Third, many recent texts are prone to a sentimental tone regarding emotions. I intend to pay attention to these challenges while I read a selection of theories and texts that addresses and debates emotions in organisations.

In this chapter, I will first address both classic literature and more recent theories on emotions, social order and organisations. I will make a short introduction to the study of emotions in social sciences and address an interest in 'questions of emotion' across academic disciplines. I will demonstrate how an interest in emotions and affectivity occupy an important place in Weber's theory of emotions in modern bureaucracies, Parsons embracing of emotions in his theory of social action and Elias' exploration of a long-termed civilising process of increasing affect-control. Next, I will show how the concern with questions of emotion is also written into the sociological literature by a later generation of sociologist such as Goffman, Hochschild and Strauss. These three theorists pursue William Shakespeare's dictum, 'all the world is a stage and the men and the women merely players'. By doing this they embrace emotions in a micro-sociological perspective with an explicit focus on the daily, step-by-step efforts, taken on by each member of society to constitute a functioning social order.

Second, I will address texts that specifically focus upon emotions in health care practices. The first body of research in this section will argue that emotions are important social components in organisational functioning of health care organisations. The second body of research will envision, through the nursing profession, how emotions are personal properties that can be used, manipulated and controlled in the hand of others.

Third, I will discuss and outline a theoretical framework which combines the approaches without falling prey to the three aforementioned analytical challenges. This framework will open up different perspectives in relation to the remaining chapters of the thesis.

Emotions and epochal battles in social sciences

In the last decades a significant interest in emotions has arisen in the social sciences. Authors have phrased this interest as a sign of an 'affective turn' (Clough and Halley, 2007); a 'new epistemology' (Athanasion, Hantzaroula and Yannakopoulos, 2008); an 'affective society' (Watson, 1999), and a 'post emotional society' (Mestrovic, 1997). Some authors have even characterised contemporary society as 'generation emotion' (Ankowitsch, 2002). The fields in which this intellectual attention to emotions expands are many. It includes anthropology, psychology, sociology, political science, education studies, geography, philosophy, genetics and neuroscience.

One can also witness an increase in the interest of questions of emotion within organisation and management studies. While the topic of emotions was largely non-existing in organisational studies in the second half of the twentieth century, to-day, many researchers are writing emotions back in the game. The acceptance of and interest in emotion-oriented research is manifest in a range of special issues of journals that hit the newsstands these years – for example *Human Relations* (2007); *Human Resource Management* (2002) and *Journal of Health Management and Organisation* (2005). The journal *International Journal of Work, Organisation and Emotion* also testifies this interest. Strangely – as if by an act of academic interaction – while doing research for this chapter I received a call from *The Journal of Political Power* for a special issue on 'Power and Emotion', where the editors frame emotion (and power) as essential features of the conduct and constitution of social life.

The above occurrences point towards a revitalisation. The organisation theorist, Stephen Fineman, situates the status of emotions in organisation studies in *The Sage Handbook of New Approaches in Management and Organisation* (2008). He writes:

'In the past decade or so emotion has moved from being a marginalized, even silenced, discourse in organisational and management studies to being something of a "must" [...] Emotion is no incidental fuzz to "proper" business, but a substantive feature of what happens and what matters' (2008: 239).

Previously, organisation studies have indeed been occupied with addressing emotions and emotional aspects of human behaviour but they have done it, I will argue, along the lines of what is expressed in the above quote of Fineman, in ways that effectively separated emotions from organisational life. A well-placed example is Morgan's (1986) image of the organisation as a psychic prison that fosters a kind of 'production psychosis' where individuals' feelings are trapped in the rational structure of the organisation (1986: 224). Along the lines of this captivating image, organisation and management researchers have traditionally approached emotions as vital objects of resistance or as individual properties that were being hemmed in by rational organisation structures in the service of predictable means and ends. These previous attempts - as illustrated in Morgan's theorising - to overcome an apparent neglect of emotions tend to reproduce the ontology of the older approaches – for example Frederic Taylor's rationalistic technocratic theories of organisations (see for example Taylor, 1967 [1911]) or Weber's work on rationalised bureaucracies (see for example Weber, 1978 [1925]) – that they were critical of. In relation to organisational analysis, these attempts fostered some guiding principles of 'individual feeling' on one side and 'organisational rationality' on the other (e.g. Albrow, 1994), which in my opinion is not

suitable for a coherent understanding of how emotions function within the framework of organisational life.

The contrasting position developed and offered throughout this thesis is therefore quite different from the above example. Emotion and reason are seen as intertwined components constituting the activity or the conduct in question both for the organisations and for the individuals. My analysis of emotions at work in a cancer clinic thus departs from the traditional principles of 'individual conduct' and 'organisational conduct'.

As I have mentioned in the introduction, a small group of sociologists have sought to explore 'questions of emotion' through their theorising on persistent sociological concerns – for instance the question of 'individual' and 'society'. Even though Weber, Parsons and Elias, whom I will turn to next, wrote in manners motivated by a faith in empirical science and the capacity to observe human behaviour and social arrangements by highly detached means (Greco and Stenner, 2008), they all had a profound interest in the place of emotions in modern society.

One of the reasons for turning to these classic sociological approaches to understand emotions at work in a cancer clinic today, is to overcome the always present danger of claiming a unique novelty associated with the study of emotions; an approach, which easily would oversee historical continuities and important achievements within previous research. Roughly spelled out, one might even say that we do not need new theories of emotions. Instead, what are called for are detailed empirical inquiries of already existing propositions that arise from domains of already existing social theory. It might be that there has been a tendency towards emotion-related studies in different fields over the last decade, but emotions per se have always been an integrated part of understanding. As a means to stay away from redescribing 'epochal battles' (Hunter, 2006) and to avoid approaching questions of emotion in a

² Making epochal claims have a long history in the academic tradition. The main problem with making different classifications of forms of society (such as, for example, 'affective society', 'acceleration society', 'network society and et-

overly sentimental way, for example by approaching them either as novel 'objects of resistance' or as pure 'sources of documentation practice' (see my introductory story from the cancer clinic), this chapter will seek to establish a nuanced, combined theoretical framework. This framework will point both backwards in relation to its engagement with classic sociological concerns of human conduct and lead forward in relation to situating the inquiry in a very contemporary context. Also academia stands on the shoulders of giants.

Weber and emotions in bureaucracy

In his work, Weber linked bureaucratic forms of rationality with the rise of modern capitalism. The prime manifestation of the rationalising process was the emergence of the bureaucratic organisation. This organisation type was designed to achieve specific goals, characterised by a high degree of division of labour and detailed norms and rules that governed the behaviour and manners of the employed staff in a certain way. The bureaucratic organisation was an almost perfect rational form because it stands under the principle of *sine ira et studio* (in Latin: 'without hate and zealousness'). For a bureaucracy to develop the most efficient methods of achieving its goals, it must depersonalise every possible work procedure. This vocation explains the codes of conduct guiding the activities of the individual employee – including principles of self-control and, not least, self-constraint. Weber writes in *Economy and Society* (1978):

'Bureaucracy develops the more perfectly, the more it is "dehumanized", the more completely it succeeds in eliminating from official business

cetera) or turns ('affective turn', 'emotional turn', 'linguistic turn' and etcetera) as the point of departure, is that one tend to emphasize societal changes as essential, leaving previous conceptualisations of existing orders behind as something completely different from new emerging forms. See for example Schanz's critique of postmodernity (Schanz, 1999) or Hunter's discussion of intellectual 'epochal' battles in social sciences and humanities, and how these battles have provoked endless, yet fruitless, series of dialectical reconciliations (Hunter, 2006).

love, hatred, and all purely personal, irrational, and emotional elements which escapes calculation'

(1978: 975).

In this sense, emotional conduct is dismissed as being irrational merely because of its lack of calculability; its instances of impulses and, not least, because of its foreignness to official rule following. Weber further writes:

'[He] performs his duty best when he acts without regard to the person in question, *sine ira et studio*, without hate and without love, without personal predilection and therefore without grace, but sheerly in accordance with the impersonal duty imposed by his calling, and not as a result of any concrete personal relationships' (1978: 600).

A straightforward interpretation of this statement may be that Weber and his ideas of bureaucracy eliminate everything which cannot be made calculable, namely individual and unruly emotions. In extension of this interpretation, some authors have argued that the rule-based iron cage of organisational control suspends the emotional elements of the individual and leads to an increasingly dehumanisation of organisational life. One strand of this argument stems from the work of Bauman for whom instrumental processes of the rational bureaucracy result in a 'dehumanisation of the objects of bureaucratic operation' (quoted in Du Gay, 2000a: 40). Hence, one of Bauman's concerns is to save elements belonging to the emotional, irrational and private sphere from the rational, non-emotional and public sphere. The work of Weber has indeed been approached from different positions, and my suggestion is that we look for alternatives to Bauman's dualistic account if we are to understand emotions and reason as intertwined modalities of human conduct. In relation to my analysis of emotions in a (bureaucratic) hospital setting, I find it useful to reflect on how We-

ber's theorising on questions of emotion gives way to thinking about particular work settings and the forms of conduct which emerge from these settings.

Du Gay challenges the 'dehumanisation thesis' by stressing that the emphasis upon depersonalisation in Weber's work, for example when Weber (1978) writes 'that bureaucracy develops the more perfectly, the more completely it succeeds in eliminating from official business personal, irrational, and emotional elements' (see full quote written above), is a commitment to the purpose of the office, independent of personal idiosyncrasies. For Du Gay (2000a), the stress on *impersonality* as a crucial feature of bureaucratic rationality in Weber's descriptive analyses, is not tantamount to a denial of humanity or emotional relations as long as 'these do not undermine the ethos governing the conduct of that office, through for example, opening the doors to corruption or encouraging inappropriate forms of patronage' (2000a: 75). If we follow this line of thought, we see how Weber does not dismiss emotions as such, as long as they do not set aside formal procedures of the office. In this regard, a good medical professional for example, is 'impersonal' in a manner which is objective, so she or he is able to attend to the matter at hand. Hence, the conception of impersonality also refers to professional work activities and the professional (bureaucratic) capacity to treat for instance clients as cases, without paying attention to their status or personal attributes, such as type of personality or (deviant) character. This is clearly the case in the cancer clinic, where doctors must decide upon cases unaffected or unbiased by personal relations. The bureaucratic regulation of emotions has two main purposes. First, it shields the employees from getting (too) emotionally involved in particular cases. Second, it protects the recipient of bureaucratic services, the person who attend the office for help, from being judged according to the office holder's personal motives.

To some, the preoccupation with emotions in Weber's work – as a companion to reason – might be a surprise in the light of readings of more traditional understandings of Weber and his genealogy of the *Berufsmench*. However, his concep-

tion of rationality and rational action might not be a definitive. Most obviously, Weber's analyses of 'charismatic authority' (1978: 241-245) demonstrate how processes of rationalisation do not lead to a disappearance of emotions or an emotional deficit. Instead, emotions are redistributed in a variety of different forms of conduct through social life. Reason and emotion, in short, are never entirely separable. According to Hennis (2000), Weber himself was horrified by 'the emergence of which was the central theme of his studies: the arrival of the "specialists without spirit" and – perhaps even more topically – "sensualists without heart" (2000: 80).

Certainly, Weber does allow for emotions in the bureaucratic office as long as they are a part of rational work processes and not, for example, an exercise of personal patronage, based on holistic beliefs or magic talents. We also see how the management of emotions in public administrations has profound value for the social order and for the purpose of the organisation to stay intact. Weber (1978) shows in his analyses of the bureaucratic organisation, how possible conflicts emerge if a substantive justice is oriented toward some concrete instance and person, because it will unavoidable 'collide with the formalism and the rule bound and the cool "matter-offactness" of bureaucratic administration' (1978: 980). Hence, we see that Weber in relation to the relative positions of reason and emotions points to the fact that there are serious amounts of (emotion) work involved in being impersonal and ascribing impersonality to social interaction and, therefore, we also see how this vocation of detachment becomes a central issue of concern in both his writings and in relation to the research manners he applies to the study of social realities.

Parsons and the conduct code of 'affective neutrality'

As inheritor of Weber, Parsons' theorising also embraces emotions. The tendency to see emotions and rationality as opposite ends of a continuum is perhaps more persistent in the work of Parsons than it was the case in Weber's work. This is mainly due

to Parsons' emphasis on the process of modernity as a long-termed history of rationalisation of social relations – without exceptions.

Parsons is preoccupied with exploring the conditions of possibility of human conduct in modern society. His understanding of modernity is inspired by Tönnies and his distinction of Gemeinschaft (i.e., community) and Gesellschaft (i.e., society) as a way to understand the transition from pre-modernity to modernity as a move from nature to society. Parsons argues that this transition implicated a transition from emotions to reason. Drawing on an analysis of Tönnies' social typology, Parsons believes that 'community' is characterised by affectivity while 'society' is characterised by affective neutrality. The pattern variable points to the rules and values, which determine the activities of social actors in specific situations. In this regard, his concept of 'affective neutrality' also refers for instance to the specificity of emotion (or affect) management that is expected or appropriate in given organisational forms of interaction. For example, contact with other individuals in a bureaucracy is maybe most effective when devoid of emotions and characterised by 'affective neutrality'. According to Parsons, 'affective neutrality' is a significant feature of professionalism and professional work practices. In one of his descriptive analysis, he deals explicitly with modern medical practice and its practitioners (see Parsons, 1951: 428-79). The functional specificity or the affective neutrality of the medical profession enables medical individuals to perform their clinical tasks 'without regard for persons'. Parsons writes:

'[Affective neutrality] enables *the physician* to penetrate sufficiently into the private affairs, or the particular nexus of his patients to perform his function. By defining his role in this way it is possible to overcome or minimize resistances which might well otherwise prove fatal to the possibility of doing the job at all' (1951: 459).

In contrast, affectivity points to the expression of gratification of emotions or an individual pursuing of 'any interests private to himself as distinguished from those shared with the other members of the collective' (ibid). Following the latter perception, we may conclude that Parsons did not deny the presence of emotions. Rather, he delegated emotions to the private spheres – for example the sphere of the family or friendships. From this follows that if individuals anyway go public with their emotions, these emotional outbursts can be interpreted as disturbances to the social system.

In sum, we see how Parsons' conception of rational (i.e., instrumental) actions contrasting expressive action (i.e., emotional) may frame the relationship of emotion and rationality as being that of two opposites. However, if this was the case there would be no reason for including his theorising in the framework of this thesis. Instead, and very much like the interest of Weber, Parsons is preoccupied with how public office holders keep inappropriate feelings at bay in their professional work activities and how they manage to sustain an image of a certain kind of person. Ideally, the vocation of 'affective neutrality' provides the conditions of possibility for a doctor and his patients to meet on impersonal ground in the medical encounter. Parsons provides us with a conception of the clinic that is hemmed in from personal beliefs and value-laden judgments. This particular kind of space furthermore leaves issues such as economic status and personal attributes free of observation. He writes:

'It is also important that doctors should not let their personal dislikes of particular patients be expressed in a poorer level of treatment or even positive "punishment". And doctors would scarcely be human if they did not take a dislike to some of their patients' (1951: 459)

As indicated in the quote, Parsons also acknowledges that this functional setting is heavily challenged in doctor-patient relationships. The emotional expectations of the patients and their families, which are typically sources of hope and frustration, are contested in medical practice – and especially in situations where no explanations or treatment suggestions can be provided. The doctor must then work hard to regulate the emotions of frustrated individuals. The powerful attachment that patients express through their emotions in a medical encounter may be hemmed in effectively by the doctor if she or he adheres to a professional attitude of, for instance, 'affective neutrality'. From another point of view, physical intimate situations can also provoke emotional reactions (e.g., disgust, anxiety) in professionals. Here the professionals must work through various techniques to sustain themselves as professional persons. However, because of the particular function of the medical encounter, Parsons suggests that doctors – even though they are expected to have a purely rational orientation to undertaking their work – are constantly susceptible to personal responses in their intimate dealing with others human beings. This makes Parsons' theorising on medical practice an apt work for this thesis, because he situates the delicate balance between emotional expectations of individuals and professional codes of conduct in the very heart of the doctor-patient relationship.

Elias and historical processes of 'social restraint towards self-restraint'

Weber's interest in emotions at work in specific socio-historical settings is continued in the work of Elias. As registered recently, Elias' influence on organisation studies is considered minimal and to date his approach has been largely neglected by the field (see for example Van Iterson, 2009). This is puzzling because the broad interests of his historically informed account of *The Civilizing Process* (Elias, 2000) – ranging from the account of modern state formation and changing standards of manner and morality, to forms of affect-control and self-constraint in social relations – provide enough intellectual goods to engage with empirical oriented organisation studies. Elias' work sheds light on the forms of emotion management and bodily control, which emerged as part of the historical processes of rationalisation. He emphasises

how emotional standards are depending upon specific social arrangements, practices and techniques rather than being essential properties of a human being. This is interesting in relation to the questions raised by this thesis, since Elias' work provides the means to understand the emotional conduct of doctors 'outside' the individual practitioner. He offers a position that turns our attention to tendencies in our society, which insert for instance demands for 'compassion' as a normative performance allotment in today's professional bureaucracies.

A core theme in Elias' work is how changes in power structures are reflected in changes in the 'psychological make-up of people' (2000: 369). He links the notion of *civilité* or *civilization* to changes in codes of conduct, and shows how emotions are increasingly controlled in rationalised (i.e., civilised) societies. By doing this, he develops Weber's (1978) historical view on bureaucracy, and the impact of rational practices on the organisation of the social sphere and interpersonal relationships. In his work, Elias links an emergent notion of *civilité* with changes in the production and display of emotions, or, as also Goffman later turns his attention to, changes in how individuals go public with emotions.

One of Elias' core points is that, as processes of civilisation have developed, so have demands for emotion management increasingly become internalised. He demonstrates how rules of behaviour and manners became both far more complex and far more controlled since the Renaissance, as particular evident in the emergence of the modern state and the 'civilised' individual. Elias writes:

'As more and more people must attune their conduct to that of others, the web of actions must be organised more and more strictly and accurately, if each individual action is to fulfil its special function. Individuals are compelled to regulate their conduct in an increasingly differentiated, more even and more stable manner' (2000: 367).

The development of an increasingly complex web of social interdependencies in turn makes it necessary for individuals to constrain and manage their emotions – both of the self and of others – and hence an increasing social restraint towards self-restraint was established (Van Krieken, 1990, 1996). Elias pin-points how the experience of certain emotions, such as embarrassment and shame, emerged as bodily expressions in the sixteenth century and encouraged individuals to increasingly repress their passions and to constrain themselves from the vantage point of others. Certain feelings then gained status as instruments of control of self and others. As regulatory rules they were accepted out of fear of losing out in the ongoing competition for status, power and economic resources and not least out of fear of feeling shame and embarrassment in front of others. The behaviour associated with civilité was hence directed to the regulation of the conduct of individuals. What was getting rationalised in these civilising processes was, according to Elias, not only the product of individuals, but primarily 'the modes of conduct of certain groups of people' (Elias, 2000: 412). This included proper table manners of people eating together, gestures, facial expressions, ordinary dressing and handling of natural bodily functions, such as spitting rules and how to blow one's nose in public.

Consequently, specific codes of habits of thinking and patterns of affect-control are characteristics of particular organisations. Elias demonstrates this argument empirically in French court societies, the nuclear family and areas of sport and leisure. At the court, for instance, highly detailed codes of emotion management and bodily comportment were developed that served to distinguish individuals in terms of their social status. For a knight to sustain his 'knight status', he was dependent on a rationally calculated adherence to certain codes of manners and rules of impression management (Kuzmics, 1987). Whereas previously in history, violence or emotional outbursts were met with affect, now, calculated situations were met with calculated appearances.

It is important to mention in this context that Elias' theory of the civilising process of human conduct does not involve a repression of emotions or emotional life as such. In other words, Elias' historical analyses of changes in personality structures and socio-political changes and of how these correlate involve an analysis of how the experience of some emotions (e.g., embarrassment) expands and how the experience of others emotions involves restraints (e.g., anger) in Western societies.

What started out as merely external control in the name of monitoring gradually became internalised through the entire structure of society as the conscious ability of individuals to automatically regulate instincts and bodily and emotional expressions. By studying manner books from different eras, Elias furthermore analyses the development from external social restraints on manners and etiquette to a more informal approach to emotional conduct. However, as Elias emphasises, the shift from social restraint towards self-restraint implies often more well-choreographed and socially sanctioned rules of appropriate emotional display. In others words, codes of conduct have become less rigid and more differentiated but at the same time new demands to the 'affect-economy '(Elias, 2000: 399) have emerged, which includes the emergence of highly sophisticated techniques of emotion management. This development has been given further notice in the work of the Elias follower, Wouters (1986, 1999), who describes the processes of 'informalisation' of emotions like this:

'the relaxation of the social codes [...] in combination with increasing social demands on self-control; it implies a change in the patterns of social control and self-control and also a higher level of reflexivity on the part of individual people (1999: 416-417).

The expression resembles the one used by Elias when he speaks about processes of 'constrains to be unconstraint' (Elias, 2000: 365-379). It indicates that processes of self-constraints have not only become more flexible, at the same time they have also

become stricter in recent developments of controlling emotions. These dimensions take on new forms in our current era of for example self-help culture. The well-known situation where a self-help coach says: 'just be your natural self in order to give other people the impression of the person you really are' does not represent a loosening of emotional display, but rather represents a refinement of emotion codes through seemingly more 'natural' or 'sincere' practices. The coach's call for 'authentic' behaviour or 'just-be-you-behaviour' may employ lots of calculated (emotion) work on behalf of the client.

In sum, the work of Elias contributes to an understanding of emotions at work in specific historical and organisational contexts. It may provide us with a preliminary lens to look through when we analyse for instance processes of formalisation and informalisation of feelings in a hospital setting. His perspective draws attention to how the conducts of doctors and the exhibition of specific emotions in medical relationships are interwoven with social rearrangements of cancer illnesses and the introduction of new forms of governance in relation to medical practices and manners.

Goffman and the staging of a cynical performer

Elias provides an excellent platform from which to understand the historical mouldering of emotions through processes of rationalisation. His preoccupation with the sociological concern for the relationship between human agency and the social world is manifest in his preoccupation with social formations of individuals. His sense of self-hood and the formation and regulation of individuals' conduct – including the display and expression of emotions – is also well developed in the work of Goffman.

Goffman was, as his work clearly demonstrates, not exactly embarrassed in his dealings with emotions. The main thrust of his micro-sociological perspective on emotions may be his explicit focus on the daily, step-by-step efforts, taken on by each member of society to constitute a functioning social and interactional order. He

proposes that emotions are socially constituted and actively managed on a daily routine basis by reflexive individuals who freely participate in the creation of all kinds of emotional performances. Goffman's perspective challenges the traditional way of thinking about human agency and social structures and we may see how emotions become a cornerstone in that manoeuvre. Goffman (1959) writes:

'As human beings we are presumably creatures of variable impulse with moods and energies that change from one moment to the next. As characters put on for an audience, however, we must not be subject to ups and downs [...] a certain bureaucratization of the spirit is expected so that we can be relied upon to give a perfectly homogenous performance at every appointed time [...] we are helped in keeping this pose by clamps that are tightened directly on the body, some hidden, some showing (1959: 63-65).

Goffman stresses the role of the situation; the frame and the interactional order, which give direction to human conduct. As such, a situation always carries with it an official frame of what it is appropriate to feel. This frame is set within a wider structure of values, norms and cultural beliefs in which individuals work effortful trying to make emotions and frame consistent. Within this frame individuals can act in a strategically and calculated fashion appropriate to the situation.

Like Elias, Goffman's work implies that there are generic emotions of everyday life, such as embarrassment³ and shame. He understands embarrassment as the disparity between self-identity and social identity; i.e., a threshold between at the one hand the way individuals see themselves as competent members of society and at the other hand the way others see them (Goffman, 1956). As an experienced emotion and bodily expression, embarrassment is an important element of rational action. Embarrass-

³ The emotion embarrassment plays a central part in especially Goffman's early studies (Goffman, 1956; 1959; 1961). He argued that embarrassment had 'generic properties of interaction' (1956: 264).

ment, along with feelings of guilt and shame, can be understood as a central emotion to organisational control, prompting the way an individual is judged in the face of others and, thus, the way one can sustain self-control and not be 'gripped' with feelings in inappropriate situations. Instead of turning these emotions merely inwards (consider Goffman's famous distinction between front stage and back stage), Goffman was curious towards the emergent emotions that surface during face-to-face encounters. The stage and performance concepts as typically drama metaphors are crucial in Goffman's conception of social life. Self-presentation amounts to the way:

'the individual in ordinary work situations presents himself and his activity to others, the ways in which he guides and controls the impression they form of him, and the kinds of things he may or may not do while sustaining his performance before them' (1959: Preface).

Goffman assumes that the self we live by constitutes action as well as it is constituted by it. Even though selves are modelled as personal selves, the image of the personal self is itself a presentation that is publicity accomplished and constantly worked on and worked with. The self is therefore considered a social product⁴. It is crafted by interplay of circumstantial demands, restraints, and resources on the one hand and self-constituting social actions on the other. Even though the self is constrained by culture, language and biological peculiars of one's life, it is given some slack in the dramatic performance in front of others. Performers can engage in strategic, cynical behaviour to achieve not only dramatic goals but also specific goals that are unknown or unrecognised by an audience.

⁴ Goffman's analysis of the presentation of the self is a critical attack on the social institution of a self-contained individual. We see a similar critique developed in the work of Elias, where the myth of a closed self, 'homo clausus', which is located inside human beings, is challenged (Elias, 2000: 479-481).

Goffman's conception of self in everyday life is constantly preoccupied with how this self stands in the eyes of others and how it implies states of different emotions: pride, humiliation, shame. His thesis was that if one or more of the perquisites for interaction are unconvincing or are altogether missing, then strong emotions would occur. In this way, the impression one makes can be threatened by inappropriate conducts. Performers may play incomplete because they are flushed with emotion and therefore feelings might intervene in their performance. Goffman (1959) mentions three groups of inopportune events. He writes:

'First, a performer may accidentally convey incapacity, impropriety, or disrespect by momentarily losing muscular control of himself. He may trip, stumble, fall; he may belch, yawn, make a slip of the tongue, scratch himself, or be flatulent; he may accidentally impinge upon the body of another participant. Secondly, the performer may act in such a way as to give the impression that he is too much or too little concerned with the interaction. He may shutter, forget his lines, appear nervous, or guilty, or self-conscious; he may give way to inappropriate outbursts of laughter, anger, or other kinds of affect which momentarily incapacitate him as an interactant; he may show too much serious involvement and interest, or too little. Thirdly, the performer may allow his presentation to suffer from inadequate dramaturgical direction. The setting may not have been put in order, or have become readily for the wrong performance or may become deranged during the performance (1959: 60-61)'.

Generally, Goffman's approach to understanding emotions at work captures that individuals actively manage emotions according to the rules of a particular situation. These rules include *expression rules* and emotional performers who are capable of managing emotions in accordance with such rules. Expression rules are conventions

guiding the display of feelings that performers must adhere to in the situations. The rules oblige performers to 'give off' some feelings and not others depending on the situation. In abiding by the expression rules and suppressing and controlling alternative expressions, performers can create a favourable impression on others. Goffman assumes that when individuals appear before others he or she will try to control the impression they receive of the situation. The arts of impression management refer to the sample of techniques and attributes that are required of a performer 'for the work of successfully staging a character' (1959: 203). Essentially, when performers are 'on stage', they must express the emotions appropriate to the part they are playing. In this sense, emotions function as social ventricles. For example, when a doctor discloses a diagnosis of cancer to a patient, he may deliver the information in a respectful, anxiety-reducing way, so he and the patient together can focus on the next step of the treatment process. A contrasting behaviour, for example if the doctor makes fun of the situation and adheres to a humorous behaviour, would make an impression that is totally out of place in relation to the interaction rules of the specific situation. Goffman argues that in order to avoid negative emotions (like anxiety or confusion) individuals are more than ready to follow interaction rules such as expression rules.

Although Goffman's view on the role of emotions in social interactions is much more complex than I have hinted at above, he basically understands the expression and display of emotion as depending on the specific contexts or stage performances in which they occur. In regard to my empirical analyses of doctors' conducts, expression rules are equipping the doctors – as performers on a stage – with certain capacities to perform in the cancer clinic. The role of 'doctor' and the role of 'cancer patient' are defined by the setting and the rules that structure, for instance, the medical encounter in the outpatient clinic. 'To be a given kind of person is not merely to posses the required attributes, but also to sustain the standards of conduct and appearance that one's social grouping attaches thereto' (1959: 81). However, how do we, according to Goffman, focus on the emotional 'ups and downs' of routine daily inter-

actions? Well, the answer is that we become aware of emotion work in situations where individual's feelings do not fit the situation and there is a discrepancy between the latter and legitimate feelings – of what one 'should' feel and what one 'tries' to feel. In his analysis of social interactions in welfare institutions, Goffman shed light on how often polished selves – the client, the mental patient, the social worker, the doctor - become disorganised in the course of routine social interaction and how these selves try to overcome or avoid the problems through preventive or corrective practices. This disorganisation becomes an objective for the emotion-interested researcher.

Hochschild and the search for an unmanaged heart

Goffman's approach to emotions in human encounters is a stepping-stone for Hochschild's work on how (capitalist) society makes use of emotions through the commercialisation of human feelings and the loss of authority this involves. Hochschild adopts important elements of Goffman's framework to her work, but adds a critical edge drawing on Marxist concepts of alienation and Freudian concerns with pathologic emotions and the suppressing of affects in modern societies. Let us take a closer look at her social theory of emotions. Her conception of 'manufactured feelings' and 'authentic' feelings demands some attention, because it installs some confusion, which I believe to be connected to her understanding of the self. This issue will be explained later in this section. I have first and foremost chosen to engage with her work because of her major influence on the last decades' empirical emotion studies, in particular studies of emotions in health care.

First, Hochschild (1983) provides us, in her own words, with 'a set of illustrated ideas about how society uses feeling' (1983: 17). Society reveals an *emotion culture*, consisting of a complex of ideas about what people are supposed to feel in various types of situations and how we must manage our emotions in front of others. This emotion culture is composed of various *emotion ideologies*, which define proper

attitudes, feelings and emotional responses in social activity. *Emotion rules* specify the emotions that individuals should feel, experience and express in specific situations. Hochschild writes:

'Feeling rules are standards used in emotional conversation to determine what is rightly owed and owing in the currency of feeling. Through them, we tell what is 'due' in each relation, each role. We pay tribute to each other in the currency of the managing act. In interaction we pay, overpay, underpay, play with paying, acknowledge our dues, pretend to pay, or acknowledge what is emotionally due another person. In these ways [...] we make our try at sincere civility' (1983: 18).

Feeling rules, as implicit codes of conduct, are thus the ideological strategies we reproduce to deal with distressing, uncomfortable or inappropriate emotions and feelings. *Emotion management* is the type of work it takes to deal with feeling rules. Traditionally we may think of emotion work as something we do when we care for our families. However, Hochschild (1979) takes the concept further by situating it in a commercial context. She defines *emotion work* as:

'the act of 'trying to change in degree or quality an emotion or feeling [...] Emotion work refers more broadly to the act of evoking or shaping, as well as suppressing, feeling in oneself' (1979: 561).

Emotion work involves individuals' management of feeling in order to conform to the demands of a particular situation. The use of the word *work* stresses that we have to do with something that is actively done to feelings. This work is done with a purpose of producing suitable emotions, 'I tried to calm her down', 'I was crying inside but forced myself not to show it'. We see here how Hochschild takes Goffman's important distinction between front stage (where performance take place; where a

'show' is on) and back stage (the work going on behind the scene) one step further in her work. She distinguishes between emotion work as belonging to the private realm (back stage – for example the family, friendships) and *emotional labour* as belonging to the public realm (front stage – for example the work place, the office). Emotional labour is 'the management of feeling to create a publicly observable facial and bodily display' (Hochschild, 1983: 7). Through this term, Hochschild coins the invisible character of the dimensions of the relational work people do as part of performing their job.

In the private realm, emotion work is considered part of our private lives. In the public realm, emotional labour is sold for a wage as a commodity. Hochschild associates this type of labour with jobs that require workers to produce a positive or negative emotional state in others and endure supervision and control over their emotional activities. She argues:

'[E]motional labour includes knowing, and assessing as well as managing emotions, other people's as well as one's own [...] as part of the personal manager's emotional labour, he has to learn the company's "emotional map" (1993: X).

As such, real-time emotions are a large part of what workers manage on their job and 'emotional labour is a big part of white-collar work' (1993: xii). This kind of labour is significant in professional bureaucracies such as hospitals, welfare offices and schools, and private companies such as call centres, air companies and fun parks⁵. In her analyses of how flight attendants manage to sustain the pleasant demeanour dictated by feeling and display rules in the face of rude passengers in air traffic,

⁵ Drawing on Daniel Bell's (1973) definition of service work in post-industrial society, this includes jobs in transportation and utilities, distribution and trade, finance and insurance, professional and business jobs, jobs deriving from demands for leisure activities (recreation and travel), and jobs that deal with communal services (health, education and government.

Hochschild challenges Goffman's concept of *surface acting* by focusing on how individuals cope with emotion rules and emotion ideologies and how people are forced to engage in behaviour that arouses negative emotions (e.g. stress, alienation). Surface action refers to a strategy of pretending to feel for instance happy, when one does not feel happy at all, and hence the emotional display of happiness is merely a display act. *Deep acting* in contrast, takes us one step further away from the stage on which we 'just' play a character. Hochschild (1983) writes:

'For Goffman acting is surface acting. The actor's mental focus is on the slope of a shoulder, the angle of a glace, or the tightness of a smile; not on any inner feeling to which such gestures might correspond [...] we need a self with a developed inner life '(1983: 226).

In building upon the work of Goffman, she is also adding to his work by focusing on the inner voices of actors. She writes: 'The emotion-management perspective fosters attention to how people try to feel, not, as for Goffman, how people try to appear to feel' (Hochschild, 1979: 560). By doing emotion work and comply with feeling rules, individuals try to alter their inner state and what they feel 'deep down'. In contrast with 'surface acting', deep acting describes the actual work on individuals' emotional state. For example, an individual can try to change her expressive behaviour such as trying to cry so as to accommodate her inner feelings.

According to Hochschild, Goffman is black-boxing both emotional reflexive actors and social structure, who is the origin and causes of change in codes of conduct in his work. Her focus on 'the self as emotion manager' (1979: 555) is a way to install a battling inner self in micro-sociological theory. It is a self which is capable of feeling; who knows what to feel in specific situations and who is capable of managing own feelings. However, I find it analytically difficult to differentiate between what individuals consider belonging to them (for example feelings representing their

'inner' selves), socially appropriated feelings, and feelings that are displayed for public exchange and consumption (e.g. Wouters, 1989). We see how the inner/outer distinction is well-developed in Hochschild work when she distinguishes between the commodification of (phony) emotions as capitalist properties and an authentic inner self with sincere feelings. The demands to corporately restricted emotion management result in 'the issue of estrangement between what a person senses as her "true self" and her inner and outer acting becomes something to work out, to take a position on' (Hochschild, 1983: 136). By drawing a relatively sharp distinction between private, emotional self and emotional demands of public jobs, Hochschild notices that 'emotional labour poses a challenge to a person's sense of self' (ibid). I will argue that one of the main problems in drawing this distinction, is the idea that 'authentic' and 'sincere' private emotions are entities belonging to a self-contained self (reconsider Elias notion of 'homo clausus' – see Elias, 2000: 474) which must be protected against occupational exploitation.

Hochschild never applied her emotional labour perspective to the medical profession or to nursing. Especially the latter would have been a natural expansion of her empirical inquiries into the study of emotions at work. One of the reasons for this omission might be found in the way she constructs the argument of professions and professional codes of conduct in the discussion of emotional labour and jobs (see Hochschild, 1983: 56-76). To limit emotional labour to specific jobs, she includes three criteria in her theoretical framework – first, face-to-face or voice-to-voice contact with the public; second, the requirement that employees should produce a particular emotional state in another person; and third, methods of supervision and training of the employees which allow the employer a degree of control over the emotional activities of employees (1983: 147). Compared to a flight attendant, Hochschild notes that doctors do have personal contact in public and they try to affect the emotional states of others, but they do not work with an emotional supervisor immediately at hand. She writes: 'Rather, they [doctors] supervise their own emo-

tional labour by considering informal professional norms and client expectations' (1983: 153). Surprisingly, when Hochschild talks about emotional labour as efforts or work undertaken in most service jobs, it in reality comes down to include only a few emotional labour jobs such as flight attendants, hairdressers and beauty stylists. She clearly makes the limitation to serve her focus on gender and social status: the exploitation and cost of emotional labour, typically goes on in lower income, female jobs. In building this argument, she seems to ignore emotional labour of 'true' professions, which is where this thesis makes an empirical contribution to her work.

In sum, Hochschild's contribution to the work of Goffman is to demonstrate how individuals do not simply manage their emotions to give off a good impression, but also how they are forced to violate their 'inner' feelings to comply with situational feeling rules that are inherent in certain job profiles. Her idea of deep acting includes that individuals take on laborious work to alter their 'inner' feelings so these feelings fit with outer expectations and social norms. The latter concern is linked to broader questions of individuality, power and social structure, and Hochschild connects emotional labour to a 'dark side' of (capitalist) ideology in Western societies. According to this approach, we can think of emotion as a covert resource which can be exploited and regulated like money, knowledge or physical labour. Those who perform emotional labour thus become subject to 'the rules of mass production' (1983: 198). From this follows that also our feelings now emerge as targets to colonising attempts and organisational control, and thus nothing is sacred to capitalist exploitation.

Despite illustrating ideas about how organisations use feelings, we see how Hochschild's approach to emotions is open for criticism on a number of accounts, because her theoretical grounding in the image of the self as a self-sufficient and quite independent being, exhibits some problematic conceptual traps. Especially her distinctions between inner/outer; public/private and 'manufactured or fake feelings'/'authentic or sincere feelings' have had widespread appeal in studies of emo-

tions in health care, where they reclaim sincere or authentic feelings from cooperate attempts of control, manipulation and various forms of management. I will return to this issue later in this chapter.

Strauss and invisible, informal emotion work

Next, I will turn to the writing of Strauss and his conception of *sentimental work* in his sociology of organisations. He may provide us with a missing link in relation to Hochschild's theorising, which enable us to approach emotion work done by professionals (doctors) not only as work done for a wage but as work required for other activities to be done both effectively and with respect for the individuals involved. In *Social Organisation of Medical Work*, Strauss and his co-authors (1997) explicitly focus upon work in relation to the management of illness trajectories as specific organisation phenomena. The concept of trajectory is used to analyse social interaction and it refers to the 'complex interactive acts required to manage that course, along with the consequences or outcomes of that management' (Corbin, 1991: 33). The concept has profound value for Strauss et al.'s approach to the organisation of work because it allows him to break the overwhelming category of work into pieces and interactive acts and to explore their consequences for actions and, not least, for how actors respond to changing conditions.

In the specific work on illness trajectories and chronic illness, Strauss et al. (1997) claim that the work of physicians, nurses and associated technicians has been 'radically and irrevocably altered by today's prevalence of chronic illnesses' and the technologies developed to manage them' (1997: Preface). On this background, the authors ask 'how, in detail, has that work changed' (ibid.). In order to focus explicitly on work and change in work, the authors brake down work into different categories of work, such as 'Machine work'; 'Articulation work'; 'Comfort work'; 'Safety work' and 'Sentimental work'. What counts as work in this optic, 'does not depend a priori on any set of indicators, but rather on the definition of the

situation' (Star and Strauss, 1999: 14). For instance, to die with grace in a busy medical ward can under some circumstances be defined as a kind of legitimate work and keeping a marriage going may be fun, but it also involves tremendous amounts of work. The fundamental pluralism of action paths in workplaces, such as trajectories or marriages, means struggle, conflict and negotiation on an everyday routine basis about what should be done in a given situation – and by whom. One category of work is *Sentimental work*. According to the authors:

'Sentimental work presents an 'important, varied, often subtle, and sometimes very complex type of work [...] and is present as ingredient in any kind of work where the object being worked on is alive, sentient, and reacting' (Strauss et al., 1982: 254).

A great deal of 'real medical activities' depends on sentimental work (1982: 274), especially the main line of diagnostic and therapeutic treatment in a trajectory will be affected if sentimental work is not done or is done ineffectively. For example, the doctors have to manage the emotions of patients while they are continually making mutilating bodily interventions such as taking biopsies and provoke other situations which involve great physical pain. Sentimental work is likewise evoked to ward off patient anxiety for instance in cases where a diagnosis of cancer is disclosed and the doctor tells the patient not to worry too much. Like physical labour, sentimental work requires experienced, skilled and reflexive work, which is subject to divisions of labour.

Sentimental work is often considered invisible mainly due to its informal, ad hoc status in medical work. By invisible, I refer to work that is not actually seen (consider Goffman's notion of back stage work) or work that is kept secret, regarded as shameful or dirty, taken for granted or performed 'when those engaged do not think of it as involving work' (Strauss et al., 1997: 148). Strauss et al. show how sen-

timental work, regarded as invisible work, appears in health care organisations in the service of managing and shaping patient trajectories. Opposed to Hochschild's account, there is nothing extraordinary or alienating in sentimental work. It is necessary work to be undertaken by professionals, patients and relatives. Furthermore, Strauss et al. argue that sentimental work and the baseline for doing this kind of emotion work have been profoundly affected by changes in the medical organisation, for example by rearrangement of patient trajectories and technological developments.

Strauss and his co-authors suggest that emotion management is a common feature involved in 'technical' medical work, not only expressed in the work of lower middle-class, female work, as suggested by Hochschild, but also in work conducted by professionals such as university teachers, politicians and doctors. The authors demonstrate how one may approach emotions in the social organisation of everyday medical work similar to the way one approaches other organisational phenomena. Emotions do not have a magical or sacred status in the organisation. Instead, they constitute an important part of rational medical action. Thus, sentimental work may be regarded as a mundane, routine activity that professionals (doctors, nurses) must take part in either 'because it is necessary to get the work done efficiently or because of humanistic considerations' (Strauss et al., 1997: 129). I think the last sentence is important here in relation to a side-ordering of efficiency and humanistic considerations in medical activities. This idea of a side-ordering of various considerations becomes especially important later in the thesis when I analyse various forms of emotion work in the accelerated cancer pathways, and where so-called 'soft' dimensions of care, such as attention to the patients' psycho-social well-being, are intertwined with concerns of productivity and performance measurement.

Literature on emotions in health care

One of the areas in which research on emotions has been most developed is health care. Within this research field, micro-interactional processes of the management of

emotion in everyday routine work have been studied and linked to broader discussions of how social and cultural factors impinge on the formation and regulation of the conduct of health care workers.

There have only been few empirical studies directly examining the role and function of emotions outside of the nursing profession in the overall area of health care (see Mann, 2005 for an overview of the literature on emotion work and labour and nursing). This is probably due to the assumptions many researchers have in relation to the 'caring' role of the nursing profession. One reason why the study of emotional aspects of, for instance, doctoring is undeveloped may be that researchers have been inclined to adopt a dichotomy articulated within the nursing profession itself that traditionally sees emotion work as belonging to the caring role of nurses while doctors are detached from that sort of things and only are involved in processes of giving information about technical interventions (e.g. Smith and Gray, 2000). I have expressed the facilitation of emotional division of labour previously in this thesis, e.g., the introductory story from the cancer clinic, where a nurse took ownership of the expression of empathy.

One of the core issues of theorising on emotions has in previous sections been addressed as the issue of a longstanding bifurcation between emotion and rationality. This debate continues to be reflected in empirical studies of emotions in health care and I will therefore next turn to authors who 1) intend to bridge this divide and authors who 2) continue to work within this divide.

Bridging the divide. In short, the literature, which I have chosen to engage with here, has a profound attention to the complex intertwinement of emotion and rationality in relation to emotions at work. It addresses the importance of emotions to basis purposes and values of health care organisations.

Working within the divide. Second, I also engage with literature that explores how nurses react to structural and economic changes in the work place in relation to their 'felt' and displayed emotions. I ask the reader to reflect on how these 'felt changes'

of nurses are reframed in the reviewed articles in relation to a notion of 'authentic' feelings.

The texts to be examined reflect these two core positions. Each of them will be taken up in turns. Expositions by the various authors will be highlighted as they speak to those positions. Last, I will sum up this listing and point forward to how I intend to approach emotions in my empirical inquiry.

Bridging the divide: Emotion and rationality in health care

'Emotion is rarely seen as systematic or structured and is often used as a contrast with rational' argues James (1989) in a text on nurses' regulation of feelings in a hospice setting (1989: 17). According to James, this polarised thinking presents an image where rationality is hold supreme and where emotion is seen to contain negative connotations of unpredictability and irrationality. From this perspective follows that emotions are seen as an uneasy fit with organisational values of efficiency, standardisation, time tabling and performance measurement in health care organisations. In order to pave the way for a more coherent image of rationality and emotion, James suggests that for the most part the management of emotions is a routine, predicable social process that relies on conscious, skilful work rather than on 'personality' or 'natural skills', as Hochschild suggested. In emphasising the management of feelings within social processes, such management is a day-to-day matter, relying on professional codes of conduct.

The dichotomous nature of rationality and emotionality, which is evident from a public/private or outer/inner separation of the two concepts, is also challenged by Bolton (2001). Set against a backdrop of structural changes affecting the British public sector services in the late 1990's, Bolton demonstrates how nurses working in a NHS trust hospital are able to juggle the emotional demands made to them by presenting different faces to patients. Sometimes a smiley face is displayed, sometimes a humorous face is needed and sometimes a professional face makes the social interac-

tion work. Nurses can therefore be described as *emotional jugglers*. They are able to wear 'masks' and present professional demeanour, while at the same time being emotional present and fully engaged in the interaction order of the hospital setting. Compared to flight attendants or service workers in a call centre, nurses are 'knowledgeable agents', as Bolton writes elsewhere (2005: 103). The difference comprises that nurses are 'capable of mixing all forms of emotion management according to rules other than those solely controlled by the organisation' (ibid). This capacity for emotional juggling enables nurses to fulfil their job obligations, which are compatible with the instrumental goal orientation of for instance processes of productivity, while providing emotional care for individual patients. Bolton does not theorise through a distinction between 'public' or 'commercialised' feelings on one hand and 'authentic' or 'genuine' feelings on the other hand. Instead, care workers can present a 'smiley happy' face as part of their job without this face necessarily signalling a commodified feeling that is packaged to be consumed by the patient.

Bolton highlights in her work some of the contradictions of emotions at work, but she does it in a way which I define as a 'non-sentimental' approach to workplace emotions. Through a critique of Hochschild's concept of emotional labour, which Bolton thinks is inadequate for capturing the complex emotion management of professional care workers, she also scrutinises the assumption of normative control of emotions in Hochschild's work. This form of control implies that consumer capitalism has 'appropriated all of our feelings so there is no longer any room for sentiments, moods or reactions that have not been shaped and commodified via the "commercialisation of intimate life" (Bolton, 2005: 2). The non-sentimental approach to emotions does not emphasise some emotions, for instance those of the private sphere, to the disregard of other emotions, for instance those of the work place. Hence, it would also be a mistake to approach professionals' detached performances as 'cold' performances, devoid of compassion or 'real' feelings. I think the important argument Bolton makes here, is that it takes the same amount of hard work for nurses

to stay detached as it takes for them to appear, for instance, compassionately engaged. Both kinds of professional appearance involve rational, skilled performances of social actors.

The body of research argues that emotions are important social components in organisational functioning of health care organisations. Rather than being 'illogical', 'irruptive' forces, emotions enrich the attainment of organisational outcome, such as the outcome of the medical encounter. Emotions are central to both patient care and are necessary for health care professionals' participation in clinical work procedures.

Accordingly, a central study by James (1993) pays attention to the management of emotions during the disclosure of a diagnosis of cancer. James argues that 'cancer is a particular apt disease to review in order to analyse the management of emotions in health care organisations' (James, 1993: 96). Emotions are evoked within a diagnosis of cancer because of the diagnosis' physical implications and its social consequences. The consequences of the disease for the individual patient mean that the disclosure of a diagnosis is not just about passing on information, but is also a mean to regulate the disbelief, fear and chaos a diagnosis often evokes. The doctors, who have knowledge of the cancer illness, also have the power and responsibility to regulate and manage the feelings surrounding the disease. According to James, professional care workers learn skills of emotion management commensurate with their position and role in the hospital organisation. She observes some of the techniques through which emotions are managed in a cancer unit, and she makes these techniques applicable to ways of organising emotions in the hospital organisation. The techniques include: the use of particular kinds of space and time (e.g., the waiting room, the medical encounter, the time table); consequent denial of negative emotions; limiting the amount of information released (e.g., holding back information of survival prognosis); formal and informal disciplinary rules (e.g., the regulation and expression of 'professional' feelings, such as feelings of the 'caring' nurse), and hierarchical ordering of the possibility of emotions (e.g., the doctor manages the emotions of the medical encounter by constraining its expression and hereby also dictates how others – for instance nurses – should manage feelings). James' study shows that emotional labour is tougher for some than for others, depending on their level of involvement with the cancer patients.

From this perspective follows that health care workers' capacity to act as 'emotional jugglers' or to conduct themselves as certain kinds of social persons in medical relationships depends on competing forms of knowledge and status. However, we must also assume that the different ways of managing emotions during the disclosure of a diagnosis of cancer are dependent on occupation-specific socialisation practices which define the rules for appropriate expression of emotions for workers.

From an education perspective, Larson and Yao (2005) argue that emotion management training is required in medical practice and they identify the need to teach the acting skills of emotional labour. Empathy is often evoked in thinking on the emotional relationship between health care worker and patient, and this is also the case in this article. The authors define 'empathy' as a 'powerful skill' (2005: 1100) and they propose that empathic responses can be learned through intensive training. They argue that doctors are more effective and enjoy more job satisfaction if they recognise that their work include emotional labour, and if they reflexively engage in the managing of patients' feelings. The training of emotional skills thus includes both those of 'stage performance' and those of coping with the after effects of a stressful work.

This argument is backed up by similar studies that are preoccupied with the training dimension of empathy and emotion management (e.g. Teherani et al, 2008; Neumann et al, 2009). These studies assume that when emotion management is explicitly taught, doctors will experience less stress on the job and the possibility of job burnout will be reduced. The education perspective captures that techniques or

skills of emotion management can be trained alongside other work resources to improve the health outcome of encounters between doctors and patients.

The reviewed texts on training of emotional skills emphasise that emotion work in health care has many connotations associated with self-defensive techniques. Two classic texts deal especially with the issue of how health care workers often attempt to limit their emotional investment within medical relationships through the use of precautionary or protective strategies.

Smith and Kleinman (1989) look at how emotion management is trained in medical schools. They call this training 'the hidden curriculum' of the medical education. The training is 'hidden' because it is neither subject to explicit training or learning methods nor is it something that the students collectively talk about. The authors explore through participant observation of medical students how affective neutrality (Parsons, 1951) is learned as a professional ideology. When dealing with patients, the students learn to manage 'unprofessional feelings' (e.g., disgust, aggression) and to control unwanted emotions (e.g., anxiety, fear) involved in their work (Smith and Kleinman, 1989: 57). Basic emotion management strategies, for instance strategies of objectification, provide students with uniform guidelines and resources for managing their feelings in front of other people. These strategies are also part of a larger protective system to help shielding the individual doctor off from showing strong emotional responses in public. Through the use of professional codes of conduct – such as for example that of 'affective neutrality' as protective shields – doctors are able to display various feelings in medical encounters without becoming alienated on a personal scale.

Menzies' (1960) analysis of nursing and the emotional transactions between nurses and patients in a London teaching hospital is a classic study of how protective shields work in health care. Menzies' core argument is that the intimate body work involved in nursing and the daily confrontation with human suffering and death, means that the job situation 'arouses very strong and mixed feelings in the nurse: pity, compassion, and love; guilt and anxiety; hatred and resentment of the patients who arose these strong feelings; envy of the care given the patient' (1960: 98). In face of these unaddressed feelings aroused by the special circumstances of the work setting, the organisation of emotions in nursing takes on characteristics of a 'social defence system against anxiety'. This system consists of defensive techniques of emotion management to protect the nurses against the experience of strong negative emotions. Menzies lists a long catalogue of techniques, which includes: splitting up the single nurse-patient relationship into task lists, which are allocated to a group of nurses instead of to a single nurse so the relationship do not get too intimate; depersonalisation, where patients are named as cases by numbers and diagnosis codes; detachment from excessive involvement and denial of disturbing feelings; ritualised performances, where standardised nursing procedures actively discourage nurses to take on decisions on their own; 'impression management' in front of patients, where 'brisk, reassuring behaviour and advice of the "stiff upper lip", "pull yourself together" are characteristic stage performances (1960: 103), and so on. In sum, the social defence system does facilitate the evasion of feelings, but the self same techniques which help to protect the nurses also reduce the humanistic aspects in the provided services. As such, the 'paranoid-schizoid defence system' (1960: 117) only provide a momentary relief from the hard work of managing the feelings of others.

It is important to notice that doctors and nurses are not the only persons within medical relationships who make use of protective techniques in their emotion work to limit their emotional investment and to contain anxiety. Also patients and relatives make use of these techniques when they attend the health care system for help.

Lupton (2003b) criticises the contemporary movement of consumerism in health care that on one hand assumes an autonomous patient who is self-reliant, informed and reasonable and on the other hand assumes a doctor who is willing to

share his own feelings and thoughts with the patients and understand them as individuals in the medical encounter. She points to occasions where 'consumer patients' demand that the doctors engage in the patient-doctor relationship as a meeting between two equal individuals. From this perspective, patients tend to construct the doctor in an anti-medical light as the 'other', 'them' or the 'system' as part of a strategy to hem in their own disturbing thoughts and emotions. This kind of emotion management, Lupton argues, which works through the use of protective or precautionary strategies, becomes problematic in cases of serious illness, where people are especially vulnerable and anxious. Lupton finds that patients must have a certain degree of trust in their doctors' abilities and legal accountability for the medical encounter to be carried out successfully. Without this trust, patients might find themselves experiencing even greater uncertainty and anxiety than they already do. In situations, where people are genuinely ill they need doctors who comprise a 'reassuring' alliance in the 'mutual project of becoming well' (2003b: 169). Hence, a doctor's job belongs, according to Lupton, to a significant other service sphere than for example Hochschild's flight attendants. This stand point implicates that a so-called 'rational', 'observing' or 'bureaucratic' doctor might prove to be a better help to an ill patient than the service provider who attend her clients as a saleswomen, who offers value-free goods.

Working within the divide: The colonising of 'sincere' emotions in health care

Bone (2002) examines dilemmas of emotional labour in nursing under 'market-driven health care' based on qualitative interviews with practicing nurses in California. She argues that the invisible and often undervalued work of emotional support conducted by nurses is being displaced and transformed today within new structural arrangements in managed care organisations. She analyses how lack of time, increased work load and lower staffing become the biggest structural constraints to doing emotional labour. To meet the changing organisational conditions, nursing becomes a question

of mastering 'emotion standards of *detached involvement*' (2002: 146, my emphasis). According to Bone, this rationalisation of emotion work might subject nurses and patients to new means of disciplinary practices and co-optation. She sees 'genuine' therapeutic emotional support as the least accounted for dimension of nursing today. In this way, personal and therapeutic elements of healing may be hemmed in by technical defined skills. The danger is that nurses adapt emotional labour to new managed care contexts. One of the results of this adaptation might be that 'nano-second' emotionalities replace more traditional 'caring' nurse-patient relationships with great costs for both patients and the nursing profession. Bone argues that this implies for the latter, that nurses may experience tensions between empathic, personal concerns and demands for rational, organisational detachment.

Olesen and Bone (1998) argue that environments of structural and economic change in the American health care system entail alteration of codes of conduct. When patients become customers in a health care market, the emotional components of interactions change. The authors point is to demonstrate that changes in social and economic contexts influence emotional labour and the gamut of emotional expression in organisations. When nurses attend patients as customers, the discrepancy between codes of conduct (prescribed by a professional ethos) and demands in relation to what one should feel in a given situation generate problems. The authors suggest that these problems take the form of nurses experiencing 'ambivalence, shame or anger in discrepant situations' (1998: 134). The result is that nurses start to construct new emotional expressions to match the awkwardness of novel nursepatient interactions, where rules of feeling now resemble those developed in other face-to-face service jobs (e.g., flight attendants, shop floor employees, restaurant servants).

Following this line of thought, studies have explored how new forms of control in nursing, especially the use of consumer feedback or patient satisfaction as a

management control strategy, constrain nurses and force them to perform 'manufactured' emotional labour which they experience as alienating (Cooke, 2007).

Theodosius (2008) observes that nurses, like Hochschild's flight attendants, have very little time in current treatment regimes to do emotional labour. She argues that nurses' 'sense of *emotional authenticity* is lost in prescriptive puppet-like emotion expressions. This is distressing because emotional labour is an essential component of the need and purpose of nursing care' (2008: 47, my emphasis). In another text, Theodosius (2006) claims that she wants to 'rescue emotion from emotion management'. In her opinion, care consists of emotional components which cannot be managed or even articulated, which make these components all together unmanageable. Theodosius' work is a good example of how an interest in emotions in health care represents an opening for the nursing profession to reclaim its role as care providers, holding on firmly to a traditional emotional territory of nursing. Her stories mime the introductory story from my cancer clinic, where the nurse tried to monopolise emotions to nursing practice.

I mentioned in the beginning of this section on emotions in health care, that only few studies so far have been engaged in examining the role and function of emotions outside of the nursing profession, and in relation to this I mentioned that this was probably due to the assumptions many researchers have in relation to the 'caring' role of the nursing profession. These assumptions, as we have seen unfolded in the above discussion, may find inspiration in a theoretical grounding of emotions in the inner realm of individuals, in particular when drawing on inspiration from the work of Hochschild. One of the consequences is that the constructive nature of 'caring' is often mistaken for an 'authentic' caring personality, which lies within the individual (female) nurse. Another issue I will mention here is the fusion of emotions with gender. A persistent conceit exists through which men are associated positively with rational thought and action while women are negatively associated with emotional reactions.

tion. Recently, I read in my morning newspaper about a journalist who claimed that 'females are by nature both more empathic and more soft' (Dagbladet Information, 2011). This obviously false distinction between male and female feelings, which facilitates a gender division of emotions and emotion work, is also present in studies on emotions and professions in health care.

This latter body of research on the nursing profession is noteworthy in relation to its rather sentimental attitude to thinking about emotions in health care. Here, emotions are typically approached as 1) properties which have been colonised by the 'system', 'capitalism' or 'bureaucracy', or as 2) properties which have been lost in a current environment of managerialism and now must be brought back into practice. In other words, the literature tends to envision emotions as personal properties that can be used, manipulated and controlled in the hand of others – for example in the hands of the 'system', the 'organisation', the 'bureaucracy', and so on. From this follows that individuals must resist the organisation and protect their 'real' feelings from getting colonised by others. Emotions can be approached as a site of resistance or the last frontier from further standardisation and commoditisation. The perspective emerging from this division locates emotions in a private sphere where each individual has the responsibility of providing shelter from the outer realm. In retaining a theoretical grounding of emotion in the human being instead of in the social realm, the perspective provides support for the idea that emotions are isolated, individual entities. This polarised thinking presents an image of bureaucratic organisational life devoid of emotions, where 'authentic' emotional experience, which is embedded in the individual, is opposed to 'manufactured' or 'commercialised' feelings, that is dictated by 'the organisation'. Such accounts, I argue, may then continue to essentialise emotions and to sustain a dichotomous way of thinking of emotions in health care

The number of analytical challenges, which seem to accompany the interest of emotion in organisations, points to the importance of developing a nuanced theoretical framework which does not limit itself to a restricted focus on 'individual feelings' on one side and 'organisational rationality' on the other side. To bridge this divide, one therefore needs to leave the attention to authenticity, interiority and individuality of emotions behind. In the next section, I will suggest how we may approach emotions in a way that avoids this kind of conceptual entrapments.

A combined theoretical framework to the study of emotions in organisations

Time has come to draw from the various approaches, points and concerns that I consider fruitful for a balanced and nuanced approach to the study of emotions in organisations. This chapter has reviewed several major perspectives to address questions of emotion.

A possible way to summarise this chapter is to lend the word to Wittgenstein (1999 [1957]) for a short comment. In his investigation of philosophical problems, Wittgenstein criticises general explanations of inner feelings and private sensations, and he points to the circumstance that, for an utterance of an inner feeling to be judged meaningful, it must be possible in principle to subject it to public standards and rules of a specific context. Describing the form of a philosophical problem, he writes:

We ask "What does 'I am frightened' really mean, what am I referring to when I say it?" And of course we find no answer, or one that is inadequate. The question is: "In what sort of context does it occur?"

(1999: 161)

This rather simple example has two significant aspects. First, to be 'frightened' or to feel 'fear' is dependent on social situation or context. Second, though Wittgenstein

does not indicate it directly here, emotional expression and experience of 'fear' is grounded in possible historical continuities and discontinuities in a long-termed history of emotion codes or standards of emotional behaviour and restraints. The meaning of 'I am frightened' is therefore not fixed in stone, but is rather, to some reasonable extent, context varying. A contrasting position would search for the essence that lies behind the appearance of a phenomenon, such as seeking to discover the reality behind the bodily expression of fear. The 'fear phenomenon' would then be traced as an innate essence in individual biology and only secondly would one starts to situate emotions in the social realm.

One of the key characteristic of the sociological theory on emotions, that I have reviewed in this chapter is the stress on the contingent character of emotions, depending on variables such as context, particular situation, historical time, culture and trends. Instead of approaching emotions as universal stables, the literature emphasises the performative value of emotions in particular social arrangements. Following this perspective, it becomes possible to trace and diagnose contemporary forms of emotionality or changes in social conduct as depending upon complex historical circuits of individual, cultural and social relations. Emotional expression then follows from the social arrangements and power relations that surround it.

In order to sum up my inspiration from the sociological theory on emotions, I have listed some key points which have major influence on my exploration of questions of emotion in the cancer clinic and forms of professional conduct therein. First of all, some of the important learned lessons:

- Emotions are produced effects rather than essential dispositions
- Emotions emerge through assemblages of a larger scale (societal patterns, power structures and psychological make-up of sentiments)
- Emotions are produced through organisational codes of conduct and standards of regulation and affect-control

- Emotions serve as a function in maintaining social order
- Emotions can be worked at by individuals to produce a reasoned social
- Emotions are an essential part of professional work and as such a part of professionals ethical handling of individual person cases
- Emotions may enrich the attainment of organisational outcome

My own theoretical approach, as developed in the remaining chapters of the thesis, attempts to bring together many of the foci developed under the headline of 'emotions as historic, cultural and social constitutions'. This overall theoretical framework then signals an orientation towards constructivism. A constructivist perspective regards emotions as a series of relative constitutions, which are dependent upon the socio-historical settings in which they occur. These components are contested and reframed at the level of everyday practices of individuals and organisations. Emotions are shaped and experienced by human interactions; they are framed and reproduced through social practices. The perspective implicates that emotions are never seen as an independent reality, relying upon a universal vocabulary. Instead, it emphasises the plasticity of emotions in relation to their characteristic of being components in the production and co-production of social reality.

When one acknowledges that emotions are socially constituted, one also implicitly pays attention to the productivity of emotions, based on the simple argument that emotions both *are* practice and are constitutive of the possibilities for practice (Williams, 2001). By making this claim, it becomes possible to pay attention to what it is that count as emotions, which means to look at emotions as more or less specific practices, procedures, products and configurations and, not least, as concrete processes of social world making. If we understand emotions through this constructive/constitutive lens, as I do in the thesis, the point is not so much to debate the onto-

logical status of emotions⁶, but rather to turn the attention to how emotions and emotional expression become manifest in specific socio-material processes of organising.

Rather than repressing emotion as the 'other' of reason, this perspective on emotions demonstrates the mutually constitutive relation between reason or rationality and emotion. Instead of being irrational or an expression of irrationality, emotions function as cues and markers that form a necessary orientation system fundamentally enabling perception of reality and, ultimately, forms of rationality (Townley, 2008 – see also Jagger, 1989). The strength of the emotion perspective may therefore be its ability to transcend forms of dichotomous ways of thinking about rationality and emotions in social sciences. Furthermore, it might even provide the way to deal with a central sociological concern with regard to the relationship between personal troubles and public issues of social structure (Williams and Bendelow, 1998). The perspective includes asking questions about how individuals' activity and experience are interlinked with codes of conduct that are themselves historically and socio-culturally grounded. In this way, emotions are interlinked with and shape relations of power and the governance of social conduct.

My theoretical approach addresses the importance of emotions to basis purposes and values of health care organisations. This perspective emphasises, in line with other researchers, that emotions are no adjunct to work and the instrumental goal orientation of work life (e.g. Putnam and Mumby, 1993; Ashforth and Humphrey,

-

⁶ I find it necessary to mention a few, but important omissions here. First, I will not be defining 'feelings', 'passions' and 'sentiments', 'mood', temperaments' or explaining their relationship to emotions. It has the consequence through the text, that I make use of the words 'feelings' and 'emotions' as if they were one and the same. Second, I have more or less ignored the question of biology. Emotions have long been a 'hot' topic of affective sciences. In disciplines such as neuroscience, the search for distinct emotional systems and universal emotions (e.g., rage, distress, excitement, happiness) is developing. However, with reference to the constructivist/constitutive approach to emotions, I assume that emotions are not simply psychobiological phenomena, but are historically formed components of social action. Third, I will mention the discussion on affect and emotion as another important omission. I intend to use the two terms as if they were interchangeable. More than anything else, I think the uncompromising distinction in terminology – between emotion and affect – has more to do with intellectual heritage and the need to distinguish oneself from other groups of academics, than with actual conceptual difference. In this project, I am not concerned with drawing radical terminological lines, which make it difficult to engage with operational practices and lived experience of individuals towards which I have turned my attention.

1995; Fineman, 1993, 2006). In fact, many rational strategies of health care are pursued on highly emotional grounds and what we describe as rational, for instance in medical relationships, is in fact very emotional. The approach described here involves what I term a 'non-sentimental' attitude to thinking about emotions in organisations. This attitude embraces both emotionality and rationality in its conceptualisation of organisational activities, focusing on how management of emotions are formulated and formed through training and practices in particular contexts and circumstances. In this way, I attempt to stay clear of the analytical 'dangers' addressed in the very beginning of this chapter.

Chapter 4: To study emotions - Focusing, gathering and writing up material on emotions

Introduction

In the previous chapter, the management of emotions in work and organisational settings has been conceptualised through the thinking of a distinctive group of sociologists and through the review of emotion studies, particularly in relation to health care. The way emotions are approached in the literature has been shown to have an impact on how one understands emotions in empirical inquiries. This is methodologically significant since emotional conduct is difficult to observe and understand. For researchers who seek to emphasise the work that goes into the management of the emotions, it is useful to study emotions and emotional practices in organisations using a variety of research methodology, including ethnographic methods. However, before describing the methods and the amount of material used for this research, and before reflecting upon how the selection of methods enabled me to gather material on the research core topics, I first need to confront a more pressing matter. Namely, how did I come to define my object of inquiry in the first place? This is an important question, since the answer both leads back to an explanation of my personal kind of access to the researched organisation, and points forward to a methodological discussion on how to study emotions in work places. Therefore, this chapter begins with a description of the situatedness of my PhD project and my access to the cancer clinic and its occupants. Next, it explains how I defined my object of inquiry. Then, it moves on to reflect on how to study emotions and discusses the problems involved in this process, including problems of *detachment* and *involvement* in social science studies. Finally, the chapter recollects how writing up material on emotions becomes a reflexive act of generating emotional conduct in organisational life, and how this writing process may also be approached as an affective separation from the field of study.

The situatedness of the PhD and my access to a cancer clinic and its occupants

In this section, I will describe my relations with and access to a cancer clinic and its occupants. I will do this through an explanation of the situatedness of my PhD project, because this situatedness also determined the access to my field of inquiry. Furthermore, I will explain how my type of access both became a problem as well as a solution to defining my object of inquiry. The personal biography provides further means to discuss how research on emotions in health care comprises certain methodological challenges.

As previously mentioned, a major centre⁷, located at the university hospital in Denmark, where I also conducted the research, has funded a larger part of my PhD project in a partnership agreement with Copenhagen Business School. Before beginning this study, I worked four years at the hospital as an internal consultant in the centre's administrative unit. Originally, I had started working in a Cochrane Collaboration unit⁸. Looking back on this job experience, I think my current occupation with soft dimensions of medical care somehow counter intuitively derived from this initial work with golden standards of medical research methodology, such as clinical algorithms, meta-analyses and Cochrane reviews. I hope to shed light on the rationale behind this reflection as the chapter develops. On the basis of an educational background in the humanities, I was later employed in the centre as a consultant in communicative affairs to support both the heads of the centre and the centre's clinical staff members in order to improve the external and internal communication with political institutions, clinical collaborators, patients and lay people in general. The job title provided access to what I like to think of as the hospital's engine room. My perception of this engine room comes close to Goffman's (1959) dramaturgical under-

⁷ In Denmark, university hospitals are typically structured into medical and surgical treatment centres, diagnostic centres and administrative centres, hold together by a managing committee.

⁸ The Cochrane Collaboration is an international network where its members work together to develop, update and promote systematic reviews which are based on evidence based research methodology. The reviews are published online in a database which is called *The Cochrane Library* (see also www.thecochranelibrary.com).

standing of a 'backstage' region, which in his terms is an organisational region where:

'[...] the team can run through its performances, checking for offending expressions when no audience is present to be affronted by them; [...] poor members of the team, who are expressively inept, can be schooled or dropped from the performance. [...] the performer can relax; he can drop his front, forgo speaking his lines, and step out of character' (1959: 115).

One can compare the hospital's engine room to the outpatient consultation room that a patient attends if he or she for instance needs to be tested for hypertension. The latter region gives the person access to a concrete work setting where particular actions and roles are performed. This region is what Goffman terms the 'front region'. However, the front does not immediately give access to the place where the hospital organisation typically constructs its 'illusions and impressions' (1959: 114). I mention this here, because through my 'free access' to the hospital, I not only had access to various confidential material on the hospital organisation, such as policy-making reports, standard contracts for surgical and medical treatment units, financial overviews and strategic visions for future treatment developments. In this engine room, I also had access to staff members' reflections and discussions about how they experienced, thought and felt about doing different kinds of work at the hospital; how they felt about changes, and what they thought about new managerial attempts to regulate their conduct. When I changed status from consultant to PhD researcher, the scene of my everyday life shifted from the hospital to the university, but a very useful peephole remained open into the hospital world. The consequences of this kind of access are worth some reflexion because it may have affected the way I came to approach the front regions of the hospital.

A deal was set up. When I signed the PhD contract with Copenhagen Business School, I also signed a contract with my former work place. As part of my PhD contract, I agreed to work eight hours a week at the centre. During the contract period, I was thus enjoying a weekly workday at the hospital, where I frequently went to meetings with administrative colleagues and meetings with clinical staff (doctors, nurses, secretaries and dentists). Parts of my work schedule lay in extension of my previous consultant tasks within communication, and I also continued to rely on my personal contacts. I had previously collaborated with some of the doctors and nurses from the cancer clinic. I had met the clinic's head of research (the person was also the former head of clinic) through a common involvement in developing the centre's research profile. This person presented me to a research group at the hospital, who was interested in breast cancer illnesses and their treatment, including diagnostic methods, surgical techniques, nursing care, exercise and early recovery, anaesthesia, pain and vomiting medication. The group was led by a concern for ways to improve interventions to make patient trajectories both more effective and more efficient. The meetings in the group whetted my appetite for a sociological investigation of accelerated medicine. I found that new ways of organising cancer illnesses at the hospital were not without consequences for the conduct of medical practitioners in their day-to-day work. I will return to this issue later in the chapter.

When I decided to do a PhD, the head of research granted me official research access to the cancer clinic and its occupants. He also introduced me to the staff members when I later arrived on the ward.

At the hospital, I had my own desk and my own computer and a sign on the office door with my name on it. In the beginning of the PhD, my consultant job was to facilitate change processes in the centre, especially in relation to processes that had to do with the merger of clinical units. However, as I got more involved in my fieldwork, I needed to withdraw myself from these process facilitation tasks, and from then on, I spent most of my weekly workday writing official documents, newsletters,

web texts, and etcetera, for the heads of the centre. Through this preoccupation with merely administrative tasks, I was being legitimately sheltered off from the clinical environment. The argument for this separation was twofold: I did not have the necessary time to engage myself satisfyingly in the work. And the more I got involved in my fieldwork, the more it became necessary to separate the two job functions; i.e., to separate the function as consultant from the function as PhD researcher. The last couple of months of my project, I had almost entirely stopped working as a consultant in the administrative unit, and the weekly workday at the hospital was mostly spent writing up notes and talking to colleagues about dimensions of my project. Three and a half month before handing in the thesis, my contract with the centre came to an end and I stopped working at the hospital altogether.

These observations hopefully explain that not only did I sign a physical contract with the hospital when I started my PhD project. I somehow also signed an emotional contract with the clinical environment at the hospital. The fact that emotions are not epiphenomenal but are part of one's job is illustrated well in studies on emotion, work and organisation, as reviewed in the previous chapter. The fact that people are expected to 'give themselves to their work', as for example Boltanski and Chiapello (2007: 98) argue in their analysis of flexible capitalism, also involves that the work organisation may employ one's emotions both through the work contract and through the formal and informal work activities it demands of its employees.

In my case, this 'emotional contract' meant that I felt attached to my work place; I identified with my colleagues and the clinical staff members and their routines, and I felt committed to defend the hospital's precarious ways of doing things when this was demanded of me. In relation to my employment as a consultant, this attachment probably added some value in relation to my personal qualifications, especially when considered from the managers' perspective. However, the value of this attachment was questionable when it came to my function as a researcher. For instance, during my consultant years, I had participated in some work of streamlining

surgical procedures into so-called 'speed-units'. From an administrative/economic perspective this was an interesting experiment because by changing the time tabling of the operation ward, one may succeed in changing some sacred clinical rituals: an earlier morning start in the operation ward meant that the involved doctors could not participate in their morning conference in the clinic. This example shows how I in my previous consultant job reproduced attempts of controlling the medics through new forms of governance techniques. It was now the self same attempts, I wanted to scrutinise in my research. But in order to thoughtfully problematise these procedures and to be able to observe them from a detached, sociological perspective, to see them from a safe distance where I did not share solidarity with either administrative or clinical concerns, it was extremely necessary for me to exit the hospital's 'engine room'.

This described difficulty of detachment from my former occupation also affected how I came to define my object of inquiry through taking some serious detouring routes. I will next explain the routes to the development of a focus on emotions in accelerated medical work. Subsequently, I will return to the issue of emotional attachment in relation to social science studies.

Defining my object of inquiry: From foci to a developing focus on emotions

The grounding of this PhD project was from the beginning attached to an empirical interest in accelerated medicine and the introduction of accelerated treatment regimes in the Danish health care system. The concept of accelerated trajectories was described in 2007 in an influential policy report, which was published in relation to the Danish Government's launching of a quality reform of the public sector (see The Danish Ministry of Health and Interior, 2007). The concept was emphasised as 'an important experience which needs to be systematically implemented at all hospitals in Denmark' (2007: 94). On behalf of the Danish Government, the National Board of Health developed from 2005 to 2008 the so-called 'cancer packages', which are stan-

dardised, accelerated cancer trajectories, which soon afterwards were implemented throughout the health care system. I mention these things to explain how the PhD project was being fostered by specific changes to the organisation of medical work that took place while I was employed at the hospital as a consultant. The heads of the centre, where I was employed, were interested in gaining knowledge about how patients experienced the restructured trajectories. Due to a 'patient-centred' concern, they were also interested in how doctors could make themselves 'more' and 'better' available to cancer patients and their relatives in relation to the psychological and social problems associated with cancer illnesses and their treatment. These problems, which were addressing cancer patients' psychosocial health needs, have been documented in larger patient surveys and reports (see for example KB, 2006; Region H, 2008).

While working on different projects in the administrative unit, I did some work together with a breast cancer clinic at the hospital. The clinic was generally described in the administrative unit as extremely well-organised and highly skilled in its ways of attending to the needs of its patients. The clinic's ways of organising cancer illnesses and their treatment in standardised trajectories were also by the heads of the centre thought useful for other clinical units at the hospital. Hence, I spent some work time writing up the clinic's experiences and collecting various materials on accelerated medical work for merely administrative use. This initial gathering of material and the contact to the people who were a part of these trajectories became a valuable help when I was enrolled as a PhD, because it gave me a platform from which I could depart. It also gave me a research site at hand: a cancer clinic and its occupants. However, what this material did not help me with, was defining my precise object of inquiry and I will next explain my difficulties in doing that.

Broadly speaking, the cancer trajectories are, among other characteristics, defined by 'standardisation', 'quality', 'speed', 'accountability', 'audit' and the aim of meeting various 'performance' and 'measurement' targets. These key charac-

teristics led my attention to literature on public performance measurement (see for example Power, 1997) and Post-Foucauldian literature on governmentality (see for example Osborne, 1993; Dean; 1999; Miller and Rose, 2008). Following this line of thought, I wanted to explore the introduction of 'cancer packages' as a new form of governance, that, and here I was inspired by Osborne, entails liberal ways of organising medicine (Osborne, 1993: 345). I understood the latter as a new alliance of political, clinical and managerial concerns which aims to insert economic initiatives in various aspects of clinical practice to improve the output of health services⁹. The 'cancer packages' provided a good example of this tendency. When I started my fieldwork, I started to apply this approach not only to the documents that I had previously gathered, but also to the people I interviewed as part of my fieldwork. Specifically, I went to see people who were preoccupied with the development of cancer trajectories. Among these preliminary interviews, I remember talking for three hours with a retired professor in surgery, who came to visit me at Copenhagen Business School in November 2008. He introduced me to the historical development of breast cancer illness and its medical treatment breakthroughs from 1970 to today. At the same time, I began to reread 'The Birth of the Clinic' by Foucault (1973), which is a brilliant book about how the 'clinic' emerged as a site for learning and curing practices in the eighteenth century, and how this new organisational form of the hospital brought certain objects into focus, such as the (patient) case and the observing person (doctor). I was interested in how the role of the latter may have begun to take on new dimensions since Foucault made his observations - from that of Foucault's observing person to a more personal or subjective person.

During autumn 2009, I started making observations in the cancer clinic. I observed doctors when they attended patients, relatives, colleagues and others in the clinic and its surroundings. I observed interactions in the outpatient clinic, operation

⁹ See Roelsgaard Obling, A. (2010) Pakkede patientforløb og styringsambitioner i det danske sundhedsvæsen, In Kjaer, P. and Reff, A. (eds.) *Patienten i centrum*. Copenhagen: Copenhagen Business School Press, pp. 101-129.

theatre, inpatient ward, and staff meeting room, corridors and doctors' private offices. When the doctors had conference meetings with their colleagues or attended patients in the outpatient clinic, I sat in. If they had coffee with colleagues and discussed their experiences, I would go too. I accompanied them on rounds in the clinic's ward and sat in on conversations with newly operated patients or worried relatives. When the doctors used electronic devices such as computers to access different databases or safety report systems or recorded patient stories with help of a dictating machine, I watched and listened. The choice of doctors being followed was not based on any particular criteria. Some doctors were suggested to me by the head of the department, others because of their availability on my observation days.

A central concern was to specify the aspects of medical work in the trajectories I actually wanted to focus on. Still struggling to make this decision in this phase of the project, my observations were unsystematic in a way that they surely discovered many aspects of interaction and everyday behaviour of the clinic and its occupants; however, in trying to cover 'every' aspect of medical work, I accomplished nothing. Should I focus on the production of medical knowledge in the trajectories? Should I keep the attention on clinical governance and audit practices? Or should I look at certain kinds of social interaction and for instance explore how doctors frame deviant patient cases or 'misfits' to the standardised procedures in the trajectories? Instead of confronting these difficulties, I decided to spend four month at Goldsmiths in London. Due to the lack of focus, my contact person (and later cosupervisor), Monica Greco, and I discussed some possible ways to frame the questions my fieldwork raised. She suggested that I looked into the literature of medical humanities or narrative based medicine (see for example Montgomery Hunter, 2001, 2006) to help me capture some of the different aspects of medical work the doctors in my trajectories were called upon to perform. This is literature which is often written by doctors who have turned to humanities to understand the more artful character of medical practice and organisation. This artfulness includes the communicative, emotional and personal aspects, which are part of doctors' job when they attend to the needs of patients in medical encounters. Generally, doctors' skills of intuition or talent, certain professional behaviours (detachment, empathy), and their mastery of practical procedures are described in this literature as the 'arts of medicine' (see for example Nettleton, Burrows and Watt, 2008).

Coming from evidence based medicine and standards in cancer treatment to new modes of governance in health care to these more intangible aspects of medical work surely expanded my view on clinical practice. I began to reread my field notes in this light, paying particular attention to the character of the different kinds of work processes I had observed. Indeed, I found this new focus on the various conducts of doctors inspiring because it provoked a reductionist, EBM perspective, which merely sees doctors as applying scientists, whose job is to remove uncertainties and individual humanistic concerns from clinical practice (for a 'purified' form of this argument see Baum, 2007).

I became preoccupied with questioning the intangible aspects of medical work from a governance perspective, which sees medical practitioners as constrained by standardised, corporately framed diagnostic and prescribing procedures, and I began to explore questions like: How do we study these intangible aspects? How are these aspects reframed by the medical practitioners in ways that make them available for external observation? And how may these aspects become visible objects of inspection?

A small breakthrough came when my co-supervisor, after yet another discussion of my fieldwork exclaimed: 'It seems to me that the only elements which are left to orchestrate in your cancer clinic are emotions'. We had previously discussed what actually separated the doctors in my cancer clinic from Hochschild's flight attendants in Delta Airlines, so the emotion perspective was not an entirely new perspective to me. In light of this discussion and of my previous interests, I decided to focus my inquiry upon emotions and emotional aspects of medical work in the cancer

trajectories. Greco introduced me to her systematisations of the field and helped me to find my own place in emotion studies. When I later returned to Copenhagen, long conversations with my supervisor, Signe Vikkelsø, stimulated me to extend the range of my thinking. Especially her continuous pin-pointing of the importance of detailed empirical descriptions, inspired me to look into emotions as they anticipate in concrete practices in the cancer clinic. This likewise helped me to understand how emotion work takes place alongside other aspects of medical work; each aspect mixed up with one another, and each aspect equally necessary for the line of medical work in the cancer trajectories to be carried out successfully.

The shift from various – way to many – foci to a developing focus on emotions induced alterations in my methods and material. The earlier work generally assembled two types of evidence: 1) material on changes in the organisation of cancer treatment – the character of speed, the value of evidence, the methods of accountability, and so on – and 2) unstructured observations of practices, such as doctors ways to relate to principles of evidence based medicine in their daily activities. In the further process of defining my object of inquiry, the developing focus on emotions included that my methods and material increasingly focused on: 1) the description of how and when emotions occur as social components in practice and for what purposes, and 2) the examination of how doctors experience and understand emotions in various parts of their conducts. As a result, the inspiration from literature on 'performance measurement' and 'audit cultures' transmuted into a concern of locating contemporary demands for certain emotions and emotional expression within their specific social contexts. In this way, attempts to govern the conduct of doctors through various techniques – for instance through audit devices – were linked to the issue of how doctors anticipate more or less rationalised forms of socially sanctioned explanations and ambitions

Gathering material on emotions

In relation to the refinement of my object of inquiry, the next step of my fieldwork was a series of 14 semi-structured interviews with doctors in the cancer clinic. Compared with my previous explorative interviews, which aimed at gaining knowledge of the formal organisation of accelerated cancer pathways, in this interview series I wanted to talk with the doctors about their personal (i.e. individual) understanding of emotional aspects of their work in the pathways. Concurrently with my own empirical research, I had been working with a colleague, Nanna Mik-Meyer, on an article based on a larger qualitative interview study, concerning how general practitioners (GPs) approach patients with medically unexplained symptoms (MUS). Our main argument in this article was that GPs not only use traditional biomedical diagnostic tools when attending patients with MUS, they also rely on their personal opinions and evaluations of a patient's particular circumstances in deciding whether the patient is legitimately sick or not (see Chapter 8 for the entire argument). Due to these findings, I was riveted by the ambivalence of emotions in one of the core practices in medical 'truth making' practice; namely medical diagnosis. In extension of this article, I wanted to explore in further detail how doctors in the cancer clinic experience emotions and the way emotions form part of their professional conduct. Current ways of organising technical aspects of medical work in the trajectories may also affect the organisation of emotional transactions in the trajectories, and as a result affect how doctors experience and understand emotions, or so was my hypothesis.

Following a constructivist perspective on the study of emotions, I had no intention of trying to trace what the participating subjects 'really' felt or to determine the authenticity and 'under-the-surface' feelings of the informants. Instead, I was interested in their presentations of personal biographies of emotional experience and emotional challenges. My initial interview guide was constructed so as to provide an understanding of doctors' views on and responses to their work life, and to understand how these people frame and reframe emotional injunction in their work. I was

interested in 'how it feels being a doctor'. Characteristic of a constructivist inquiry, the attention in the interviews was focused on the activeness of the interviews. This is not a new catchword to qualitative interview methods. Holstein and Gubrium (1997, 2002) argue in their work on interview method that all interviews are active interviews, hereby emphasising that the interview is not so much a neutral conduit or a knowledge container as it is a unique site for producing reportable knowledge. The constructivist phrase 'all meaning is socially constituted' (Berger and Luckmann, 1966; Garfinkel, 1967) gains momentum in active interviewing method, because it here becomes obvious that meaning and meaning structures are actively assembled in the interview encounter through interviewer and respondents' collective accomplishments. The emphasis on process in the interviews also included that the questions I asked in the interviews varied a little from interview to interview because the respondents conceived the questions asked differently. However, except differences in respondents' ways of answering questions, I tried not to lose too much track of the research topics guiding the interviews and the content of my questions, and how these were understood and communicated by the respondents. All interviews (with exception of a few clarifying interviews in the beginning of the research) have been audiorecorded and transcribed. Shorter interviews during participant observation were not recorded. Neither were more spontaneous conversations. In addition, notes on some of these conversations were written during the fieldwork.

I also revisited the doctors in the clinic. When emotions are approached as contingent social components, depending on social situation, context and historical time, then the ability to understand emotions at work is highly depending on methods to identify and describe how the conduct of doctors is organised in specific situations. At that time, I found fieldwork in the clinic to be a reliable choice of method because in order to observe how individuals can both 'affect' and 'be affected' in social interaction, and to recognise the complex interweaving of emotion and reason which makes up the course of doctors' job, I needed to be part of the context. Importance

was hence given to direct 'in situ' observation of concrete sequences of activities in the clinic (see also Goffman, 1989; Baszanger and Dodier, 2004). However, compared with my previous fieldwork, I now had my attention focused on the situations in which doctors offered emotional services or responses aiming to facilitate patients in particular situations. In other words, I paid attention to how doctors managed emotions in relation to patients, relatives – and not least themselves – in the cancer clinic.

To sum up, my fieldwork in the cancer clinic was spread out over almost two years from June 2009 to January 2011.

First, I talked with several people to gather material on the historical development of the accelerated cancer pathways. In addition to these interviews, I later interviewed doctors in the cancer clinic to explore more directly how they experience and understand emotions at work. The appendix shows the entire list of interviews (see Appendix A). The interviews are indexed with anonymous names and I refer systematically to those names throughout the thesis.

Second, I made observation of a training workshop in 'empathic communication'. I have field notes from this workshop with doctors and nurses from the cancer clinic, and I have three transcribed interviews with heads of the cancer clinic and the consultant who facilitated the workshop. In addition to this material, I have a collection of documents from the workshop (invitation letters, teaching material, work shop presentation). From my previous employment at the hospital, I have material from workshops of a similar kind, where doctors were trained in having difficult conversations with patients – for instance conversations about poor diagnosis or future health perspectives.

Third, I have observed lots of encounters in the cancer clinic (conferences, operation theatre, ward rounds, and etcetera) and I have spent 15 days observing patient-doctor interactions in the cancer clinic's outpatient consultancy rooms.

The appendix shows the type, location and frequency of my observations (see Appendix B).

Finally, I have listed all the official documents I have used through the thesis in the appendix (see reference list). The document material include policy documents, clinical guidelines and practice recommendations, strategy reports, patient satisfaction surveys, medical textbooks and practice journals and locally circulated information material such as letters and e-mails. Material is also drawn from attending both academic and practitioner conferences on the organisation of cancer illnesses and their treatment.

Starting out from various foci on medical work to a developing focus on emotions, I began more systematically to ask the question: *how is emotion framed, trained and performed by doctors in accelerated medical relationships*?,

Each chapter or article in the thesis takes as its point of departure a smaller part of the fieldwork, and each chapter deals in particular with one of the above raised concerns. Chapter 5 deals with policy documents gathered in the early stages of the project that inset emotional injunctions to doctors and I analyse how doctors in the cancer clinic reframe these injunctions. Chapter 6, which is preoccupied with analysing how certain emotions are trained in the clinic, draws on material from the training workshop. Chapter 7 and 8 are both preoccupied with a concern of how emotions are enacted by doctors. However, the two chapters take different material as their point of departure. Chapter 7 relies on the small semi-structured interview series with doctors in my clinic, while chapter 8 relies on a larger interview study with GPs in the primary health care sector. Even though the chapters vary in their ways of engaging with the fieldwork, they also overlap and – that is the aim – enrich each other. Of course this sometimes means that they appear repetitive, which I think is one of the article-based thesis' main structural challenges and therefore not unique to my particular thesis.

I think every researcher is occupied with defining when his or her material collection is sufficiently satiated to successfully carry out the research project. In his writings on 'intellectual craftsmanship', Wright Mills (1959) argues that, once one has decided upon a topic and has entered it, one does not need to 'study' it, because suddenly it will appear to be everywhere. I experienced this insight in relation to my process of gathering material, because suddenly also apparently unrelated issues emerged as casting new light on the research and its discussions. For example, it occurred to me during an academic seminar that fellow researchers within organisation studies have recently started to argue that we, as researchers, should strive for putting back feelings into our research methods. I noticed that personal characteristics of the researcher, which are closely tied to emotions such as 'authenticity' or 'sensitivity' or even 'love', were brought forward by some of the participants as a methodological solution to an apparent 'coldness' or 'detachment' of social scientific inquiry. Some of the arguments, which were brought forward in these discussions, had a similarity to the arguments of 'more emotionality' in relation to the conduct of doctors in the cancer clinic. The two different settings – that of cancer treatment and that of social science research - thus seemed to share a language of emotions, in which some emotions were approached as more gratifying than others. At the time, I remember that I found these arguments strange. Anyhow, they deserve attention in relation to some methodological problems concerning involvement and detachment that I have faced in this study of emotions. I will next turn to this issue.

Problems of involvement and detachment in the study of emotions

One cannot say of a person's outlook in any absolute sense that it is *detached* or *involved* (or, if one prefers, 'irrational', 'objective' or 'subjective'). Only small babies, and among adults perhaps only insane people, become involved in whatever they experience with complete abandon to their *feelings* here and now; and again only the insane can remain totally unmoved by what goes on around them. Normally adult behaviour lies on a scale somewhere between these two extremes' (Elias, 1987 [1956]): 3, my emphasis).

In an essay concerning knowledge in social sciences, Elias (1987) discusses some methodological difficulties in relation to how the researcher should be engaged with her object of inquiry. In relation to this, he discusses the terms 'involvement' and 'detachment'. The terms do not refer to two separate sets of human attributes, such as one psychological or emotional and the other scientific or rational in character. Instead, he argues that one cannot separate the two phenomena and he denies that any sane adult could be either wholly involved in or wholly detached from what goes on around them. Thus, according to Elias, it is not possible to obtain any ultimate detachment or complete objectivity in relation to studies of social arrangements. What Elias draws attention to here, is the double character of demands to researchers within social science studies. On the one hand, the researcher's ability to sustain affective neutrality and detachment is crucial for carrying out her scientific work. On the other hand, the researcher's sense of sensibility and her affective experience of particular situations and participants, contribute to the production of the knowledge of social processes which comes together in the research. I think this double character is worth some consideration, especially in studies like mine in which one of the aims is to redescribe the public sphere as saturated with emotions. The methodological

challenge consists in doing this from an 'affectively neutral' stand point and without being 'gripped' by emotions. And how does one manage this?

To keep a distance from taking over the statue-like figures of 'involvement' and 'detachment', as being radically in opposition to each other, I have tried to study emotions: 1) without sentimentality, and 2) with sensibility to the performative value of particular contexts. In making this kind of methodological vocation, I have foremost tried to stay clear of any sentimental approaches of either bringing back emotions into medical practice or of protecting certain emotions in this practice. This does not mean that I have intended to replicate historical attempts at separating reason or rationality from emotion in social sciences, where a foregrounding of the former tends to silence the characteristics of the latter and hence leave it out of empirical studies in the context of work and organisations. Instead, I have paid attention to the relatively contingent character of emotions, consisting of the dependence upon variables such as social situations, historical time, and institutional context. To study emotions without sentimentality is easy to say but difficult to invoke in practice. Because what and whose emotions do we, as researchers, operate through when we attempt to categorise some practices as more emotional than others or when we attempt to highlight a particular emotion in empirical descriptions to the disregard of other emotions?

In a previous section on 'approaching emotions', I referred to Wittgenstein's explanation of how we may approach emotions as 'meaningful emotions' in relation to the expression and understanding of emotions in social life. The core of his argument was that the role and function of particular emotions vary across contexts. From this followed that emotions are of many types and may have different intensities in human conduct. To take a glaring example, in a novel, the American novelist Raymond Carver (1995) discusses the category 'love'. The title of his novel is perhaps not surprising: 'What we talk about when we talk about love'. He describes an interchange between two married couples and their discussion of 'love'. He writes:

'There was an ice bucket on the table. The gin and the tonic water kept going around, and we somehow got to the subject love. Mel thought real love was nothing less than spiritual love [...] Terri said the man she lived with before she lived with Mel loved her so much he tried to kill her [...] I picked up Laura's hand. It was warm, the nails polished, perfectly manicured. I encircled the broad wrist with my fingers, and I held her' (1995: 138-139).

'Love' is definitely an emotive word. In the above written quote, it conveys something of the author's experiences. However, the statements, 'real love was nothing less than spiritual love', 'loved her so much he tried to kill her' and 'I encircled the broad wrist with my fingers, and I held her' also give texture to the meaning of the individuals' feelings of love. The first sentence presents a transcendental definition of love, while the next sentence presents a more extreme, pathologic version of the word. The last sentence is somehow placed in-between the two previous sentences. On one hand it presents a physical expression of love. On another hand, it contains some uneasiness, especially uttered in the last four words 'and I held her'. In offering these interpretations, the reader is no blank sheet. It is precisely by these reconstructions that the signs of the sentences are imputed with a certain kind of emotionality. Hence, our own identity or biography matter in emotion inquiry. The researcher's emotion work is both a part of the approaching and the validation process (Fineman, 1993; 2006).

The relative embeddedness of researcher and the researched objects in emotion inquiry is not only a mean to reflect upon the nature of emotions. It is also a mean to address a political concern. Because whose emotions do we actually resonate with when we ask questions of emotion in our research? The significance of the expression of certain emotions may be attached to a moral labelling (Goffman, 1990b).

In one of the articles in my thesis, I draw on an example of a patient who 'expressed too much aggression' and a patient who 'was cold and did not express any emotion at all'. The doctors in my clinic presented these examples as examples of deviant emotional outbursts and hence as not appropriate for the representation of a 'patient'. Other sources, for example newspaper articles, public statements from patients, and debates in psychiatric circles might represent these emotional events very differently, depending on interest. If we take emotions to be shaped through social processes of all sorts, then we are necessarily confronted with 'many possible voices to represent feeling and emotion' (Fineman, 2006: 688). As emotion researchers, we must therefore try to avoid foregrounding a particular language, for instance a language of 'sentimentality' and pay attention to how the framing and enactment of emotions also are a kind of moral currency.

The above account has introduced some of the problems of conducting research that is sensitive to claimed characteristics of emotions and to altering contexts. I have shown some of the general problems in emotion research (focusing, gathering, appreciating emotions). At the same time, some specific research problems in the context of my personal biography were highlighted, such as my role as researcher and my problems in relation to involvement and detachment with my field of inquiry.

Writing on emotions

In relation to the above considerations on 'involvement' and 'detachment', I acknowledge that I have my own emotional investment in this PhD project, even though I have tried to stay clear of the spirit of a 'passionate sociology' (Game and Metcalfe, 1996). My methods have definitely generated ways of talking and writing about and presenting emotions in the scrutinised contexts and my involvement has contributed to a production of an awareness of social constraints in specific interactions. However, research and writing activities are always a subjective matter, and as I have tried to show in this chapter, my own preoccupations and interests are inflected

in my experiences and form part of the reflexive act of generating insight through the process of writing up the thesis. Whenever I discuss emotions in the thesis, the reflections are thus a mixture of personal experiences, other peoples' writings on the topic, and of course, the material I have gathered in the fieldwork.

In the first section of this chapter, I described my access to the cancer clinic and its occupants, and I described the chronological process of becoming detached from my field of inquiry. Not only did the physical process of distancing myself from the hospital help in that matter. Also the process of writing up the thesis was a useful detachment procedure. In one of his essays, Elias (1987) has a picture of a fisherman who is captivated in a maelstrom and thus involved in, what he terms, 'a critical process which at first appeared wholly beyond his control' (1987: 46). I use this analogy to reflect upon my own difficulties in relation to my access conditions and my involvement with the hospital. Elias continues his essay by explaining how the fisherman in the beginning 'clutches at some imaginary hopes. Fantasies of a miracle, of help from unseen persons may have crossed his mind'. However, this does not change a thing. First when the fisherman realises that he must stand back and turn his thoughts away from himself to the situation in which he is caught, he manages to change the situation and then 'he began to think more coolly' (ibid). I think this insight is important in relation to my own experiences with affective separation from the field of study. I am not less emotionally involved today, than when I started out writing this thesis. Neither am I less engaged. However, an ability 'to think more coolly' about things have slowly emerged during the research process and the additional writing process, that have detached me from the activities and the people I formerly was a part of. This process of detachment also implied that I eventually started to make sense of the things I had observed and to ask focused questions to the empirical realities I had experienced. Hence, this emerging self-awareness also helped me to refine my object of inquiry. The thesis presented here is a result of this process.

One thing is how my methods and my process of detachment have helped me to refine my object of inquiry. Another thing is to explain how this object is presented throughout the rest of the thesis.

With regard to ethical concerns, it should be mentioned that while the participating organisation has co-financed my thesis, it has been difficult to provide it with total anonymity. As far as possible, I have tried to conceal the identity of the local cancer clinic and its surroundings, especially in the four articles, where the precise naming of these institutions did not have any function anyhow. Of course individuals – professionals and administrators as well as patients and relatives – have been given full anonymity throughout the research. Denmark is a small country. The professional cancer community is equally smaller and busting with interconnectedness. Thus, the most general identifiers have been used to identify the source of the quote. I have followed the British Sociological Association on how to conduct research in an ethical responsible way. This includes general rules for participant acceptance, and procedures for processing and storing of data. It also entails recommendations for how to communicate the result of this research in appropriate ways in journals and public media. Apart from rules on ethical responsible research no formal ethical approval to conduct a research project like this has been required in Denmark.

In relation to practical concerns, I have translated all fieldwork citations from Danish to English. This implies that translation of the participants' spoken language in some places has been reduced in relation to its vividness and its degree of detail. However, I have strived to maintain the original meaning of the citations, and I hope that the language deficit does not make the meaning of my translations too ambiguous or unclear to the reader. In relation to writing a thesis, which is situated at a hospital, I have paid attention to learning some of the idiosyncrasies and vocabulary of the doctors present in the study. In situations where I have had difficulties to un-

derstand a medical terminology, I have turned to people at the hospital or doctors in my circle of acquaintances for help.

In relation to accountability concerns, I have clearly marked throughout the thesis, and especially in the four analysis chapters (Chapter 5–8), when for example utterances from my material are directly used as empirical evidence or when I have reconstructed these utterances as reconstructions of realities. Finally, which brings me to the end of this chapter, throughout my work I have tried to make transparency between my own engagement in the field, the methods involved and the process of generating and redescribing social arrangements in this field.

Chapter 5: The 'compassionate' doctor - Emotional injunctions to medical professionals in accelerated medicine

Abstract

This paper focuses on emotional injunction that programs of 'accelerated medicine' impose on medical professionals. The paper explores recent reforms of the public health sector as modes of clinical governance that encourage doctors to become more emotionally available to patients. It further examines how these reforms inset increasing demands on doctors to manage the emotions of patients in recordable, measurable and standardised ways. Empirical attention is drawn to descriptions of accelerated cancer pathways and how these descriptions anticipate emotional injunctions to doctors, such as injunctions to doctors to exhibit 'empathic' engagement or 'authentic' engagement, while they respond to the needs of patients. An Eliasian perspective on the management of emotions in work and organisational settings is used to make sense of the injunctions and to capture how the conduct of individual professionals is interwoven with recent reforming drives. The perspective is furthermore applied to debate processes of formalisation and informalisation in relation to the ways people are expected to behave in medical relationships.

Keywords: Emotional injunctions, accelerated medicine, doctors, Elias, public health care sector reforms.

Introduction

Over the last two to three decades a body of work has emerged on the impact of the wave of public health sector reforms in Western societies on the medical profession (e.g. Dent, 1993; Fitzgerald and Ferlie, 2000; Doolin, 2002; Harrison and McDonald, 2008; Waring et al, 2010). The impact of these reforms has been explored by critically questioning the role of New Public Management and principles of quality, au-

diting and performance measurement that attempts to make medical treatment more effective. Ewan Speed (2011), for instance, has argued that recent reforms in the UK National Health Services (NHS) have led to significant changes in regard to governing medical work. He introduces how these reforms entrench organisational forms, which 'dominate and regulate health professionals through less and less visible, but more and more pervasive command and control mechanisms' (2011: 101). One of these mechanisms of control is manifest in the reforms' emphasis on 'quality' or 'quality improvement' (2011: 94). As a mode of governance, the emphasis on quality involves both a regulation of instrumental aspects of clinical work procedures and a cultivation of 'softer' relationships, i.e., a control of social and psychological attributes of working subjects. The reforms install a political aim to treat 'the users of services as queens not pawns' (2011: 99). As 'queens', patients must be attended by engaged doctors, who invest themselves on a subjective level in the delivery of health care services. Likewise David A. Buchanan and Louise Fitzgerald (2011) demonstrate how a new generation of performance and measurement targets, aimed at improving quality and efficiency, includes demands on professionals to exhibit 'compassionate care' (2011: 72). They ask rhetorically how the delivery of compassion 'is to be measured, by whom, how often and where, and against what standards and benchmarks' (2011: 73). Compared to the delivery of traditional medical services – for example to take someone's blood pressure, the measurement of emotion conducts involves yet unsolved problems.

If we expand the empirical field of investigation to the public sector and civil servants at large, the sociologist Paul du Gay (2008) draws our attention to contemporary political and managerial demands for increased enthusiastic responsiveness in public offices. He claims that public servants are met with two emotional injunctions: one, which requires bureaucrats to be passionately responsive to the needs of their clients, and one, which requires bureaucrats to be enthusiastic advocates of particular policies. The demands for responsiveness and enthusiasm when applied to

professional work of civil servants include the aim to 'inculcate in bureaucratic conduct a sense of compassion or close identification with others' feelings' (2008: 336).

However, the idea of bringing compassion – or even better bringing *back* compassion into clinical procedures as something which has been lost in 'an era of scientific-bureaucratic medicine' (Harrison, 2002) and medical expert dominance, I will argue, is not only manifest in current reform attempts, but can also be found in medical practitioner literature, namely medical humanities literature or narrative based medical literature. This literature, which is often written by doctors, has turned to the humanities to understand the 'uncertainty and turmoil in medical practice and organization' (Brody, 1998: xv). The English historian of ideas, Monica Greco, describes the discipline as work 'who has turned to humanities with the aim of enabling doctors to address better the complexity of each medical situation, particularly with regard to its more intangible, personal and communicative aspects' (Greco, 2008: 30).

According to one author, Dr. Audrey Shafer, who is also the editor of the medical humanities journal, medical practice should not only be more effective but also strive 'to perfect and pop out a humanistic, compassionate, complete doctor' (Shafer, 2009: 4). Among the most significant features of the literature is the request for what the general practitioner Rita Charon (2001) terms 'empathic engagement' and 'authentic engagement' (2001: 1897-1898) in medical relationships. She argues that these forms of engagement can be ways to 'improve the effectiveness of the professionals' in current times when medical procedures speed up as a result of rearrangements of the health care system. She writes that through authentic engagement and true responsiveness with patients, doctors can achieve effective, engaged therapeutic relationships and be 'moved to act on the patients' behalf' (2001: 1897). In relation to processes of clinical treatment and care, Charon describes 'empathic engagement' and 'authentic engagement' as specific competences, which the doctor is required to exhibit in patient-doctor interactions. She furthermore writes that when

requiring such competences 'it may be that the physician's most potent therapeutic tool is the self, which is attuned to the patient through engagement, on the side of the patient through compassion, and available through reflection' (2001: 1899). Doctors are thus required to turn to individual patient cases with 'passionate', 'personal' and 'engaged' commitment.

In this article, I want to pursue the argument that public health care reforms and the request for medical professionals to exhibit 'compassionate' care, and forms of 'empathic' engagement raised in medical humanities literature, are hardly distinguishable: both inset emotional injunctions to medical professionals, which include a call for doctors to become more emotional available to patients and to exhibit a particular emotional behaviour in medical relationships, namely that of a 'sincere' or more 'natural' behaviour. On one hand, we see a movement towards rationalisation of medical services, tied to tropes of 'effectivity', 'quality', 'standardisation' and 'audit'. On the other hand, we have a movement towards a heightened degree of emotional sensitivity required on the part of individuals, encouraging a seemingly emancipation of emotions in medical relationships.

I will argue that the current framing of the role of emotions in medical work, and an apparent relaxation of emotional codes of conduct in this work, is manifest concurrently with formalised attempts to manage the emotions of patients in recordable, measurable and standardised ways. The latter tendency hence represents increasing constraints towards processes of emotional control in medical work and new modes of clinical governance rather than a loosening of emotional behaviour and manners.

The paper explores this assertion through an analysis of recent health care reforms. These reforms are understood as framing the role of emotions in medical work. In particular, I focus on the changing emotional demands the treatment concept which I term 'accelerated medicine' impose on doctors and on how doctors are expected to exhibit a variety of emotional skills in patient-doctor interactions, as for ex-

ample the exhibition. The inquiry draws on empirical data material from research on the introduction of accelerated cancer pathways in a cancer clinic at a major university hospital in Denmark. The article is structured as follows: First, political and managerial documents issued by, among other agencies, The National Board of Health, Denmark (NBH), describing the organisation of the accelerated cancer pathways and the institutional framing of the role of emotions in these pathways, are reviewed. Second, attention is paid to the possibilities for doctors of managing patients' feelings (and own feelings) in the rearranged pathways.

The approach to the management of emotions in work and organisational settings

To approach emotional injunctions to doctors in accelerated cancer pathways, I will use an Eliasian perspective on emotions, however without applying his greater framework of figuration sociology to the inquiry. His perspective is worth exploring in order to draw attention to how professional codes of conduct and exhibition of emotions in medical relationships are interwoven with the introduction of new modes of clinical governance in relation to cancer illnesses and their treatment. This section first explains some core elements of 'The Civilizing Process' theory (Elias, 2000) to understand socio-historical developments of formalisation and informalisation of emotions in processes of civilisation. Next, the section departs from the Elias follower, Wouters, to explain how we today witness an increasing demand for 'smoother' manners in health care organisations while these manners increasingly are formalised in measurable, standardised ways.

A core theme in Elias' work is how historical changes in power structures are reflected in changes in the 'psychological make-up of people' (2000: 369). Elias links the notion of *civilité* or *civilization* to changes in human codes of conduct and demonstrates how emotions are controlled in rationalised (i.e., civilised) societies. By doing this, he develops Weber's (1978) historical view on rationalisation, bureauc-

racy, and the impact of rational practices on the organisation of the public sphere and interpersonal relationships. Elias describes the social advantages of 'those able to moderate their affects' (2000: 370) and he demonstrates how specific forms of 'affect control' and 'management of the emotions' are products of dual developments of socially instilled agencies and self-restraint. These modes of regulation are different from earlier versions of external control such as physical force or acts of extreme violence. In his view, the increasing functional differentiation of society has since the Renaissance caused an expanding mutual dependence of its members, which has led to a social necessity for controlling one's feelings in public. One must interact and 'be hospitable to a potentially vast array of others' (Newton, 2001: 488), which also includes that one must interact with a 'smooth' face within employment and professional settings.

Also Weber stressed what he saw as a dual process of rationalisation and emotional self-restraint in the emergence of modern societies. Indeed, one of the strengths of bureaucracy was that its employees would follow standardised rules and procedures 'without regard for persons' and conceal personal (i.e. individual) feelings, such as love and hatred, in dealing with human affairs (Weber, 1978: 600). In relation to emotions, and in conflict with many readings of Weber, the emergence of the bureaucratic organisation was not equal to a development of emotional deficit. Weber acknowledged that emotions are an essential part of professional office work, but he argued that emotions must be controlled in matters appropriate to an ethic handling of office cases. However, Elias goes even further than Weber in emphasising the importance of affect control as a constitutive element of modern societies. New patterns of emotion management are connected to the development of peoples' increasing capacities of foresight, calculation and control. This also includes the capacity of individuals to observe and regulate themselves from the vantage point of others and to understand that expression of emotions has social consequences. Elias demonstrates for instance how the experience of repugnance and embarrassment en-

courages individuals to increasingly repress some passions in public encounters (e.g., anger, rage) while it opens up the possibility of the expression of other emotions (e.g., sympathy, happiness). The key objective here is that in professional work consisting of rational, bureaucratic actions, emotions are not as such neglected. Instead, emotions are increasingly being mouldered as part of the development of rational human agency. This happens in a complex interplay where people balance their emotions vis-a-vis others. Likewise, as also Goffman (1959) has described, people need to attune their conduct to that of others and their 'web of actions must be organised more and more strictly and accurately [...] in an increasingly differentiated, more even and more stable manner' (Elias, 2000: 367). Emotions, passionate impulses and desires undergo a civilising process with the result that they become increasingly formalised. The behaviour associated with civilité is shown to be directed towards the regulation of the conduct of individuals through detailed behaviour rules. Elias explains through lengthy empirical inquiry into etiquette books of the upper middle class in Europe from the late Middle Ages to the Renaissance, how highly refined rules of etiquette, such as how to eat meat and use knife and fork at a table, how to undress in the bedroom and how to blow one's nose and spit in front of others represent a transformation of human manners. He argues that through the civilising process these formal standards of socially sanctioned emotional behaviour are gradually replaced by more informal – and often more implicit - codes of conduct as an 'increasing social restraint towards self-restraint' (2000: 365).

In addition to Elias' theory of civilising processes and the 'commingling of patterns of conduct', Wouters presents the argument that self restraint today has become both more skilled and more subtle. He refers to this trend as '*informalization*' (Wouters 1986; 1999; 2009). The term refers to 'the relaxation of the social codes [...] in combination with increasing social demands on self-control; it implies a change in the patterns of social control and self-control and also a higher level of reflexivity on the part of individual people' (1999: 416-417). The expression resembles

the one used of Elias when he speaks about processes of 'constrains to be unconstraint' (Elias, 2000: 365-379). It indicates that processes of 'self-constraints have not only become more flexible, at the same time they have also become more strict' in recent developments of emotions control (Wouters, 1986: 1). According to Wouters, this trend gained momentum in the 1960's and 1970's, but is anyhow still one of the main characteristics of the civilising process. If we draw upon the Woodstock Festival as an example, we might get the impression that the festival and its surroundings' emancipation rhetorics include a radical emancipation of emotions, since the refinement of self-restraint implies less rule-based standards of emotional behaviour and more 'naturalness' in expressions. But Wouters argues that informalisation implies a 'different pattern of self-restraint [which] demanded not only greater sensitivity to varieties and greater flexibility in social conduct, but also a higher level of self control' (1986: 1). The apparent relaxation of emotional control and the performance of seemingly natural or relaxed 'decontrolled' actions go hand in hand with a refinement of emotion management: such is the argument.

Where do we see empirical evidence of informalisation in present-day work settings? According to Van Iterson et al. (2001) fieldtrips to public offices, such as tax bureaus, social welfare offices and hospitals, may give one the impression of a 'lowered acceptance of power and status differences, abating ceremony in meetings' and 'increasingly relaxed interaction' between people in the observed places (2001: 507). The authors also draw our attention to 'the growing tolerance for informal clothing, the use of first names and colloquial speech, confessing private feelings and expressing emotions [...] and in general the blurring of the boundaries between "work life" and "private life" as examples of aspects of informalisation processes (ibid). In addition, Van Vree (2011) points to transformations of the ways in which people behave in professional meetings to provide evidence of the informalisation trend. Generally, dominant meetings manners – regulated by rigid rules and customs to keep polite manners and to control meeting participants when social tensions increased -

have become smoother, easier and more flexible. However, whereas meetings have lost some of the formal characteristics, attending meetings now places greater demands on one's own initiative and feeling of responsibility. Informal meetings codes constitute just another set of pressure on how to behave, which might be experienced as even more restraining than previous formal rules. Participants in meetings must still act appropriately in front of others, but it is becoming more complicated what kinds of rules are guiding their behaviour. People thus need to find an agreeable balance between formal and informal behaviour; between formally defined standards of professional behaviour and informally unstated concerns of 'truly engaged' behaviour.

To sum up, processes of informalisation do not represent a loosening of emotional display or more relaxed emotion codes, but rather their refinements through seemingly more 'sincere' or 'natural' practices. Health care reforms which demand more 'empathic' engagement or more 'authenticity' of doctors' conduct, call for research attention, because they might tell us something important about the organisation of refined professional behaviour in health care organisations.

Method

The article is based on reflective analyses of material from a study concerning the management of emotions in accelerated medicine, which was completed in Danish health care services between June 2009 and January 2011. The selected material in this article provides evidence relating to some of the consequences of recent rearrangements of cancer care in Denmark, especially in regard to the emotional demands that these reforms impose on doctors working within cancer services. The Danish study supplements previous studies from especially UK National Health Services (NHS).

The analysis has the following structure. First, I review documents issued by The National Board of Health, Denmark, The Danish Government and clinical

articles published in practitioner journals, using a relatively simple 'informed content analysis' (Prior, 2003: 21). The documents contain detailed trajectory descriptions and I focus on their definitions of technical, social and psychological components of medical work in the trajectories. The sources of data are approached to analyse 'the active contribution of texts to organizational processes' (Cooren, 2004). The document material is selected in relation to its status as public reform material, which is exemplified in policy documents such as The National Cancer Plan I-III and The Health Packet 2009. As reform attempts, these documents have brought about new ways of organising health care processes as essentially a matter of changing existing routines to provide efficient and effective cancer diagnostic and care, or so the explanations go. In the material, particular categories are highlighted, which are categories of 'empathic' engagement and 'compassionate' care, such as, 'empathy', 'authenticity', 'engagement', 'responsiveness', and so forth. The second stage of the analysis builds upon interview and observational data from a cancer clinic to discuss the practical circumstances of managing the feelings of others in the programs of accelerated medicine.

Accelerated cancer pathways and the conceptualisation of the 'stand up patient'

In Denmark the organisation of cancer illnesses and their treatment have recently been rearranged to improve the provision of health care services to patients with cancer. The National Board of Health, Denmark, has introduced two national 'Cancer Plans' (NBH, 2000; 2005a) and an outline for Cancer Plan III has been approved at The Danish Governments annual budget proceedings (The Danish Ministry of Finance, 2010). A fundamental element of the Cancer Plans is the implementation of the so-called cancer treatment packages (in Danish: 'Kræftpakker') to optimise therapeutic procedures and minimise delays at hospitals and improve continuation and coordination across the primary and secondary health care sector. The treatment package concept bears close resemblance to the widespread concept of 'accelerated medi-

cine¹⁰: both emerge from an ambition concerning the establishment of economic, effective and efficient patient trajectories in the health delivery system (Roelsgaard Obling, 2010). I intend to use the term 'accelerated cancer pathways' or 'trajectory programs' throughout the article to minimise confusion to the reader.

Accelerated cancer pathways have been developed as an attempt to contain costs in an era of rising health care expenditures, limited financial resources and a political attention towards matters of public concern, such as empowering patient choice ('patient-centredness' rhetorics) and reducing waiting times for diagnostic and treatment. The trajectory program also seeks to meet clinical objectives by reducing the mortality rate of Danish cancer patients. The immediate effects of the implementation of the trajectory programs have been noticeable in regard to more rapid diagnostic and treatment procedures (NBH, 2010a), reduction of length of hospital stay¹¹, standardisation of clinical procedures across hospital units, quality improvement initiatives concerning for example standardisation of after surgery recovery procedures, and new administrative processes of documentation and performance measurement (NBH, 2010b).

A pervasive concern in the trajectory programs is what I term the *stand up patient* concern. Opposite to a traditional concern, where the patient must restore her bodily malfunction through bed rest and a non-defined length of hospital stay, the 'stand up patient' concern promotes the notion that patients become more ill if they stay in their (hospital-) bed for too long. Thus the point is here that the length of their hospital stay must be reduced. From this also follows that patients may get cured more efficiently through accelerated diagnostic and treatment processes, where wait-

¹⁰ I intend to use the terms 'accelerated clinical pathways', 'optimised clinical pathways' or 'treatment packages' more or less synonymous through the text. All terms indicate an occupation with continuation and standardisation in a given line of treatment and therapy, and focus on economics, quality and performance measurement. The term 'accelerated medicine' is used as an assemble term to capture core themes of accelerated cancer pathways, such as 'packaged diagnostic and treatment'; 'optimised care'; 'accelerated trajectory'; 'fast track surgery' and 'integrated care'.

¹¹ The general length of hospital stay for women who has been surgical treated for breast cancer has recently been reduced from 3,6 days to 1,2 days (Gärtner, 2010). This also includes women who had been operated for a tumour in the breast and where the whole breast has been removed.

ing times – including recovery time - are reduced while they are hospitalised. The patients' efforts to regain strength are transferred to locations outside the hospital – for example to private homes or rehabilitations centres. In a document issued by the Danish Government concerning quality reforms in the public sector it is stated that the result of the accelerated programs is 'that patients faster can get back to a normal life' (The Secretary of Ministers, 2007: 1). The health benefits to the patients are invoked as the key driving agent of the reform, but means of resource allocation also lie in the centre of these reforms. Advocates for the 'stand up patient' concern focus on accelerated trajectories as a future way to organise illness trajectories that will include 'optimised quality and economics with a reduction of hospital beds as a derivative consequence' (Kehlet and Hoejgaard, 2004: 4707).

Emotional injunctions to doctors in accelerated cancer pathways

An essential part of the accelerated cancer pathways is the introduction of new modes of clinical governance, i.e., new forms of controlling professionals' activities in clinical work. The activities in the accelerated cancer pathways are organised by written 'trajectory descriptions' which in detail determine how doctors ought to diagnose, treat and inform patients in the pathways. As originator of the trajectory descriptions, The National Board of Health, Denmark, describes in a policy document the aim of introducing the accelerated cancer pathways in the public health sector as that of providing 'optimal' services along a variety of parameters. It says in the document:

'The aim of "cancer packages" is to provide optimal diagnostic and treatment to patients in regard to shorten the pathways and hence improve prognosis, quality of life and decrease the anxiety caused by waiting time, where the cause of delay is unknown' (NBH, 2008a: Foreword)

The term 'packages' defines 'patient pathways, where every step is organised as time and content well-defined events, which follow a pre-booked trajectory plan' (NBH,

2005a). Furthermore, NBH emphasises that it is of crucial importance to 'diagnose and treat most [cancer] patients in *very fast* trajectories' (NBH, 2008b: 1, my emphasis). The descriptions show how hospitals must adopt ways to streamline and accelerate patient flows through what resemble methods of lean management and business process redesign. In other words, it is performance and management practices similar to those used to reduce financial costs, improve quality and address problematic bottlenecks in the Japanese car industry. The contention is that tight process control and closely integrated operations are more productive (Lowe et al., 1997).

The trajectory descriptions, which determine clinical work procedures as well as social and psychological aspects of these procedures in the accelerated cancer pathways, are mainly based upon evidence based, national and nation-wide clinical guidelines and recommendations¹². They consist of detailed procedure descriptions of the pre-hospital care phase of the pathways (for example pre-diagnostic examinations in primary care); the diagnostic phase; the treatment (surgical) phase; and the adjuvant phase. But descriptions of more general elements in relation to rehabilitation, palliation, provision of nursing care and delivery of patient information (NBH, 2005a; NBH 2008a) are also included in the documents.

A closer look at one of the descriptions, namely a description of the organisation of breast cancer illness and its treatment sheds light on the treatment phase of the program (NBH, 2009a: 15):

 0 Day. Managing patient referral; booking of patient-doctor consultations and examinations (3 days)

10

¹² Clinical guidelines are frequently used by doctors in their daily medical work. The guidelines typically guide medical decision making processes in the treatment of patients. Typically in cancer programs, the nation-wide clinical guidelines are developed by a professional society which is in charge of keeping the guidelines up to date and who are responsible for communicating the content of these guidelines to doctors across the private and public health care sector working within the field. See for example the work of The Danish Breast Cancer Group (http://www.dbcg.dk/). National clinical guidelines are developed by The National Board of Health, Denmark. The clinical guidelines support the various actions in the patient pathways and are thus a fundamental part of the development of the cancer 'trajectory descriptions'. See for example the NBH document on 'Connection between clinical guidelines and 'trajectory descriptions' (NBH, 2009b).

- 3 Day. Preliminary examination (including clinical examination); patient information; patient interview (nurse, anaesthetics, physiotherapist); eventual supplementing picture diagnostic tests (2 days)
- 6. Day. Patient time to use for reflexion (3 days)
- 6. Day. Valuation of co-morbidity
- 7. Day. Surgery (hospitalisation and (optimal) discharge of patient)
- 8. Day. Eventual discharge of patient

This is an example of a standard pathway, running from the moment a patient has been examined for breast cancer, has received a cancer diagnosis, undergoes surgery and gets discharged from the hospital. Altogether this leaves seven perhaps eight days for the doctors in a breast cancer clinic to deliver a malign diagnosis to the patients and her relatives; to prepare the patient emotionally and practically for future bodily interventions; to remove the cancer tumour, stitch up the patient and discharge her from the clinic. The doctors do not make this work alone. Besides nurses and secretaries, new supportive staff functions have been introduced to facilitate the 'flow time' of the pathways. The introduction of a 'personal' contact person is a good example of this. The role of this contact person is to establish human continuation in the pathways and 'to secure that patients do not experience a feeling of lostness in the system' (NBH, 2005a: 21), as the acceleration of procedures necessarily entails that the patients meet many different professional faces, each responsible for smaller bits of the pathways (see also NBH, 2008b). In addition to this support function, a pathway coordinator or case manager position has been introduced at hospitals to facilitate the speed-ups of the pathways (for example through pre-booking of diagnostic examinations) and to secure that involved professionals across medical specialties and clinical departments co-operate around the patients' treatment processes. The two examples of supportive staff functions are not distinguishable from the core function

itself but are attempts to ensure a much needed coordination of divisions of labour and clinical personnel in the rearranged pathways.

What emotional injunctions to medical professionals do the trajectory descriptions contain? This is a proper question since one can assume, when departing from an Eliasian perspective on emotions, that structural rearrangements of cancer treatment also imply an introduction of explicit emotional demands to doctors in order to establish standardised patterns of professional behaviour in the pathways. The National Board of Health (2008a) states that 'patients ought to experience a transparent trajectory where every future step is well-defined and where possible questions can always be answered' (2008b: 5). In order to prepare the patients for the prescheduled sequences of the accelerated trajectory and to 'reduce the insecurity of patients' (2008b: 2) a doctor's job includes taking active part in emotion management activities. This involves reducing, by means of strategic work activities, the feelings of anxiety experienced by many patients facing serious illness, such as cancer. It also involves preparing the patients for their mutilating surgery in which they are probably going to have a breast or parts of a breast removed; as well as preparing them for the subsequent chemotherapy and for the following years in which they must attend control visits at the hospital to ensure that their cancer illnesses are not progressing. For the concept of the 'stand up patient' to work in practice, patients must be regulated in detail through the pre-scheduled treatment procedures – emotional, behavioural, practical - so that the accelerated cancer pathway can be fulfilled on time and so that the clinical interventions, for example fast removal of the breast and early recovery, can succeed without disturbing interruptions – for example caused by an anxious, angry or too depressed patient.

To prepare the patients for accelerated procedures of diagnostic and treatment and to reduce anxiety, the descriptions inset demands of emotional behaviour that doctors must exhibit in the pathways. More specifically, the precondition for a successful cancer pathway is that doctors 'exhibit sincerity, obligingness, trust and

comprehension' in their encounter with patients (NBH, 2005b: 2). In continuation of the doctors' plight to give off certain emotions, they must also 'provide room for patients' spontaneous expressions of emotions' (ibid). The doctor is thus responsible for promoting a 'particular ethos or emotional atmosphere' (Goffman, 1967: 35) in doctor-patient interactions. The policy document Cancer Plan II, explains how this ethos is connected to the development of concrete competences or skills of the doctors (NBH, 2005a). It describes how public hospitals concurrent with the establishment of the accelerated pathways are told to 'develop and promote a culture where health professionals have competences and skills to identify with patients' values, emotions, responses and mentality' (2005a: 2). An appendix to the Cancer Plan II document explains how these competences and skills ought to be trained under a rather broad umbrella, termed 'training of good and efficient communication skills'. Efficient communication is understood as 'empathic, dialogue-based, respectful, and individual' (NBH, 2005a: 54) communication between patients and doctors, and 'good' communication is regarded as a solution to social and psychological difficulties that might occur in the accelerated encounters. Efficient communication techniques are here regarded as even more necessary than in traditional illness pathways, mainly because there is less time for the health professionals in the accelerated pathways to give room for patients' emotions and to prepare these individuals to undertake treatment in the pathways, or so the argument goes in the appendix to the Cancer Plan II. Also the many specialised work activities, which are spread out on many different hands in a very accelerated time span raise a legitimate concern in relation to informing patients about what is going to happen next; who is going to perform the next bodily intervention and who are responsible for what during the planned treatment process, and thus a concern about the attainment of the patients' approval of those interventions. However, the trajectory descriptions anticipate interpersonal difficulties in the pathways as merely technical difficulties. The compressed organisation of cancer diagnostic and treatment is presented as fostering patient needs for more information and better communication. Are these two needs met, no 'inappropriate' emotions will occur that would slow down the flow of actions.

Individual professionals and 'one shot available' for managing patients' emotions

The acceleration of medical work in the pathways has opened up a spectrum of new ways of coordinating and adjusting the doctors' clinical activities. The very conditions for managing patients' feelings in the pathways are restricted, mainly as an effect of the pathways' focus on manufactory speed and its character of rigid business process-design. The compressed ways of organising medical work in the pathways, as emphasised in the trajectory descriptions, also provide the frame for managing emotions in the patient-doctor interactions. An intern consultant who has introduced a cancer clinic's doctors to empathic communication techniques at the hospital, where the research for this paper was conducted, explains what sort of difficulties the rearrangements of the pathways might imply in relation to doctors' possibilities of managing the emotions of patients and relatives. She says:

'The rearrangements of the clinical pathways, which imply that patients undergo breast cancer surgery in the morning and are getting discharged a couple of hours after the intervention, might interfere with the way the professionals relate to their patients. In practice, the doctors may have only one shot available to make sure that the patients are emotional stable for the coming medical interventions. In other words, that the patient has understood the delivered messages and that she is prepared for what is coming up next'

This 'one shot available' of doctors to regulate the feelings of patients in a way that they are prepared for a compressed, future line of diagnostic and treatment procedures, and the way in which this 'shot' is handled is hence a crucial concern. As previous mentioned, it does not take many emotional disturbances – for example an aggressive patient who is furious at the system and therefore wants to make legal complaints – for the accelerated pathways to be deprived of their speed. An editorial in Journal of Danish Medical Association (Gaub, 2010) draws attention to what this 'one shot available' might involve in relation to emotional demands to doctors. The editorial sums up current experiences with the introduction of accelerated cancer pathway in the following way:

'The speed-up diagnostic process of the pathways raises great demands to the continuity between patient-doctor and patient-nurse, to communication and to the provision of empathy' (2010: 273)

It is evident that the speed concern not only affects the organisation of technical aspects of medical work, but also constrains 'softer' relational procedures in patient-doctor interactions, such as for example the doctors' provision of 'empathy' or 'responsiveness' in those interactions. A practitioner article on accelerated cancer pathways in the same journal states the importance of professionals' capability to show a mixture of 'personal' commitment, 'responsiveness', and 'empathy' in interactions, in order for the patient to be 'recognised as an individual person by the staff and not just a number among others down the line' (Junge et al., 2010: 274). A consequence of this attention to 'soft' dimensions of medical work is that 'this issues substantial demands to physicians' and nurses' capacity to encode patients' individual reactions over a shorter time' (2010: 277). The article further remarks that the introduction of accelerated cancer pathways contains three emotional injunctions to the staff involved: namely that of 'personal commitment', 'responsiveness' and 'empathy'.

If we move the empirical attention for a moment from public health care reform documents to evidence collected in a local cancer clinic, it becomes evident

how discourses of 'empathic engagement' and 'compassionate care' are articulated in the clinical environment. As also mentioned in the introduction of the article, the emotional injunctions to doctors are reframed by the practitioners themselves in their everyday work. This is evident at staff meetings and morning conferences, where continuous discussions of the quality improvement of the pathways are held. These discussions often take form of a discussion of experienced communication problems in patient encounters or the doctors share stories about particular troublesome characters (patients, relatives), which they needed to engage themselves in, in extraordinary or more involving ways than normal. In an interview conducted for this research, a doctor (David) gives this explanation of the current focus on provisions of 'empathic' engagement in the cancer clinic. He says:

'Before our patient pathways were redesigned we actually thought that we provided empathic responses [...], but then we discussed the issue during a staff meeting and we came to the conclusion, that we actually bore more resemblance to a hotel facility than to a therapeutic facility. Previously, we did not provide empathic care or empathic communication to our patients while they were hospitalised and while they underwent treatment in our clinic. After this insight and the following rearrangements of the cancer pathways, we added empathic communication as a crucial element to our activities in the pathways'.

'Empathic' engagement is not a free-of-charge kind of engagement that the individual doctor can freely choose to provide at will. Instead, it becomes 'a crucial element of the pathways', which must be performed alongside the line of more technical work tasks (e.g. to take a blood pressure or to remove a breast). Furthermore, the exhibition of 'empathic' engagement is followed by the same recording and measurement demands as every other clinical activity in the clinic (e.g. taking a biopsy sample; exam-

ining the lymph nodes under the armpit; prescribing medicine for nausea and vomiting): the provision of 'empathic' engagement must be recorded in the patient record system. David explains this request for recording practice:

'In a modern hospital, where everything is measured and controlled, one must always record one's conduct. This includes a registration of the exhibition of empathy. If you are not willing to do this, the clinic's spending account may get reduced by the hospital management, because then we are not able to document that we spend our time in the consultancy rooms providing sincere care. If your job is to manage an emotional problem of a patient and if this problem is not measurable, then you get into deep troubles in cases where you need to explain your rationale for doing things to others, for example in auditing. We thus need to write down what we do. By doing this manoeuvre we also make sure that everybody follows the same standard procedures'

A possible reading of this statement is that clinics or hospital units that record their public exhibition of 'emphatic' engagement most effectively in the record system are likely to keep their granted budget, since they can prove that the consultancy time spend in the clinic is spend managing 'an emotional problem of a patient' and they can thus prove that they meet diverse interests, such as political and managerial pledges and clinical objectives. It also embraces lay expectations towards 'being met by the doctor as a whole person instead of a case among many others'. On the doctor's part there was not any irony in the statement with regard to this rather contradictory dimension of the provision of empathy. The 'soft' dimensions of his work must be surrounding by the same quality standards and biomedical rationality as the 'harder' aspects of his work, or that is the impression he gives.

Various forms of emotional conduct not only has to be recorded in patient records, as written confessions of engagement to secure the annual budget, but must furthermore be recorded as part of the concerns of a national quality plan, entitled *Quality First* (Region Hovedstaden, 2010). This quality plan outlines ten strategically objectives for 'bringing the patient in focus'. One of these objectives is that patients ought to feel safe, respected and understood when discharged from the hospital. To facilitate that this objective is followed at local department levels, an annual performance measurement of how individual professionals manage to make patients feel confident before hospital discharge has recently been implemented (I draw here on an internal auditing document from a university hospital in Denmark, 2010). The annual assessment of the management of patients' feelings is introduced together with initiatives to educate doctors in knowing the emotions of others through medical training and communication courses. Through this training, the doctors are expected to be able to manage the emotions of patients both more effectively and more efficiently.

Discussion

As has been shown, the occupation with various forms of engagement in accelerated medical relationships is saturated with demands of efficiency, quality and new techniques to improve the outcome of the pathways. Herein lays an apparent peculiarity: the more the importance of 'compassionate' care is stressed in accelerated medical work and the more emotional display of the doctor is focused upon, the more emotions become subject to measurement, standardisation and control. These manoeuvres are easily understood on a paper level. Putting them into practice proves a lot more difficult, because how does one actually measure the delivery of for example 'sincere, 'empathic' engagement? In the beginning of the article, I quoted the two English scholars, David A. Buchanan and Louise Fitzgerald (2011), for rhetorically asking how the delivery of compassion in patient-doctor interactions is to be measured and by whom. The previous quote from a doctor suggested that when there is no hard

data available, the obvious response is to generate some, and thus to measure empathic responses and emotional behaviour alongside other clinical interventions.

There is nothing radically new in suggesting that for patient-doctor interactions or professional-client interactions to succeed it requires emotion management on the part of professional workers to regulate the feelings of patients or clients in public meetings. For instance, it takes elaborated effort to reduce feelings of alienation when patients attend the medical system (James, 1993; Lupton, 2003b; Bolton, 2005). Management of emotions typically operates through what Elias terms 'emotion codes' or 'codes of emotional display'. These codes are socio-historical constructed and are hence up for steady historical and cultural unravelling. In the context of this paper, doctors apply emotion codes of medical rationality to handle patients' feelings when a patient attend the clinic for advice about, and treatment of, complaints. These codes are typically internalised norms and values regarding the experience and expression of emotions, which are learned through vocational training, supervision and hours of clinical practice experience (Smiths and Kleinman, 1989). As such the emotion codes function as standards of socially sanctioned behaviour regulating the experience and expression of patients' emotions (not to forget the regulation of doctors' own feelings) in the different diagnostic and treatment phases of the accelerated cancer pathway. While there is nothing groundbreaking in suggesting that the managements of feelings is a necessary part of doctors' work, it is the very presentation of emotions in the trajectory descriptions of accelerated medical work which represents a significant trend. The descriptions contain a promotion of for example 'good' communication and 'responsiveness' as effective, personal competences of the doctor, and moreover as competences which are currently lacking in medical relationships. The trajectory descriptions' occupation with reducing waiting times and with treating patients more rapidly leads to the concern that human components of the programs are missing. Hence, emotional competences of the doctors ought to be developed to bring back the 'human' components into practice. In fact, these competences are presented in the descriptions as crucial for the success of the pathways. The emotional competences promoted by the reviewed documents issued by the National Board of Health, Denmark, are competences in tune with contemporary values of 'empathic' engagement or 'authentic' engagement, which are promoted by the medical humanities literature. This kind of literature, as explained in the introduction, exactly requests doctors to engage authentically with their patients and to engage themselves 'entirely' (i.e. they ought to deploy skills of social intimacy) in the consultancies in order to provide efficient treatment and care (e.g. Charon, 2001; Shafter, 2009).

If we accept that the trajectory descriptions can be viewed as insetting demands of specific kinds to emotional behaviour in the accelerated pathways, then the implications of emotional injunctions to doctors' work are significant. This includes formal standards of interactional behaviour to use for example in precarious situations where the doctor ought to deliver a cancer diagnosis to a patient as well as more informal concerns of dialogue-based, responsive behaviour to utilise in meetings where the aim for example is to 'empower' the patient and her 'inner' strengths in the treatment process, or to calm her down if her world is falling apart, provoked by her illness. That is, I argue, an insertion of both formal and informal rules of emotional behaviour in relation to public meetings between professionals and patients. First, as formal rules the trajectory descriptions compel the doctors to act correctly according to a set of formally defined standards – and of course according to professional emotion codes. Second, as informal rules the descriptions inset demands to the doctors to behave accordingly to a set of unstated, flimsy, yet strongly expected ways of behaving, for example through 'empathic' engagement or 'compassionate' care. This commits doctors to find a delicate balance between formal and informal emotional behaviour while they respond to the needs of their patients.

Following this Eliasian perspective, it may be argued that the distinct emotional injunctions, which are promoted in the trajectory descriptions, may represent a further step in processes of civilising emotions and, what is more, the descriptions inset a quest for the strategic and deliberate elicitation of particular emotions, while it simultaneously frame the role and experience of other 'inappropriate' emotions. During this most recent phase of social development, the challenge of doctors is to regulate their conduct, not only by means of professional ideologies, clinical guidelines and rigid rules of standardisation, but also by means of conscious reflexion on deliberate expression of emotions. The further civilisation of the conducts of doctors may lead to increasingly more complicated and more extensive modes of self-presentation in medical relationships. A possible outcome of this could be a higher risk of job burnout or morale fatigueless within the profession of medicine.

Concluding thoughts

The aim of this paper was to show how recent health care reforms and the introduction of accelerated cancer pathways inset emotional injunctions to medical professionals. Through review of documents and other qualitative material it has been demonstrated that these reforms require that medical professionals exhibit particular emotions in patient-doctor interactions and that the management of patients' feelings is conducted in ways that can be recorded and be subject to quality improvement. The core of the papers' argument is the apparent contradiction between the emotional injunction to doctors to become emotionally available to patients and the fact that this availability (for the provision of empathy and authenticity) itself must be measurable and rationalised.

The request for emotions and emotional behaviour in medical work is not as simple as it might sound. On one hand, the trajectory descriptions for example ask us to pay attention to emotional engagement; it asks for 'authenticity' and 'sincerity'. The descriptions encourage doctors to display and experience 'sincere' emotions while providing various medical services to patients. And they furthermore ask the doctors to use 'themselves' in this process. This is in tune with the lay paraphrase

'patients want real, unique people not robots when they attend the health care services for help'. On the other hand, the descriptions anticipate the need for making this emotional appearance measurable and subject to quality improvement, and hence they frame emotions in patient-doctor interactions as something, which can be managed, evaluated and measured. Urges for formality and informality of emotional behaviour in medical work complicate what it is exactly that is expected of doctors in public meetings (e.g. Van Vree, 2011). The emotional injunctions inset by recent reforms of the public health care sector hence promote both a 'humanisation' ideal and a 'standardisation' ideal in the management of emotions: the reforms expect doctors to be compassionately engaged while they also expect them to display emotions in extremely regulated ways that are measurable and can be benchmarked across department units and health care organisations.

Present trends of 'compassionate care' may be seen as a trend where emotional restraints are cast off in favour of more self-expression and liberated emotional expressions of 'empathy', 'authenticity', 'engagement' and 'responsiveness'. Wouters (1989) however, argues with reference to Elias that this development can more accurately be seen as a tendency of 'controlled de-controlling of emotions' (1989: 106). The complexity of medical procedures in the accelerated cancer pathways and discourses of 'compassionate' care and more 'human' engagement, result in increasing emotional control rather than in the loosening of ways of emotional behaviour of medical professionals in the public health care sector. If one interprets the emotional injunctions issued by the National Board of Health and other public agencies from the vantage point of Elias, the discourses of emotionality and the realisation of a 'compassionate doctor' is just a further evolutionary step in the long-term structural development of societies and the changes in peoples' social character.

The emotional injunctions to medical professionals appear contradictory when we situate them in the clinical environment of the accelerated pathways in which they are brought to work. The health care reform documents encourage proc-

esses of standardisation while expecting room for personal and unique concerns; they emphasise intimacy while expecting maximum acceleration of services; they advocate for looser or more relaxed forms of emotional behaviour while strengthening regulation and control of others' and own emotions; they grant attention to the individual patient while expecting streamlined flows of patient cases, and finally they demand reflexively orchestrated ways of managing emotions while reducing the available time for this kind of work down to 'one shot available'.

Somehow counterintuitive, then, it seems that emotions and the management of emotions play a significant role in organisational structures that systematically both accelerate and reduce human interactions in public meetings, while intending to make these interactions more 'human'.

Chapter 6: Training of controlled empathy in accelerated cancer care

Abstract

This paper uses the dramaturgical lens of Goffman to explore the training of controlled empathy in accelerated cancer care. The paper focuses on a training workshop in 'empathic communication', during which doctors from a cancer clinic learn to recognise and control the emotional framing of doctor-patient interactions. Through a descriptive analysis, it addresses how communication techniques are rehearsed and it discusses the effects of this training. It is shown that the performance of communicating empathically relies on standardised scripts, which direct and cultivate the conduct of doctors. The paper concludes that contemporary reforming drives in public health care insert a renewed focus on humanistic values in medical interactions between doctors and patients, such as a focus on doctors' modes of engagement in these interactions. However, these values increasingly become the aim of techniques of micromanagement such as qualitative measurement and performance audit. In other words, attempts to improve 'soft' dimensions of medical services entail a further standardisation of these dimensions.

Keywords: Medical training, empathy, doctors, Goffman, cultivation of conduct.

Introduction

The paper explores the training of 'empathy' and 'empathic responses' in medical interaction. More specific, it focuses on how doctors' management of emotions can be trained as a response to a current movement towards rationalisation and tighter control of medical work. In addition, the paper explores how this training may also be a response to a current movement towards making medical services — including interactions between doctors and patients - more human.

My empirical inquiry is situated in the context of present-day healthcare in Denmark. Recent introductions of managed care processes for the production and efficiency of public healthcare services have resulted in a reorganisation of cancer illnesses and their treatment in to accelerated patient pathways, focusing on clinical work procedures, quality improvement and documentation standardisation. In the Danish health care system, cancer illnesses are organised in so-called treatment 'packages'. The purpose of the introduction of these packages is to reduce waiting times and organisational delays, speed up processes of diagnosing and treatment, and strengthen the coordination of patient treatment between hospital units and sectors. Attempts to organise the treatment of cancer illnesses into streamlined production units are not only present in Denmark but for instance also in the UK (e.g. Harrison and McDonald, 2008).

On the one hand, the accelerated cancer pathways are part of a clinical strategy to improve the overall health outcome by offering patients safer and more effective care. On the other hand, the pathways are a symptom of New Public Management and managerial means to rationalise public services and directing the conduct of public professionals (Power, 1997; Du Gay, 2000b; Miller and Rose, 2008). Generally, the intention is to make health care organisations more accountable, more customer-oriented and more efficient in their use of resources. These requests for performance improvement involve a movement towards rationalising medical work. As a new mode of governance, these reforming drives not only concern the so-called 'hard' dimensions of medical work (EBM, clinical procedures, economic incentives, safety and audit tools). As an object of interest of this paper, the reforming drives also include a pronounced focus on 'soft' dimensions of medical services. I call these dimensions soft, because they, among other things, comprise the ways in which health care workers today are expected to make themselves emotionally available to patients through a facilitation of 'partnership relationships' between doctor and patient (Bub,

2004) and through 'patient-centred' interactions (Mead and Bower, 2000, Mead, 2002).

The paper takes its point of departure in this configuration of today's medical services and demonstrates how the 'emotional availability' of doctors becomes object for attempts of governance, intended to structure and control the interaction between patients and doctors as well as to cultivate the conduct of doctors. The body of the research presents material collected from a training workshop in 'empathic communication'. Participants in the workshop were health care workers from a cancer clinic. The training involved that especially doctors participated in stage performances and were trained in performing 'empathic' responses in interactions with patients. The aim of the workshop was to make these relationships more efficient; to save time and improve the quality of care. The exact clinical 'problem field', which the training workshop was an answer to, is the difficult or serious communication act where details about diagnosis and clinical treatment plans are delivered in the medical consultation.

An intrinsic focus in the accelerated cancer pathways is a focus on the development of communication skills, including empathic response skills. The assumption is that better communication between cancer patient and doctor not only improves the over-all health outcome, but also facilitates the accelerated procedures of medical interventions in the pathways. According to the imperative of effective communication, which is put forth by for instance official descriptions of the cancer pathways (The National Board of Health, Denmark, 2008a; 2009a), a good doctor masters various communicative competences. As a competence, communication can be used by doctors to help coordinate actions and emotion conducts between themselves and patients, and within themselves. Central to these ideas about communication competence is how communication can be mobilised as means of cultivating the conduct of the doctor (May et al., 2006). By claiming that problems in interactions

between patients and doctors are merely technical in nature, the training in communicating empathically is identified as a possible resolution.

'Empathy' or 'empathic responses' is often evoked in thinking on the emotional relationship between doctors and patients. Larson and Yao (2005) define 'empathy 'as a powerful skill' (2005: 1100) and they propose that empathic responses can be learned through intensive communication training. The authors argue that doctors are more effective and enjoy more job satisfaction if they recognise that their work includes the management of emotions, and if they reflexively engage in the work of coordinating emotions between people and within themselves. The training of managing emotions includes both those of 'stage performance' and those of coping with the after effects of a stressful job. The authors argue that emotion management training is required in medical practice and they identify the need to teach some acting skills of this kind of labour. This argument is backed up by similar studies that are likewise occupied with the training dimension of 'empathy' (see for example Teherani et al, 2008; Neumann et al, 2009). These studies assert that when empathic skills are explicitly taught, doctors will experience less stress on the job and the possibility of job burnout will be reduced.

The paper makes two contributions to existing debates on 'patient-centredness' and the ethics of communication in doctor-patient interaction (Mead and Bower, 2000; Mead, 2002; May et al., 2006). First, it discusses how a training workshop in empathic communication is an example of how communication relies extensively on the use of a standardised language and the proper management of emotions in order to coordinate activities and the conduct of individuals in medical interactions. Second, the paper discusses how communication is 'a slippery sociological centaur' in the organisation of medical services (Illouz, 2007). It is slippery because it merges seemingly 'soft' therapeutic action with rationalised practices in order to make doctors emotionally available in accelerated medical relationships. In addition, I argue that doctors' behaviour become an objective for performance improvement in

health care organisations, as this behaviour can be intensively trained and become subject to comparison and susceptible to cost-benefit. This development in turn challenges the understanding of 'soft' dimensions of medical services, which previous studies within health care largely perceive as invisible and informal, and therefore inaccessible to management (Strauss et al., 1982, 1997; Star, 1991; Star and Strauss, 1999).

Theoretical framework

A dramaturgical lens developed by Goffman (1959) provides theoretical inspiration for my approach. This lens is employed to analyse the workshop as a specific kind of region where the doctors, as performers, engage in a drama before an audience and strategically present themselves in ways intended to foster favourable impressions in both competing performers (patients) and audience (doctors, nurses, consultants).

Many performers—for example, health care workers, scientists and lawyers—keep their preparation for public display hidden behind the scenes. This is also the case in this research, where communication skills are trained at a medical training facility to strengthen the everyday performances of doctors in a cancer clinic. Goffman (1959) uses the term *arts of impression management* to refer to the set of techniques and attributes required of an actor 'for the work of successfully staging a character' (1959: 203). These techniques also guide the display of feelings to which performers must adhere. Essentially, when performers are 'on stage', they must express emotions appropriate to the part they are playing. Training is required to manage the impression of others and to be recognised as 'skilled emotion workers' in public encounters (Bolton and Boyd, 2003). Goffman's idea of impression management and its effect upon organisational action patterns provides a means of understanding how emotional responses are actively managed by performers according to the rules of a particular situation.

Following Goffman, Hochschild refines his dramaturgy of work settings to understand how emotion in organisations is managed in a number of ways. Hochschild (1979) defines *emotion work* as 'the act of trying to change in degree or quality an emotion or feeling' (1979: 561) in oneself as well as in others. Emotion work involves individuals' management of feelings in order to conform to the demands of a particular situation. Hochschild uses the term emotional labour for emotion management that is performed for a wage. Emotional labour is associated with jobs that require workers to produce a positive or negative emotional state in others. According to Hochschild, we can think of emotions as a covert resource that can be organised, trained and regulated as something organisations need to pursue and fulfil their commercial aims. The purpose of emotional labour, she states, is to promote a good company image, which persuades customers to believe that they receive warm and intimate service, leaving an impression of high-quality customer service. The individual worker is thus getting paid to 'smile', to be 'caring' and to be 'responsive'. Hochschild's work has had major influence on emotion-oriented studies in health care, particular with reference to the nursing profession (James, 1989; Bolton, 2005; Mann, 2005; Theodosius, 2008). However, this literature rarely deals with how the management of emotions can be conceived as a technical problem, which can be refined and directed as part of the re-skilling of doctors. This paper exhibits the detailed training through which doctors are equipped with the capacity to act as a certain sort of person, namely that of an effective emotion manager.

As a supplement to Hochschild's work on the management of emotions, Strauss et al. (1982) offer the term *sentimental work* to coin 'soft' dimensions of medical services. These authors see sentimental work in terms of often more informal aspects of medical work, which normally take place as invisible work activities 'behind the scene' of clinical action, for instance, in the undocumented provision of psycho-social support to suffering patients. Sentimental work can be employed to ward off disturbing feelings in interactions, such as patient anxiety. This paper is inspired

by the notion of sentimental work. However, I challenge the characterisation of sentimental work as merely informalised, invisible work (e.g. Star, 1991; Star and Strauss, 1999; Strauss et al., 1997) and demonstrate how this kind of work indeed has become subject to management in contemporary health care organisations.

Material and methods

The paper uses qualitative material collected from a one-day training workshop with doctors, nurses and leading staff members from a cancer clinic. This includes participant observation, interview and document material. The material has been collected as part of a larger project of research into the management of emotions in accelerated medical relationships. The larger fieldwork was carried out between June 2009 and January 2011 in administrative and clinical units of a cancer clinic and nearby facilities at a major university hospital in Denmark.

I had access to the workshop as an ordinary participant. Even though I did not take actively part in the role-playing scenarios or any of the other exercises during the day, I was part of audience. The workshop lasted approximately seven hours. Altogether 18 staff members participated in the workshop: eight doctors (female and male senior consultants and junior residents), eight nurses (female), two consultants (a female psychologist and a medical teacher). Some of the doctors in the sample had been employed in the cancer clinic for many years, while others were relatively newcomers. The same was the case with the nurses involved. I have chosen to focus solely on the doctors' stage performances in this paper in order to follow the workshop's aim of re-skilling doctors through the use of communication skills.

I did not audiotape the activities of the workshop scenarios mainly because of explicit concerns expressed by some of the participants that the scenarios were too sensitive to be recorded. I therefore wrote fieldnotes during the various workshop sessions. The notes cover both the observed activities and some informal conversations that I had with participants during the day.

In addition to the participant observations, three qualitative interviews with one consultant and two department heads are included in the material. The main focus of the interviews was to gain a deeper insight into the establishment of the workshop and to understand how the informants, as initiators of the workshop, believed that the workshop was a necessary move in the development of the clinic's accelerated cancer pathways. The interviews had a semi-structured character in relation to how the interviewer primarily asked questions of 'how' and what' to understand the background history of the workshop (e.g. Fontana and Frey, 2002; Gilham, 2005). The three individual interviews each lasted approximately one hour. They were all conducted at the hospital, and the interviews were fully transcribed. The identities of the participants in the workshop as well as interviewee identities have been concealed in the paper.

A presentation of a training workshop

The writing that follows first presents the preliminaries of the workshop. Then, it presents the drama, its performers and its audience. The workshop activities are presented using Goffman's conception of region (1959: 109-115): the workshop is conceptualised as a 'backstage' region where doctors train and discuss the emotional responses (a training and discussion that is carefully concealed from the front region audience), which they are expected to deliberately display in the front region (the cancer clinic). Finally, it discusses the stage performances in a context of new modes of governing public health care services.

The situatedness of the workshop

Doctors and nurses employed in the cancer clinic were officially invited to participate in a workshop to train communication skills and competences related to the serious conversations that take place in the clinic, including the disclosure of clinical treatment possibilities in relation to malign diagnosis. The idea of the workshop has partly

emerged from annual appraisal interviews, where staff members requested further training that would better equip them to meet emotional needs of their patients. However, the workshop is also relevant in relation to some implemented structural changes in the clinic. Recently the clinic's patient trajectories had been rearranged into accelerated, streamlined production flows. Generally, this process of redesign has four purposes: to improve the health outcome, to reduce waiting times, to regulate and standardise work tasks, and to meet the objectives of performance measurement. Every medical activity in the pathways relies on detailed policy descriptions, which have been developed in the National Board of Health, Denmark (NBH, 2008a; 2009a). The cancer pathway descriptions detail 'what needs to be done', 'who should do it' and 'how [we should] register our conducted interventions'. These descriptions also cover communication at large, from information about clinical treatment possibilities to interpersonal communication between health professionals and patients. This also includes procedures of 'empathic' patient conversation(s) and delivery of 'proper' information throughout the pathways. Communication is defined as a mean to execute the accelerated activities of the pathways effectively while attending to 'patient-centred' concerns of trustworthy, responsive communication.

The workshop was thus situated as a double-headed arrangement. On the one hand, it strengthens rationalised dimensions of the structural rearrangements of the accelerated cancer pathways. It focused on the parts of the pathways, which involve communication and emotional transactions between doctor and patient. On the other hand, the workshop meet some concrete needs of workers in the cancer clinic, who have requested such training to improve their way of handling emotionally challenging situations in the pathways.

Invitation to join the workshop

Let me first describe the practical set-up of the workshop. During a planning phase, the head of nursing has met with two consultants from respectively a development department and a medical training department at the hospital. Their meetings have resulted in a detailed training description and they have collectively constructed two illness cases. These cases are typical of emotionally challenging situations in the clinic and they frame the content of the later role-plays.

Approximately one week before the workshop, the staff received a formal letter from their department heads informing them about time and location of the planned event. Workshop participation was mandatory (i.e., 'On October the 8th, you need to attend a training course at 8.30 AM'), and the staff was divided in half so that they could join workshop in two shifts without completely closing down the clinic's acute activities. A description of the training course was attached to the letter, which was signed by the consultants 'who look very much forward to finally meeting everybody'. The purpose of the workshop was explained to staff members in the letter. It says:

'Through the use of communication tools, participants will be able to structure and manage a difficult patient conversation, while being conscious about what they are bringing to the conversation – on both a personal and professional level – with respect to the patient'.

The construction of a respectful conversation between patient and health care worker has often been conceptualised through the notion of 'empathy' and the development of skills of empathy and sympathy (Mark, 2005). The emphasis on learning skills of empathy is further explained in the invitation:

'Achieving and controlling skills of empathic communication is not a question of personal style or of having the right personality traits [...] empathic communication skills result in effective consultation, satisfied patients and an increased overall patient comfort'.

The emotional components of communication involved in the service culture of health care organisation can be labelled 'emotional labour' (Hochschild, 1983). This includes the management of patients' feelings through for instance recognised and controlled communication skills. The workshop invitation situates a need for the establishment of 'patient satisfaction' in the consultation between doctor and patient, and the question becomes how the training in techniques of emotion management can be effectively achieved at the workshop.

The training sessions

On a particularly rainy October morning, half of the clinic's nurses and doctors gathered in a training facility. Department heads had joined a similar session the day before and were therefore not present on this particular day. In their role as hostesses of the event, the two consultants welcomed the participants and introduced them to the workshop activities. The workshop consisted of three separate sessions: an oral presentation on how to perform a serious/difficult conversation, a two-person teamwork (including discussions about and training of empathic communication skills in smaller groups) and role-playing scenarios with simulated patients. The first two sessions provided the education platform on which the subsequent role-playing scenarios were based.

One of the consultants explained at the beginning of her oral presentation that:

'It is not necessary to be a psychiatrist to use empathic communication. What is required is that you are self-aware of clues for empathy, as these clues arise during encounters. Typically, the clues will come from patients' emotional outbursts, and these will be the clue to your empathic response. Today, we shall focus on training these emotional responses'.

The professional conversation, the participants were told, is a 'patient-centred' conversation. It takes as its points of reference both the patient's knowledge and understanding of his or her illness, and the patient's particular emotions. The patient-centred conversation consists of three parts: content (e.g., information about test results); process (e.g., facilitation of conversation) and relations (e.g., emotional support through active listening). The performers were taught about the importance of facilitating the conversation process while recognising the needs of the patient. The performers were also introduced to elements of crisis psychology. Empathy, or as they were told, the ability to identify with another person's point of view and feelings, is an instrumental skill in health care worker-patient relationships through which the patient must be recognised and labelled as an emotional individual. Furthermore, empathy can be evoked to sustain the flow of medical situations.

Once in a while during the presentation, the participants were told to practise small conversations in pairs. One of these exercises was, for instance, a mirror exercise, in which the teams were told to communicate an essential experience from the clinic to one another and then to re-tell the story in detail, mirroring every sentence and summing up the main content of the story. One of the participants took her point of departure in a recent experience with a young mother who had a very bad prognosis due to her progressed cancer illness. The doctor explained how she tried to facilitate this woman and how this was difficult because the doctor experienced that she was getting too emotionally involved in the situation on a personal level. After each story, audience had to tell the performer that his or her particular experiences were both meaningful and understandable. Their task was to show cues of empathic communication and thereby appreciate the emotions, values and reasoning of another person. Listening to others, or in this case mirroring another's feelings and interpretations, was described by the consultants as crucial to the ability to prevent conflict during interaction. Moreover, active listening furthers the creation of cooperation between professionals and patients and facilitates intimate clinical procedures.

The next step in the program was to train emotional responses in order to evoke an impression of an engaged doctor in others. To focus on emotional responses, the oral presentation took the participants through the general rules of empathic communication, which they must follow in role-playing scenarios where serious messages (i.e., details on the diagnosis of cancer and treatment possibilities) are delivered to patients. Statements that facilitate empathy, the participants were told by the consultants, include queries (e.g., 'How does this information make you feel?' and 'Can you tell me what you feel now?'), clarifications (e.g., 'I want to make sure that we agree on this' and 'Do you understand why this next step is important?') and responses (e.g., 'I can understand what you feel' and 'It sounds like you are very upset now').

In pairs, the participants then discussed what tends to work and not work in interactions with patients in the clinic by drawing on their own experiences. Issues such as professionalism, emotional outbursts (e.g., anxiety or anger), recognition of patients' feelings and control of conversations were debated and these topics were subsequently discussed briefly in a plenary session before the role-playing scenarios. One participant summed up his experiences from the exercise, clarifying that:

'All of us must agree that the most important thing, no matter how much we talk about communication and attend to what the patients prefer, is that we rely on our professionalism. This means that we must act relatively certain and simultaneously try to be both intellectually and emotionally responsive, which also includes taking control with and guiding the content of patient encounters'.

Goffman (1956) writes how 'during interaction the individual is expected to posses certain attributes, capacities, and information which, taken together, fit together into a self that is at once coherently unified and appropriate for the occasion' (1956: 268).

As stated by the participant in the quote, there is no discernable difference between one's self presentation as intellectual (i.e., detached professional), who frames and control the interaction, and the presentation of an emotional self, who responds to the affective needs of a patient. However, what is important in this statement is the problematization of communication in relation to a classic conceptualisation of a doctor-centred model of medical consultation. The focus on communication may involve a refiguration of the consultation and the kind of work this particular situation involves on the part of doctors.

The next item on the agenda was a staging of two role-playing scenarios.

Staging the role-playing scenarios

While planning the workshop, the department heads had identified two typical cases of emotionally challenging situations for the staff: one case in which a patient reacts emotionally cold and distant and another case in which a relative to a patient reacts emotionally inappropriate and is too emotionally involved. The two cases represent situations in which an individual fails to respond and give off a predictable impression to a predefined, scripted situation. The patient and the relative are presented as either not emotional enough or too emotional to match a general presentation of a 'cancer patient' and a 'cancer relative'. The performers on the stage are: two professional actors who play the roles of respectively 'patient' and 'relative', male doctors who play the part of 'doctor' and female nurses who play the part of 'nurse'. The stage is designed so it looks like an ordinary outpatient consultancy room. Before the role-playing scenario started, the consultants instructed the performance team to act 'naturally' as if the stage performance was taking place in the clinic. The team was also told to keep in mind the 'props' they had been given during the morning session on the rules of empathic communication.

First stage performance: A patient who is emotionally cold and distant

The script of the role play is written down in handout notes, which the performers read through as if they were medical record sheets:

'The time is 14:05 PM and in cooperation with the nurse, you are going to talk with a patient. Your task is to inform the patient about surgery and plans for further treatment, including the duration of the hospitalisation stay, the hospitalisation procedure, adjuvant treatment (radiotherapy) and pain therapy'.

This first role-play was constructed to challenge the performers with an illness case, where the patient's behaviour is described as deviant in terms of her cold attitude. The note further explains that:

'The doctor has just been informed by the nurse that the patient has been complaining several times about the long waiting time and is now seriously upset. The patient is very annoyed with the system [the healthcare system]'.

The role-playing then starts. The patient is an elegant middle-age woman, dressed in black and carrying a laptop handbag. She is clearly annoyed with everything and complains about why everybody seems to have ignored her during the long waiting time. The two performers, i.e., doctor and patient, engage in a conversation:

Patient: I would like to know the exact date of surgery. I need to find somebody who can take care of my kids while I'm hospitalised.

Doctor: I will find you a date in our booking system. You will be asked to attend the clinic around 7 AM on the day of surgery [...] During the sur-

gery, we will take some tissue samples for pathology [...] Next, I will demonstrate how we plan to do the surgery.

Patient: I live together with my two kids. I'm divorced, and I have full custody. I need to have some answers – for example, how am I supposed to inform my work about my illness?

Doctor: We have found a malign lump in your breast, and we need to remove it. We will make a surgical excision of your lymph nodes in the arm pit. I now want to show you how we make a so-called sentinel node procedure [He draws a rough sketch of the lymph node system].

Patient: It is difficult to comprehend what all this actually entails [...] Should I be worried about my future?

Doctor: After the surgery, you will be offered radiotherapy. Radiotherapy is standard after-treatment to patients undergoing lumpectomy [...]. Now we should talk about the schedule of the day of your surgery.

During the performance the doctor manages the various impressions that he earlier learned during the group exercises to facilitate empathy. He nods; holds eye contact; makes small conversation breaks, and tries to use the phrases that the performers have practiced to provide 'active listening'.

However, most of all, he seems to have captured the objective of 'being in control of the conversation'. He neither asks the patient about her frustration against the 'system' and, perhaps more important, nor does he respond to the patient's small cues to change the script - for example, when the patient wants to talk about her future prognosis and family situation.

The doctor's display of empathy is hence boiled down to a registration of the emotions of the patient and he tries to neutralise or manage the emotions of the patient, which he finds disturbing to their relationship. The doctor directly faces the audience and tells it that he perceives the patient as 'being cold and non-responsive'.

In his opinion, she does not fit into the official 'frame' of the situation because her mode of appearance is deviant from for instance a presentation of a crying individual.

The consultant interrupts the stage performance after approximately 30 minutes. At that time the performers have agreed on a treatment procedure and a date of surgery. They have also shook hands and said goodbye. The consultant credits the doctor for following his tight script of procedures and for delivering a clear message. One audience notices: 'The performer [the doctor] did a very nice job. There is a lot of information to go through in our compressed patient interactions, and the performer managed to deliver the information details without getting tackled by the patient's attitude problems'.

The consultant then asked them, what would have happened if the doctor (an actor) had missed important bits of information. One of the participants answered: 'Then we definitely need to see or talk to the patient again'. She continued:

'These patients will make formal complaints about the quality of information; they will feel insecure and will have difficulty in letting us go, and perhaps they will turn down treatment offers. Basically, they won't have any trust in our system. We also need to pay attention to the patient's individual needs so that we can enter the provision of empathy in our medical records'.

The example demonstrates how the doctors, on the basis of some standards of communication, are extremely occupied with the delivery of detailed information about the patients' treatment possibilities and planned procedures. The more intimate matters concerning cancer illness and the effects on individuals' lives are not a part of the script of delivering a treatment plan. Hence, the script does not evoke empathy in a classic interpretation of the term; rather it provokes what I term controlled or restricted empathy. In the last quote, it is furthermore noteworthy how one of the per-

formers couples various demands to the delivery of information, including the provision of empathy and the ability to turn to individual needs of those who attend the clinic for help. It may also point to how emotional expression or behaviour, such as skills of (controlled) empathy or sympathy, in today's health care organisations needs to be performed in standardised ways in order to be recorded, audited and benchmarked across hospital department units. This thought is backed up in the interview data where a doctor ('David') stresses how the registration of the exhibition of empathy is connected to the clinic's annual budget negotiations with the hospital administration.

Second stage performance: The relative who is too emotionally involved

The second stage performance was staged with performers similar to the first play ('doctor', 'patient'), but with the addition of a family member. The stage script also resembled the first, but the key challenge was slightly altered. The script says:

'You are delayed in your outpatient program and have not been able to attend the waiting patient and her relative before now. You have not met the patient before (a colleague has conducted a biopsy the previous week and delivered a diagnosis of cancer on that basis)'

In addition, the performers are told that the patient has brought along her daughter. This daughter has a background in nursing, and she is very emotionally affected by her mother's diagnosis. The role-playing starts.

The daughter has some difficulty in understanding why a mastectomy (i.e., removal of the breast) is a necessary step to secure her mother's health outcome and she continually returns to this topic throughout the conversation. She also cries and offers some serious emotional-laden playacting. The doctor draws on various tricks to convey empathic responses to the daughter's emotional outbursts - for ex-

ample by asking open questions: 'How do you feel about receiving this information?' and 'Can you tell me about any of your thoughts right now?', and by showing that he has recognised her feelings: 'This is not easy, I know' and 'I understand that you feel upset right now'. However, the doctor is left somewhat confused 20 minutes into the role-play. By this time, he has already turned to the audience once to explain that 'a curtain has dropped between me, the patient and her illness'. He also says that 'the daughter clearly does not embrace my information'. The doctor then signals, that he is in trouble. He starts to blush and stumbles over his words and uses response cries such as 'um' and 'uh' and 'oh', which express that the interaction is in some kind of crisis. The result of a break down is serious, because the doctor risks losing 'face' in the presence of both fellow performers and audience. The consultant therefore breaks off the session to limit the extent of interaction damage and asks the performer directly what next step she proposes to reconstitute the emotional balance of those involved. The performer answers – probably in order to save his own show and 'to successfully stage his character' (Goffman, 1959: 203):

'In a real situation, I would ask the patient to take off her clothes and physically show her by touching her why it is necessary to remove her breast. And then I would explain how choices in surgery techniques have consequences for her survival of the cancer illness [...] And then I would tell her to leave her daughter at home the next time we meet.'

The last comment makes the audience laugh. It is no coincidence that performers and audience at this point start to make fun of the situation. Reconsider Goffman, when he writes that 'joshing sometimes occurs as a mean of releasing the tension caused by embarrassment' (Goffman, 1956: 271). This 'joshing' laughter functions as a sort of channel of distraction, which the performers and the audience can use in turn as a way of emphasising that 'this is not a real situation' or in order to declare that what

occurs now is not serious. The consultant then asks what was funny about the comment, and the audience explains that:

'The doctor just tries to do his job properly and stay in control of the situation and himself. The daughter and all her demands to our services, in contrast, make it extremely difficult for the doctor to deliver some important messages, which are necessary in order for the patient to make an informed choice in relation to future treatment options.'

The consultant's insistence on the seriousness of the conflict provides room for the above quoted response to the situation. It is somehow ironic that the performers (doctors) respond to the theatrical practice of the role-playing scenarios, where the 'roles' of doctors are deliberately rehearsed, as a kind of backstage area with respect to the front stage region of the clinic itself. However, the staged role-play scenarios are also proper Goffman situations – or front stage regions - in their own right. It presupposes a presentation of self on the part of the participants. The props used for presentations of 'doctor selves' include controlling the timing and style of the strenuous stage performance and an over-all capacity to play/act official/professional roles. The limitation of the analysis presented in this paper, if we attend to the workshop as a proper situation, is the lack of material on the withdrawal to the 'private sphere' where no colleagues or public are present.

Discussion

At the workshop, we see how effective communication is presented and adjusted so that performers can do a spotless performance of controlled 'empathy' on the front-stage. As such, the workshop allows us a peep-hole to the backstage region where doctors' 'empathic' behaviour is rehearsed in certain ways.

When analysing the workshop, it becomes clear how stage props, each of which defined slightly different versions of 'empathy', became available to the performers, who then tested their performances and checked for contradicting displays of empathy in the medical training facility. First, the program pamphlet, which was mailed to the participants of the workshop in advance, explained that after the training, doctors would be able to achieve a higher level of awareness of themselves and others through the use of communication tools. Moreover, doctors would be better able to structure and manage a difficult patient conversation. The material explained that especially empathic communication has been documented to lead to more effective consultation. Empathic communication thus became a toolkit to help doctors ensure an effective, time-compressed consultation. Objective procedures to manage inappropriate emotions in the medical context were at the heart of this toolkit as well as the notion of 'empathic communication'.

Second, the notion of 'empathic communication' took on a new twist during the consultants' oral presentations. Building upon concerns of 'empathic communication' as a means to recognise and manage the emotions of others, the presentations portrayed a balancing act between the professional conversation as a patient-centred conversation and the need for doctors to control this conversation. One way to understand this balancing act is to interpret the act in light of a traditional optic of power relations and medical relationships (e.g. Lupton, 2003a). According to this, the patient must be recognised as an emotional individual whose emotions must be controlled in order to facilitate the conventional objectives of the patient-doctor interaction. The purpose of the interaction then appears to be to void it of emotion. However, such a point of view would stand as a confirmation of a long held belief that emotions in organisations are irrational, idiosyncratic disturbances that are best kept undercover. In contrast, we need to pay attention to how the performers were also training in being able to manage 'deviant' feelings of others. Inappropriate emotions and not emotions per se, might cost disturbances in relation to the organisation of the

cancer pathways. A patient who is too aggressive or a patient who is too cold and not emotional enough (that is, the patient types presented in the two role-plays) might resist further clinical procedures and/or necessary treatment. A part of doctors' role is not only to make patients feel in a certain way but also to facilitate that these patients display their feelings according to some rules or standards of the situation.

Third, the team exercises added a therapeutic element to 'empathic communication'. It was considered essential to prevent conflicts in interactions through the use of for instance 'active listening' speech technique. Conflicts require time, and patient conversations in the cancer clinic are expected to take place in highly condensed time frames of approximately 15-20 minutes (exclusive the time spend on finding a consultancy room, looking for the nurse, collecting the patient record, etcetera).

Finally, through the role-playing scenarios, the professionals rehearsed the various aspects of 'empathic communication' in order to become ready to go frontstage and perform their newly adjusted skills. In its traditional configuration, the notion of empathy plays a significant role in the creation and maintenance of health care worker-patient relationships (e.g. Halpern, 2001). Techniques of empathic communication, such as 'active listening' and 'recognising responses', are intended to both contain emotions and allow them to vent. This was not exactly the case in the staged role-plays. When a patient's emotions are recognised, the main task of the doctor is to manage these emotions and to disengage them in an appropriate fashion. The doctor engages in the role-playing scenarios by delivering bits of value-free, emotionally neutral information. By treating facts as emotionally neutral, the doctors manage to separate emotions from the interaction and by performing this manoeuvre they may also avoid getting too emotionally involved in the interaction. They focused on pathology, possible treatment procedures and clinical details such as 'the exact size of the operation seam' or 'the colour of the radiation fluid used in the sentinel node technique'. In other words, emotions between and within the doctor and patient

were getting rigidly regulated by using formalised communication patterns of neutral speech to secure a scripted aim of distributing 'clean' information.

In a Goffmanian perspective, this backstage rehearsal is necessary to facilitate a smooth frontstage so that medical interactions achieve satisfactory patient outcome without too much emotional awkwardness. The shift in context from a frontstage (the clinic) to a backstage (the medical training facility) allows us to understand these performances as fabricated activities that sustain a working consensus on how effective communication, including empathic responses, are directed to be performed in accelerated cancer pathways. Returning to Hochschild's notion of 'emotional labour' (1983), we may view the doctors' own request for the training workshop as evidence of the fact that they recognise the management of emotions as an important, but difficult part of their job. However, the concept of emotional labour has a depth, which is not properly addressed through the training of communication, because it includes the work one does on oneself or the work doctors do to manage their own emotions. On this background we may see how the installation of communication skills – such as skills of empathy and sympathy – does not address the complexities of medical interactions in terms of offering relief to suffering individuals. Instead, these skills concern a standardised distribution of information and, hence, doctor selfregulatory exercises which reflect the imperatives of effective communication.

Conclusion

The paper had its point of departure in a configuration of today's medical services, which is characterised by attempts to both rationalise and humanise medical services. On this background, the paper demonstrated how communication is installed and trained as a means to control the conduct of doctors in medical interactions. The paper showed how communication skills are rehearsed at a medical training facility and it showed how certain skills, performed under the auspices of 'empathic communication', could be rehearsed as strategic components to support the medical baseline

work in accelerated medical services. Communication skills can be refined to secure formalised goals of speed and production effectivity. Patterns of empathic speech in which emotions are managed become important resources in providing customer service, increasing quality and so on.

The paper also scrutinised how the 'emotional availability' of doctors in interactions with patients becomes the aim of new forms of micro-management. More specifically, I argue that the workshop reveals how the broad notion of effective communication in the accelerated cancer pathways emerges as a means of cultivating the conduct of doctors. This research suggests that when empathic conducts are translated into clinical standards, they become available for qualitative measurement and audit processes. In contrast, Strauss et al. (1982, 1997) emphasised that sentimental work happens outside the rationalising realm of the medical world. Hochschild (1983) likewise argued that emotion work, which governs emotional exchanges in social interactions, is free of external observation (1983: 56). Contrary to this emphasis on emotional exchanges as hidden, informal activity, this paper shows that the behaviour of doctors in accelerated patient pathways is increasingly becoming a key issue of quality improvement initiatives and, therefore, a transparent property of health care organisations. During the above discussion on the training workshop, we saw how communication competence as a certain kind of behaviour is made both publicly assessable and accountable.

Likewise, doctor 'empathy' is a key concern of hospital managers, patients and policy makers. This study demonstrates how this concern includes an assessment of how empathic communication is delivered, by whom and with what effects. One must measure in patient records whether patients are treated with empathy and whether they are 'empathically' informed about diagnosis and treatment procedures. This demonstrates that 'humanistic' aspects of medical work now serve as an aim of efficiency improvement. The renewed focus on emotions in health care organisations is thus buttressed by mechanisms of control and surveillance, which are

all-the-more constraining because they are 'softer' (Harris et al., 2011). This revitalisation of emotions also means that a more salient role of emotions in organisations should not simply be taken at face value as in fact more 'human' but rather as an expression of the fact that emotions are considered obstacles to improved organisational efficiency.

The study thus adds to previous studies on the significance of current reforming drives in public health care services and their effects upon the conduct of health care workers (Olesen and Bone, 1998; Fitzgerald and Ferlie, 2000; Bone, 2002; Nettleton et al., 2008; Waring et al., 2011). Discussions about reforming drives in health care organisations and the development of techniques to cultivate the conduct of professionals should not be seen in isolation from attempts to understand the governance of individuals' feelings in other contexts of employment. For example, Terpe and Paierl (2010) explore the consequences of new service-oriented administrative reforms that have recently been introduced in labour administrations. They analyse how the emotion work of German administrative employees is influenced by the reforms political and managerial aims. However, additional work is required to build a fuller understanding of how the emotional landscape of employees in the public sector changes through reforms of old administrative structures and aims of transforming the conduct of public employees towards smoother and apparently more 'human' manners and behaviour.

Chapter 7: Doctors' emotional experience and challenges in accelerated medical work

Purpose: This paper analyses doctors' personal biographies of emotional experience and challenges in accelerated medical work. The aim of the empirical inquiry is to focus on the ways that doctors relate emotions to their understanding of professionalism and principles of standardisation and speed.

Methodology: Drawing upon a small series of semi-structured interviews (N=14) with doctors working in a cancer clinic at a major public university hospital in Denmark, the paper adopts a constructivist framework to analyse doctors' own understandings of emotions and the management of emotions in (accelerated) social interaction.

Findings: As cancer trajectories are increasingly accelerated, the available time for doctors' contact with patients has been shortened, while the range and type of managing emotions in the trajectories has simultaneously increased. This is shown to challenge doctors' emotional codes of conduct and to require new forms of emotional regulation in medical encounters.

Practical implications: The study addresses how doctors' experience and understanding of emotions are affected by recent health care reforms to rearrange cancer services.

Originality: The paper draws together previous research on emotions in health care organisations to explore doctors' own view on recent rearrangements of cancer services and to understand their experiences with these rearrangements in relation to the connection between emotion, speed and processes of standardisation. While

emotions and rational, clinical work activities are often taken to be opposites, the paper addresses that the dimensions imply each other and that affection is frequently brough to rational use in accelerated medical work.

Keywords: Doctors, emotions, rationality, accelerated medical work, public health care services.

Introduction

The American novelist John Updike's (1995) introduction to the medical novel 'House of God' discusses a widely held belief about doctors' emotional distance to patients in medical encounters.

'We expect the world of doctors. Out of our need, we revere them; we imagine that their training and expertise and saintly dedication have purged them of all the uncertainty, trepidation, and disgust that we would feel in their position...For them, the flesh and its diseases have been abstracted, rendered coolly diagrammatic and quickly subject infallible diagnosis and effective treatment' (1995: Introduction).

The paradigmatic image of 'detached' doctors, as highlighted in the quote, is manifest in sociological work on doctors and presents an image of the medical profession that is shared by many lay people as well. The core assumption is, as Updike emphasises, that clinical affairs are being dealt with by doctors in a way that is 'coolly diagrammatic'. Abbott argues how the skilful doctor must dissociate emotion from reason and apply medicine as a rational and objective knowledge system to particular cases (Abbott, 1988). And Kirmayer (1988) writes about how doctors have 'exaggerated standards for rationality based on distancing from bodily feelings and emotion'

(1988: 63). However, this separation of rationality from emotion has been widely challenged and the importance of emotion to basis purposes and values of organisational life has been addressed for the last two decades by literature on emotions in health care (James, 1989, 1993; Bolton, 2001, 2005). As for example James (1989) writes, it is easy to be persuaded about that social expressions of emotion directly contrast the predictable, logical behaviour associated with 'rationality'. In opposition to this approach, she proposes that emotions and display of emotion is not simply enemies to rationality and rational action. Instead, emotions are rational parts of everyday life as purposive, meaningful responses to specific situations.

This literature emphasises that professionals (doctors, nurses) are required to be empathic, responsive and feeling individuals 'for the patients' sake'; i.e., an important part of their job is to make themselves emotional available to others and comfort people who are in need of emotional support. Additionally, it has been argued that emotions are an important part of completing medical work tasks effectively, without too many disruptions and vexations. According to Mark (2005) 'the dominance of rationality serves both a scientific and emotional purpose' in the health care setting (2005: 279). She explains how the former 'provides the cognitive means by which emotionally unacceptable procedures and activities are allowed to occur to individuals as patients' (ibid) and argues that the organisation of health care work today needs further attention. In particular, she finds that the continual 'juxtaposition of emotion with scientific rationality in the provision of health care' (2005: 278) has great consequences for both individuals and the organisation.

If one departs from Mark's approach and takes into consideration the pace of reform and rationalisation efforts within health care, it seems reasonable not only to readdress this 'juxtaposition' but also to focus on the complex intertwinement of emotion and rationality in relation to current ways of organising medical services. This is done in this paper by discussing the findings from an empirical study, using semi-structured interview method with 14 doctors working in a cancer clinic about

how they relate emotion and techniques of managing emotion to their understanding of professionalism and principle of standardisation and speed in clinical work activities. The context of my case is cancer treatment in Denmark, which has recently been substantially reorganised. Introductions of accelerated cancer programs have resulted in a restructuring of cancer pathways, informed by managerial and clinical purposes of reducing waiting times and organisational delays, speeding up processes of diagnosing and treatment, and strengthening the coordination of patient treatment between hospital unit and sectors. I argue that these new structural rearrangements affect the forms emotional expression takes in medical work.

Through analyses of doctors' emotional experience and challenges in their work, the paper shows that emotions not only constitute important social elements of medical work, but that emotion and emotional display is also frequently enacted in a rationalised way to help doctors ensure the progress and efficiency of accelerated cancer pathways. Emotion constitutes new ways of providing effective and efficient health services in the current situation of limited economic resources to public health care and a political and managerial climate influenced by New Public Management principles of efficiency, accountability, cost reduction and speed (see for example Harrison, 2002; Harrison and McDonald, 2008). The paper discusses the implications of such intimate connections between emotions, speed and standardisation and argues that this contributes to existing debates on emotion and emotion management in public organisations, such as recent discussion concerning how employees are getting trained or instructed in managing rational emotions while providing 'compassionate', 'personal care' (Du Gay, 2008; Terpe and Paierl, 2010).

The paper is organised in the following way. First, the theoretical frame, which informs the discussion of the role of emotion and the type of emotional transactions in accelerated medical work, is introduced. Next, the method of the empirical inquiry is presented. Then follows the analysis of the emotional challenges that doctors experience and make sense of in their everyday work and the strategies they em-

ploy to handle these. Finally, the paper wraps up with concluding remarks about the complex interweaving of reason (rationality) and emotion that motivates doctors' feelings, experiences and thoughts.

Theoretical framework

'To drop the tools of rationality is to gain access to lightness in the form of intuitions, feelings, stories, improvisation, experience, imagination, active listening, awareness in the moment, novel words, and empathy. All of these non-logical activities enable people to solve problems and enact their potential'.

This quote from Weick's (2007: 15) critique is based on his studies of wild land fire-fighters and the use of rational tools in social action, as well as the emphasis attached to reason and rationality in existing studies on work and organisations. His critique suggests that emotion and rationality belong to two different organisation spheres, where the latter hems in the true potential of the former. The assumption is by no means unique. Authors have suggested since the beginning of the modern period that 'emotion became known as the enemy to rational processes instead of its ally' (Douglas and Ney 1998: 80). The consequences are that characteristics of the conduct of organisational life such as empathy, active listening and imagination (as pointed out by Weick) have been forcefully divided from supposedly more 'logical' or 'rational' activities, which otherwise characterise organisations and their basic norms and practices.

This paper follows the work of Putnam and Mumby (1993), among others, which presents emotion and managements of the emotions as entirely compatible with instrumental goal orientation and instituted purposes and values in organisations (see also, Albrow, 1994; Ashforth and Humphrey, 1995; Fineman, 2006). Putnam

and Mumby set forth a way to position emotion as 'central to the process of organizing and as integral to participation in organisational life' (1993: 36). They write that emotion 'is not simply an adjunct to work; rather it is the process through which members constitute their work environment by negotiating a shared reality' (ibid.). In this way, organisational practices can be viewed as enmeshing the rational and the emotional: they are two sides of the same coin.

In a health organisation context, the perspective also challenges dualistic analyses that separate compassion from detachment, improvisation from routine, objectivity from involvement and emotion from authentic feelings in the provision of health care services. Professional demeanour, pretence of sincere concern and welcoming smiles all facilitate and lubricate medical work procedures, including social and psychological aspects of care, which are often presented by practitioners as the 'soft' dimensions of their work (see 'cancer care for the whole patient' in Adler and Page, 2008). The health organisation context defines norms of feelings and facial and bodily display of emotion that doctors are required to use in patient-doctor interactions. As Fineman (1993) argues, these particular emotion codes are internalised as part of the conduct of doctors and are often promoted as a part of the hidden curriculum of medical training (Smith and Kleinman, 1989).

One of these emotional codes may be characterised as *affective neutrality* (Parsons, 1951: 61). Nowhere is this ideal of detachment so firmly entrenched as it is in medicine. Affective neutrality refers to self-restraint and self-discipline in relation to the amount of emotion or emotional display that is expected or appropriate to 'give off' in social interactions and how one can stay 'cool'. According to Parsons, affective neutrality is a significant feature of doctors' appearance. As such, the taming of doctors' behaviour is a necessary precondition for intimate interaction with others (patients, relatives) and one that enables doctors to attend to the complaints of suffering individuals in an reasonable and reliable way, stripped of personal (i.e., individual) bias. Parsons writes that by defining the doctors' ethos in this way, 'it is possible

to overcome or minimize resistances which might well otherwise prove fatal to the possibility of doing the job at all' (1951: 459). A number of 'control mechanisms' or 'technologies of government' (Miller and Rose, 2008) work to frame and reframe this particular ethos in practice, for example, through the self-presentation of the doctor as having a 'matter of fact' attitude to her patients, without undue personal involvement.

Hochschild's emotion theory (1979, 1983, 1998) presents a different understanding of the role of emotions and emotion management in health care, which includes strategic and manipulative control of others' feelings as part of service workers' job. Hochschild's main concern is with the social engineering of feelings as resources or commodities, which can be reproduced and exploited as part of the new (late-) capitalist spirit. Her term 'emotional labour' indicates that feelings of employers' are managed to 'create a publicly observable facial and bodily display (1983: 7). In the service industry, workers' feelings (conceptualised as feelings derived from their private spheres) are becoming public available and consumed by customers as part of a commoditised social interaction. Hence, in Hochschild's theory there is a discrepancy between 'sincere', 'inner' emotions, moods and their display and more rational, goal oriented actions in public service work. It is thus questionable whether the term 'emotional labour' actually captures the complexity of emotion and rational emotional transactions in health care (for a thorough critique of Hochschild's theory, see also Bolton, 2005). Where Hochschild frames the commoditization of emotion in labour, Strauss et al. (1982) presents a less politicised approach to the role of emotions in work, where the object being worked on 'is alive, sentient, reacting' (1982: 254). In a larger study on the social organisation of medical work the authors describe the organisation of so-called 'sentimental work'. Sentimental work involves 'staff members' and patients' maintaining of composure, keeping up of spirits of courage' (1982: 255). In this way, sentimental work is understood as emotional transactions or work completed to manage the emotions of patients and others (for example the doctor's own emotions) in illness trajectories in public hospitals. In contrast to

Hochschild's perspective, Strauss et al. suggest that there is nothing extraordinary or alienating in the provision of sentimental services. It is a necessary type of work undertaken by doctors, among others, to accomplish the 'real' medical tasks, such as taking a blood pressure or cleaning an operation wound. Strauss et al. argue that the impetus for doing emotional work has been profoundly affected by historical changes in the medical field. The paper proceeds from the assumption that current structural changes of cancer services represent a challenge to the way emotions and emotion management has previously been part of medical work, both in the sense of the required ethos and the emotional work associated with practically enacting this ethos.

Method

The sample analysed in this paper is 14 (N = 14) semi-structured interviews with doctors in a cancer unit at a public hospital in Denmark. The interview sample was collected as part of a larger study examining the effects of recent changes in the management of Danish health care services to hospital doctors. The larger study was carried out between June 2009 and January 2011.

Procedure

Access to the cancer clinic was gained through the clinic's head of research. This person also provided the contact information for the doctors. Invitations, accompanied by background information about the study, were distributed via email, asking doctors in the clinic to participate in an interview study on their experience and understanding of medical work in accelerated treatment programs. The choice of location for the interviews was decided by the doctors. The interviews were conducted in consultancy rooms in the outpatient clinic, in office facilities of the clinic and in a staff meeting room. Meeting in the staff room proved to be difficult, and this location was not used after the first interview. The first interview was disturbed by nurses entering the room for a cup of coffee. While others were in the room, the respondent was dis-

tracted, lost track of where she was in her stories and had difficulty sharing the emotional aspects of her work, as those parts of the interview were too sensitive to share with external 'intruders'. Another obstruction worth mentioning was the set time-frame for the interviews. All the interviews were to be conducted while the clinic was open, which meant, as one of the respondents pointed out, that the interviews took the doctors' time away from the patients. We were thus required to adhere strictly to the allotted time. The interviews were scheduled in the respondents' daily work roster and they were set to last between 20-30 minutes. The observation emphasises how time management is of utmost importance for medical staff. However, the importance of sticking to the allotted time might have two functions. It enables the doctor to see many patients in a short period, but it might also be used in medical practice as a strategy for maintaining authority and professional detachment (Lupton, 2003a). In relation to the interviews, this strategy was employed in the end of the interviews, where the respondents started looking at their watch or began to focus on the importance of returning to their patients.

In the interviews I had no intention of trying to trace what the participating subjects 'really' feel or to determine the authenticity and 'under the surface' feelings of the doctors. Instead, I identify different layers in the way doctors understand and present emotional aspects of their work to others. This corresponds to Goffman's (1974) work, in which he explores how and when something is framed as the 'real' thing and circumstances in which something is said and something else is left unspoken. My initial interviewing guide was constructed so as to provide an understanding of the doctors' views on and responses to their working lives, and to understand how these people frame and reframe emotional injunctions to their work. I was also interested in 'how it feels being a doctor' in a current environment of New Public Management and principles of efficiency, accountability, cost reduction and speed. To my knowledge only few qualitative studies (except from Lupton, 1997; 1998 and Nettleton et al., 2008) have sought the feelings of doctors upon these issues.

The 14 qualitative interviews were all audio-taped and transcribed. The identities of the interviewees quoted in this paper have been concealed for ethical concerns. To ensure this anonymity, some statements and other empirical evidence in this paper have been adjusted, so they cannot be trace to individual staff members in the cancer clinic. My primary interest was to listen to the discursive aspects of emotions and emotional challenges as these were articulated through the interviews. Hence a discursive approach, informed by a constructivist perspective, was employed on exploring the interviewees' views and experiences of reality, when these were articulated and made sense of to others (the interviewer).

Emotions in accelerated cancer care

So how did the doctors relate emotions to their professional work activities? The next section presents the discourses used by the participants and it provides examples of the ways that doctors describe and relate emotion to their understanding of professionalism and principles of standardisation and speed in the accelerated cancer pathways. It furthermore explores how the participants are affected by recent attempts to organise cancer treatment into streamlined production flows.

The professional role of a doctor

In the interviews, the doctors were asked to characterise their professional role in relation to their substantial work tasks in the accelerated cancer pathways. Many of them described their role in terms of 'providing efficient care' (Deborah); 'facilitating optimal treatment' (Michael); 'making the greatest effort possible in regard to the individual patient' (Erving); 'curing the patient and leaving her with a minimum of physical distress' (Sarah) and 'to provide treatment options for patients who have complaints within the area of our medical expertise' (David). In addition to providing optimal care, professionalism, according to the interviewees, consists of 'dealing with the specialised sequences of the patient pathways as good as possible' (Lucy). Mi-

chael adds another 'soft' element to the 'technical' descriptions of his professionalism:

'When one is serious ill, there is a need for somebody who actively listens to ones' complaints. This is a profound element of my professionalism, besides all of the other vital elements in the patient pathway, such as the observance of waiting times, record-keeping in our clinical databases, capacity management, etc.' (Michael)

When Michael lists some of the different work tasks that constitute his daily work, there is no significant difference in the emphasis on 'active listening' in patient-doctor interaction, keeping track of the clinic's waiting times to see a doctor, and diagnostic test methods. All elements are described as vital parts of his work. Even the severely criticised 'rituals of verification' (Power, 1997), such as rigorous data recording, is presented here as a proper professional task. Some of the doctors explained how the ability to follow clinical guidelines and recommendations, which define the medical procedures of the cancer pathways, are perceived as genuine professional tasks and something which improve the clinical quality of their work.

None of these explanations about professionalism and the doctor's role are very controversial and confirm by and large the picture provided by Parsons on the doctors' functional competences as an entanglement of technical, and social and psychological competences (Parsons, 1951). Alongside the biomedical aspects that fall under the technical competences of the doctor, such as the task of removing a cancer tumour, there are more social aspects implied in the medical job, such as emotionally preparing the patients for bodily interventions or be emotionally available to patients through techniques of 'active listening'. This wide spectrum of job tasks also affected what the doctors thought were gratifying in their work. The doctors talked about the satisfactory feeling of performing surgery and how it feels almost exhilarat-

ing to make a clean cut in the flesh and feel how the 'scalpel easily separate tissue and skin and how the sentinel node snap into ones fingers' (Eva). They also described encounters with patients as gratifying if they felt that the patients were open to an intervention. For example, one doctor said: 'One of the greatest pleasures in my work is to attend a patient in the outpatient clinic who immediately is very offensive and then slowly through our communication opens up and tells me what she actually feels' (Sarah). Most of the interviewees mentioned how patients who easily agree to a proposed treatment sequence cause feelings of contentment. The opposite can also occur; for example, a patient may respond in a way that the doctors experience as emotionally inappropriate. The interviewees talked about troubles with 'misfitting' patients and how these people might cause general frustration among the professionals when they attend the clinic for help.

'In cases where the patient declines your guidance, then she can be categorised as a kind of a misfit. Our treatment program encourages individuals to follow its recommendations. In the case of a patient who reacts pathologically, my professional task is to stay calm and not get personally affected – for example when a patient insults me or if she refuses to follow my leads'. (Paul)

The separation of emotional display from the presentation of a professional appearance is experienced as pivotal in situations where patients do not follow advices or react in ways that are perceived emotionally 'pathological'. One doctor described how she 'need(s) to be disengaged on a personal level while providing effective care' (Sarah). In the same vein, Erving explained that he ideally must separate emotion and rationality in his clinical tasks in relations to his professional codes of conduct, and he noted how this separation can be difficult to uphold in practice:

'My professionalism demands that I am able to separate the emotional aspects of my work from the more rational aspects of my work [...] However, I must admit that I emotionally connect differently with one patient than another and, therefore, also might behave differently' (Erving)

The explanation in the quote, which was heard from other doctors in the sample as well, indicates that (emotional) codes of conduct are defining the doctors' display of professional appearance. The doctors expect themselves to behave in a certain way to meet colleagues' and lay peoples' expectations of 'the world of doctors' (cf. Updike, op.cit.). Emotionality in this interpretation involves a loss of control over one's emotions; to be 'gripped by emotions' and it might as well involve irrationality. Erving was not the only doctor who recollected particular patients and explained how interactions with them challenged his ability to, in his own words, 'separate emotional aspects from rational aspects of my work'. When asked if it is considered important to separate emotions from doing rational clinical work activities, many of the interviewees strongly supported the notion that ideally one must always be able to separate the two. By saying this, the doctors reconfirm and promote the ideal of affective neutrality set forth by Parsons (1951). However, this ideal is continually getting challenged. The doctors' experiences do not leave them emotionally 'coolly diagrammatic', as discussed in the next section. Some 'troubling' persons not only challenge the doctors' ability to sustain a professional appearance and to go public with a 'smooth face', but these patients may also cause confusion in the doctors' way of attending to particular situations in the cancer clinic.

Emotional experience and challenges

While the interviewees generally agreed that that they intend to treat all patients equally (for this is regarded as an essential and indispensable dimension of their professionalism), they said that emotionally, they responded differently to some patients:

'Some patients have easier access to your stomach pit than others' (Mary); 'some people talk more directly to your heart, and they unlock your emotions' (Sarah) and 'some of my patients are more difficult to shake off than others' (Monica). Troubling patients, regarded by the interviewees as patients whom you "carry with you" after office hours, are people who somehow affect the way doctors feel. Some of the interviewees suggested that their emotional state was affected partly by the 'emotional climate' and partly by particular illness stories. By emotional climate, I refer to the interviewees' conceptualisation of the foundation for emotional attachment to others. Doctors also used the term 'chemistry' to relate the idea of emotional climate in the interviews: 'Sometimes, no chemistry is present, and you just have to accept that and go on with the line of work' (Michael); 'Of course I have cases where the chemistry is awful, no matter the efforts I make to improve it' (Deborah); and 'You can easily feel if the chemistry is just right' (Eva). When the appropriate climate is missing, doctors describe the interaction in more generalised terms: 'The other day, I had an interaction with a patient who really didn't like me. There was no chemistry between us at all' (Sarah) and 'She [the patient] was on a totally different wavelength than me' (Paul). Although the emotional climate is commonly recognised by the interviewees as important and something one should work on to improve, poor 'chemistry' is also recognised as a part of the job, where sometimes the doctor or the patient requests that another person take over the responsibility of the treatment.

The doctors explained that particular illness stories affected them and caused an emotional response: 'It is either the very young patients who get cancer or patients whose illness trajectory is wretched' (Susan); 'Young people creep into you. It is a common concern for all of us, and one can register it in the corridors of the clinic' (David); 'When you face a young mother in the clinic who is 34 years old and who has just given birth to her first child, it is impossible not to think when you enter your front door after work "oh, life is just unfair to certain people" (Lucy). To disclose a cancer diagnosis to a young woman in her early thirties is emotionally strain-

ing 'because you know her prognosis is really bad and...well, no matter how much you try to stay detached, you somehow get affected' (Erving). Most of the stories that the doctors considered troubling included stories about young women who were considered 'really ill' or 'genuinely ill'. Young women with an aggressive, advanced stage tumour are regarded, as one of the doctor's says, as 'the worst patients I can possibly think of to have on my outpatient clinic schedule' (Eva).

A certain category of 'demanding' patients are commonly described as influencing how the doctor feels – both with regard to her own role as provider of public health services and with regard to the role of the patient as consumer of these services. As some of the doctors explain, 'I get really angry when resourceful people in totally unrestrained manners try to enrich themselves with our services and push themselves forward in the service line' (David) and 'The archetype of a demanding patient is a colleague or a school teacher who attend the clinic with objectively unjustified demands for treatment' (Michael). The doctors commented that demanding patients are 'difficult to handle' (David); 'intimidating' (Brian) and 'they really strike me with contempt' (Susan). The doctors' feelings of disgust are regarded as strictly forbidden feelings to display. These feelings are specifically manifest in situations where patients behave as demanding, well-oriented consumers who have explicit expectations to the quality of offered services. A way to attend to the needs of demanding 'consumer-patients' may be to attend to them as people who has social or psychological problems instead of attending them as people who are in need of medical services (see also how GP's attend people with MUS in primary care in Mik-Meyer, 2010; 2011; Mik-Meyer and Roelsgaard Obling, 2012). Hence, it also become easier to deny them access to the clinic's accelerated services, proposing that these people look for other care options. This is for example the case when people ask for cosmetic surgery, which is a service that is not financed by the public health care system in Denmark.

Techniques to manage the emotions

In the last section, I briefly touched upon categories of patients and situations that were described as particularly troubling by the doctors. The doctors believed that some people interfered with their plans and procedures and were extra-ordinary time consuming. As one of the interviewee's explains:

'Some people respond very pathologically [...] they also make your day into rubbish because your time schedule becomes completely ruined by the work you have to do to calm the person down to be able to get on with your duties' (Susan).

One way of dealing with deviant cases is to master techniques of emotion management, which are techniques frequently used to prevent the occurrence of emotional incidents. Management of the emotions might be used when patients or relatives in the accelerated cancer pathways react to malignant diagnoses by becoming extremely upset or angry or, perhaps even worse, does not show any emotion at all. For example, a doctor explains how he 'always uses the same explanatory models and ways of presenting bad news, which he seldom deviates from, because the models have proved themselves effective as techniques' (Michael). As described in the following quote, doctors are expected to appear engaged and responsive and to be able to look at a situation from the patients' perspective.

'You also expect your real estate agent, when he shows you around in a house, to be engaged every time the two of you meet. He must show you all of the rooms, and he must be responsive to every individual need. He also needs to appear cheerful and optimistic' (Paul).

This doctor explained how, through a particular appearance, he is able to create a more comfortable situation for the patients and make them feel secure in the interaction. Neglect of emotion management might threaten the line of prescheduled diagnostic and treatment procedures, which are defined in the cancer pathways in relation to a set of standard trajectory descriptions. In the interest of both the organisation and the individual patient, the doctors are therefore willing to take on serious 'work' (Strauss et al., 1982) to maintain composure and thus be able to accomplish the prescribed medical tasks. One basic defensive measure that the doctor can employ if a situation develops is to 'start putting forward all of the practical details that need to be arranged' (Monica). To pay attention to practical details in the medical interaction is to focus on 'we need to plan your surgery' (Sara) and 'focus on all of the stuff we actually can do to cure them instead of focusing on their misery' (Monica). Another technique to address the situation might be to suggest a second treatment option. As one of the interviewees (Erving) explained, 'you might be compelled to suggest an additional examination to moderate them a bit [...] to cool them down'. The work undertaken by the doctors to maintain composure in the cancer pathways

also include work or efforts in relation to upholding the appearance of an emotionally detached doctor. When asked if it is important to manage one's emotions, the interviewees gave the description of the paradigmatic doctor as a person who must always manage to stay personally (i.e., individually) detached, even in a situation where this detachment is challenged by the doctor's own feelings (compassion, sorrow).

'Deep inside, I just want to cry with them [cancer patients], but for the sake of providing good care and of course to protect myself, I must always pretend to be calm' (Ann).

Some of the interviewees' identified a certain technique to manage their feelings that prevent them from becoming emotionally involved with individuals who are serious

ill. Eva explains how she manages the misery of suffering patients by imagining that 'after I have successfully removed the patient's tumour ...she will experience an uncomplicated healing process'. This technique becomes a technique of self-defence because, as Eva also explains: 'I make this way of thinking even in cases where I know that no cure is possible and the prognosis of the patient is really bad'. Thinking of colleagues' abilities to cure patients can provide a buffer to precarious situations, where the ability to balance one's own needs with the patients' needs becomes challenged. These techniques may be used more frequently in accelerated trajectories than in traditional trajectories, because there are many different professional faces involved in the patients' diagnostic and treatment procedures in a very short time interval. Hence, there is always a colleague further down 'the production line', which the doctor can refer to as making a difference in relation to the patient's curing possibilities.

Another technique the doctors can evoke is to focus on professional ways of (dis-)engaging with others.

'Your whole body screams in disgust with a person who is literally shouting at you and calls you all kinds of nasty names [...] nevertheless you must try to lend the person assistance, following the norms of your professional codes, which tell you to stay calm' (Mary).

The quote reveals one type of dirty work, which according to Strauss et al. (1997), consist of tasks 'so exhausting or stressful as to tip forward the non-gratifying or so ultimately the dirty side of work' (1997: 248). The doctor explained that she can rely on her codes of conduct as a strategy for moderating or minimising the experience of her more difficult tasks, as a mean to distance herself from dirty aspects of medical work, and as a way to sustain her 'professional involvement' without being gripped by personal, inappropriate feelings (Mary).

In sum, the discourses suggest that the doctors frequently use various techniques to manage their emotions. This includes techniques to: 1) help doctors to ensure the progress of diagnostic and treatment procedures in the cancer pathways, and 2) help doctors to sustain their professional 'ethos'. The former implies that the doctors rely on these techniques when a patient needs to be emotionally regulated to 'fit' better into the prescribed procedures. The latter involves that the doctor herself apply techniques to address and overcome challenging, stressful, or discrediting situations.

Emotions in standardised work procedures

Discourses of standardisation were frequently employed when the doctors explained the organisation of their work in the cancer pathways. Several of them referred to the importance of consistently following the same procedures, and one doctor mentioned, 'it is important that our patient pathways are always standardised and effective' (Sarah). This idea was supported by Mary, who mentioned 'that our patients should experience that everything is organised following a rigid plan of procedures'. It is also supported by Deborah when she said that 'at least we have a plan we can offer them'. During the interviews, it became obvious that the clinical guidelines and recommendations that doctors draw upon to make decisions about diagnostic and possible treatment procedures not only structure the sequences of action in the pathways but also provide guidance on how one should feel in given situations. For example, one of the interviewees (Eva) described how she 'sticks to my guidelines because I can't always grasp young women with cancer and their gamut of feelings', and another interviewee explains how 'one can always rely on the guidelines if one feels a bit uneasy' (Brian). Hence, the guidelines are used as strategic protective shields to employ in situations when the doctors become emotionally challenged. The guidelines help to reinforce the paradigmatic (self) image of the doctor when this image When the interbecomes threatened by dissonant feelings.

viewees were asked to comment on current political and managerial climate influences by principles of New Public Management (efficiency, standardisation, accountability, cost reduction and speed), they articulated feelings of ambivalence. A recurrent theme was that 'one wins something and loses something when standardisations are implemented' (Lucy). This theme reappeared in various negative responses, such as: 'We practice a lot of roundabout routine procedures that continuously remove us away from the patient' (Paul); 'Our patient pathways remind me of a sausage factory' (Susan); and 'I feel like an old record player that plays the same tunes again and again' (Paul). The image of a Taylor-inspired manufacturing line is also present in the next quote:

'There is no doubt, that the prize of standardisation of health care services is high [...] from being a matter between you and the patient, medical work is now more or less standard manufactory line work, where your job is to fix a few things along the line. You can then be lucky to be in charge of a significant part of the line. Additionally, things can be organised in inopportune ways, which means your job is reduced to replacing a defect bolt and that's about it' (Lucy).

The interviewees understand themselves as becoming emotionally detached from their work in several distinct ways. This detachment is significant in cases where they cognitively cannot separate one patient from another, which was apparent in statements that the doctors made about the restructured pathways into streamlined service flow, where the patients emanate in one another. Because most clinical decisions may have already been made by other colleagues, the individual doctor is left to tinker with only routine matters. As explained by one of the interviewees, 'To be left with just a small part of the assembly line leaves me unsatisfied'(Lucy). The feelings of 'too much detachment' are also present in cases where the interviewees' themselves feel physically distant from the patient, as in cases where they have not had the op-

portunity to examine the patient themselves. In the interview sample, some of the doctors explained that it feels frustrating if the initial meeting with a patient occurs in the operation theatre, where the patient is anaesthetised and fixed to the hydraulic surgical table. In these cases, the doctors felt that they had been reduced to the role of a mere scalpel: 'Sometimes in the operation theatre I try to recall the patient, who is under anaesthetics in front of me [...] it actually bothers me that she is already out of my radar, and I can't even recall a minimum of the person' (Brian).

The 'dislocation of the case from the patient's bedside', also noted by Atkinson (1995: 149), makes up a profound element of contemporary medical practice, but it furthermore leaves a feeling of uneasiness, as explained by several interviewees of this study, who felt themselves detached from their patients if they only have to deal with the pure 'cases'. This remoteness from the 'whole' patient leads to disturbing feelings: 'Sometimes I experience this confusion of loads of cases somehow alienating' (Monica) and 'I feel unsecure when my professional decision making is removed from the physical presence of the patient' (Deborah). These examples support the argument of Rose (2007), when he claims that today doctors are 'hemmed in and constrained by the requirements for the use of standardised, corporately framed diagnostic and prescribing procedures' (2007: 11). Furthermore, it widens the empirical scope of his argument in showing that 'to be hemmed' in causes various feelings of frustration among the doctors. Above considerations suggest that the standardisation of work activities in the accelerated cancer pathways hinders the development of a total impression of the patient, and consequently, it becomes difficult for doctors to assess the emotional state of patients and adjust their professional ethos to this. 'To be left with only a smaller part of the manufacturing line', as one of the interviewees' explained, might also effect the type of emotion management a doctor can provide in a split second and under conditions where the 'cases' are separated from processes of clinical decision making and bodily examinations.

Conclusion

A central aim of this paper was to identify doctors' emotional experience and challenges in accelerated medical work. Through an analysis of the ways that doctors relate emotions to their understanding of professionalism and to work activities that characterise rational organisational forms, the paper has investigated the complex interweaving of reason (rationality) and emotion that motivates doctors' experiences and thoughts.

A key finding is that doctors identified emotions in terms of resources. As resources, emotions were not experienced as being antithetical to the doctors' provision of medical service. Instead, emotions were approached as necessary components in the production of objective, reliable and standardised care. Doctors' emotional availability to patients is thus not opposed to or separate from the 'rational clinic' (Wulff and Goetzsche, 2000) or the provision of 'rational diagnosis and treatment': professional medical work is a delicate combination of both. It is therefore next to impossible to separate procedures that require rational versus emotional responses in doctors' everyday work in the cancer clinic. In this light, rational procedures of diagnostic and treatment in the accelerated cancer pathways are not predicated upon the curbing of emotions, but on the management of 'reasonable' emotions. This study therefore suggests that reason (rationality) is not opposed to emotion, but both are part of effective and efficient cancer treatment, and hence both dimensions 'make up' the doctors' minds.

The analysis presented in the paper highlights some of the emotional codes of conduct, that are characterised by the participants as a form of what Hochschild terms 'emotional dictionary' (Hochschild, 1998), which defines those feelings that are appropriate to display in specific circumstances. In this way Updike's paradigmatic image of doctors and medical activities as 'coolly diagrammatic' are to be understood as an shared ethos which includes an understanding of what a doctor should feel, how she emotionally should be available to another person, and

how she must sustain this idealistic appearance. The doctors' well-established ideas presented in this paper about emotional codes of conduct include also the Parsonian ideal of affective neutrality, which is framed by the participants as a precondition for successful and effective patient-doctor interaction. Affective neutrality allows doctors to attend to patients individually, but protects the doctors from exhaustive personal involvement. However, this ideal does not prevent doctors from being emotional challenged by 'troubling' individuals. For example, the doctors acknowledge that it takes considerable time and effort to provide an emotional climate and accommodate demanding patients. Compressed work schedules and accelerated timetables often make improvement of this climate difficult, even impossible, and the result is feelings of a failed encounter on behalf of the doctor

Another key finding is that it takes serious work for doctors to reframe the paradigmatic image of doctors as detached observers in emotionally challenging situations. One way that the doctors are able to do this is through the use of emotion management techniques that, in a Goffmanian sense, makes it possible for the professionals to stage a particular character and thereby 'saving the performers' show' (Goffman, 1959: 207). The participants of this study were aware of several techniques of emotion management, which they frequently use to manage their own and others' emotions in the accelerated cancer pathways. This work is done in situations typically where patients display too much emotion or where patients demonstrate an inappropriate display of emotion. The latter also includes the display of inappropriate emotions, which is experienced by the participants as equally disruptive as a display of too much emotion. It is therefore not enough to be a detached observer and sustain affective neutrality in patient-doctor interactions: doctors need to practice serious micro-regulated tasks of emotion management to maintain composure in these interactions, even in situations where there is very limited time to practice this kind of work.

The interviewed doctors were sensitive to a current environment of New Public Management and principles of efficiency, accountability, standardisation, cost

reduction and speed which have been implemented in health care organisations during the last two to three decades. Most of them agreed that there is a price to standardisation of health services, which they specifically experienced as a form of detachment where they gradually become removed from their traditional field of intervention, which is the patients' bodies, feelings and thoughts. However, the doctors must adhere to behavioural norms: they must demonstrate empathic, responsive or caring behaviour even in situations where their duties require them to focus on only a smaller part of the patient trajectory or in cases where their personal contact with a patient is very limited. Importantly, this paper suggests that the rearrangement of the cancer pathways into streamlined manufacturing-like processes, which include a reduction of waiting times and organisational delays and accelerated procedures of diagnostic and treatment of patients, complicate doctors' ability to incorporate emotion into the stream of medical care in a 'rational way'. The streamlining of the cancer pathways accentuates the importance of skilful emotion management because the actual encounters with patients are generally reduced. However, there is only a restricted time for putting this increased emotion management into practice and the issue is only addressed in an individualised way, i.e., as the doctors' personal reflections. The rearrangement has led to conflicting demands of the emotional availability of doctors and the timeframe set for this appearance in the accelerated cancer pathways, and one might observe how the ultrafine equilibrium between affective neutrality and affective involvement has become slightly unbalanced. However, additional research needs to be conducted to explore how the impact of public health reforms inset new emotional injunctions to doctors and thus affect the delicate doses of emotionality in professional conduct.

The findings from this study have implications for how we critically approach emotions in organisations. Different forms of organising entail different ways of putting emotions to rational use, which concurrently provide different challenges to doctors. Here, the rearranged cancer pathways can be viewed as a form of organis-

ing that neglects to address the way emotions are indispensable in a skilful medical practice. Instead, the organisation of cancer pathways builds upon the (faulty) assumption that reason and emotion can be separated and compartmentalised into isolated work tasks or steps that subsequently can be distributed among different practitioners. The structural organisation of cancer treatment thus repeats a tendency in management and organisation studies to view emotion and rationality as belonging to two different spheres. These ways of organising the 'emotional' and the 'rational' are repeated in the increased presence of so-called 'emotion programs' in health care, where social and psychological aspects of medical work are organised to better supplement streamlined, effective and measurable clinical and administrative procedures in accelerated cancer pathways (see for example the call for medical 'emotion' training or the emergence of cancer rehabilitation centres, Adler and Page, 2008). The structural division of 'rational' and 'emotional' aspects of medical work attest for a need for further exploration, because this juxtaposition indeed seem to has consequences for both individuals and the organisation, as Mark (2005), among others, previously has indicated.

Chapter 8: The negotiation of the sick role - general practitioners' classification of patients with medically unexplained symptoms

Nanna Mik-Meyer and Anne Roelsgaard Obling

Department of Organisation, Copenhagen Business School, Frederiksberg, Denmark

Abstract

In encounters between general practitioners (GPs) and patients with medically unexplained symptoms (MUS), the negotiation of the sick role is a social process. In this process, GPs not only use traditional biomedical diagnostic tools but also rely on their own opinions and evaluations of a patient's particular circumstances in deciding whether that patient is legitimately sick. The doctor is thus a gatekeeper of legitimacy. This article presents results from a qualitative interview study conducted in Denmark with GPs concerning their approach to patients with MUS. We employ a symbolic interaction approach that pays special attention to the external validation of the sick role, making GPs' accounts of such patients particularly relevant. One of the article's main findings is that GPs' criteria for judging the legitimacy of claims by those patients' that present with MUS are influenced by the extent to which GPs' are able to constitute these patients as people with social problems and problematic personality traits.

Keywords: medically unexplained symptoms, general practitioner, sick role, classification, legitimacy

Introduction

In western countries, there are growing numbers of people who report pain throughout the body that cannot be medically diagnosed (Ring *et al.* 2005). The reported pain

of these individuals cannot be fully explained by pathology: no physiological tests correlate with the reported symptoms of the patients. This is a patient group that tests the credibility of doctors and the legitimacy of patients: doctors risk professional credibility for their failure to diagnose certain patients' problems while patients who have symptoms whose causes cannot be diagnosed feel the shame of being illegitimately sick – that is, they are deviant cases for whom a classification rule cannot be applied, despite the claims that they make. These are patients who, in Jutel's (2010) words, 'feel poorly, but for whom no medical explanation can be given' (2010: 230).

In the sociological literature, such deviant and illegitimate patient cases are often characterised as suffering from 'medically unexplained symptoms (MUS)' (Bendelow 2009: 57). This is an overarching category incorporating a variety of different conditions such as chronic pain, stress, milder forms of depression and contested illnesses, including fibromyalgia, chemical intolerance, irritable bowel syndrome and chronic fatigue syndrome. Although these illnesses have little in common and this patient group is by no means homogenous, these patients do fall into a particular (residual) category in the eyes of general practitioners (GPs). The patients repeatedly consult their GPs over physical pain that cannot be diagnosed medically. However, in addition to this common feature mentioned by GPs, we note that the acronym MUS is, like any other category, constitutive in nature (see Hacking 1986, Jenkins 2000). As a container category, MUS both contains individual symptoms and complaints of patients and works as a reductionist label that organises different patients into a unitary group. As an unintended consequence of this classification, the MUS label produces and stabilises the expectations of the patients in this group. Here, we focus on that act of classification – in medical terms, the diagnostic trajectory.

This article presents the results from a research project conducted in Denmark in 2008–2009 that focused on patients with MUS (Mik-Meyer 2010, 2011). In the article we show how GPs recognise this patient group, bearing in mind that GPs focus

their professional attention as much on the suffering of the patients as on the medical classification of illness. As already demonstrated by Parsons (1951, 1978) and by Balint (1964), the role of a GP is to both address patients' subjective distress and to diagnose illness through a physical examination. When GPs attempt to find out what is the matter with someone, they must attempt to attribute symptoms to causes and make a diagnosis. This diagnostic process involves linking different symptoms with each other so that eventually the combination of symptoms can be labelled as a disease or a pattern of illness and hence given an explanation.

Patients with MUS, however, make the completion of these tasks difficult. Many of the illnesses in the broad MUS category fall outside the scope of biomedical observation. When people have symptoms that fall outside clear-cut medical diagnoses, doctors tend to either classify the symptoms as psychological in nature or ignore the patients' physical symptoms altogether (see Sabo *et al.* 2000). Hence, patients with MUS challenge GPs' traditional approach to diagnosis and intervention and what it takes to be a 'legitimate' patient in a context constituted by medical credibility. Credibility refers to a GP's ability to create an explanatory framework that categorises patients in a specific sick role through the legitimisation of their complaint, even when a medical diagnosis is absent.

This article presumes that when MUS patients turn to GPs for assistance they risk being classified as illegitimate; in turn, in the patients' eyes, the GPs can be seen to be lacking in credibility as medical practitioners if they cannot state what is wrong with them. This article provides a sociological insight into contemporary understandings of 'the sick role' (Parsons 1951: 436) when the patients in question are defined by the lack of a clear-cut medical diagnosis. We also demonstrate how GPs classify symptoms described by MUS patients and create opportunities to treat this group of people as legitimately sick individuals. In this article we address these questions by firstly, examining how different explanatory models reported by GPs relate to their construction of what it is to be a legitimate patient. Secondly, we focus upon how

GPs struggle to define 'legitimate patients' relates to the (problematic) socioeconomic status, general life stories and personality traits of patients with MUS. In our present study, however, we exclusively focus on the GPs' point of view.

Classification of illness

There has been much research on the tools of classification in biomedicine (see Bowker and Starr 1999) and how diagnostic practices can be a potent way to create a social order for medicine, the patient and the doctor. In other words, diagnosis structures become a reality for individuals. Jutel (2009) writes how 'being diagnosed gives permission to be ill. What was previously a complaint is now a disease' (2009: 278). But what happens when doctors cannot give such permission because there is no clear correspondence between the patients' complaints and physiological function? Previous research problematises the dichotomy of illness and disease (see Mol 2002) and questions, for example, the utility of classifying illness narratives as either fact or fiction (Bury 1991). Recently, Michailakis and Schirmer (2010) addressed the distinction between being diagnosed and considered to be ill as a medical matter and being so diagnosed as a political matter. This focus follows Parsons' (1978) original work, which posited that society has a functional interest in minimising illnesses because the sick role effectively inhibits the fulfilment of all other social roles.

Parsons' (1951, 1978) initial thoughts on the institutional expectations of the sick role thus inspired this article. We focus on GPs' classification of patients with a similarly strong focus on how the sick role is merged with society's expectations of its citizens. We do not depart from a dualistic approach that distinguishes between (factual) disease and (fictional) illness narratives of patients. We address the ways in which GPs approach patients who hold the sick role in cases where the patients challenge the ability of the GPs to offer a clear-cut diagnosis. We try to understand how GPs primarily attend to the (problematic) social situation and (problematic) personality of patients with MUS by recognising patients with MUS as legitimate patients and

thereby entering them into the sick role. The sick role gives moral legitimacy to the claim that patients cannot perform normal responsibilities and hence that this 'condition' prevents them from performing other social roles (see Parsons 1978: 436).

A number of recent empirical studies have examined patients living without a diagnosis, including patients with MUS, and thereby brought prominence to the interdependent relation between the patient, the doctor and the reported complaint. Nettleton (2006), for example, shows how 18 neurology outpatients in England needed (but were not given) permission to be sick. Wilemana et al. (2002) explore GPs' attitudes towards the management of patients that have MUS in primary care consultation. Focusing on the relationship between doctors and patients as well as on problems of control and authority in consultations, the authors discuss the need for more training for GPs on how to manage patients with MUS. In a study on diagnosing depression in primary care, McPherson and Armstrong (2009) show how doctors struggle at first to identify certain patients but then begin to construct a category for such patients that is characterised by non-medical features – for example, by deviant features such as emotional difficulties or manipulative tendencies. Werner and Malterud (2003) use patient experiences in primary care to explore what it takes to be a legitimate patient in the eyes of a doctor when a biomedical diagnosis is unclear (in their terminology 'a credible patient'). Similarly, Gill et al. (2010) focus on how patients take an active part in the interpretation of their own symptoms.

Classification, while it occurs in the context of the medical consultation, is a complex embedded institutional process. Other studies have addressed how the medical system and the public in general deal with patients with MUS by analysing, for example, the documents of support groups and health professionals that deal with fibromyalgia syndrome (Madden and Sim 2006), patient associations and media portrayal of Morgellon (Fair 2010), Internet newsgroup postings and public debates on chronic fatigue syndrome and multiple chemical sensitivity (Dumit 2006), encounters between genetic counsellors and clients 'without a label' (diagnosis) (Brookes-

Howell 2006) and audio-recorded consultations between GPs and patients with medically unexplained physical syndrome (Ring *et al.* 2005). However, none of these studies addressed how MUS influences the GPs' classification and recognition of legitimate patients.

Theoretical framework

This article is inspired by a theoretical approach to the negotiation of the sick role that relates to GPs' ideas of what it takes to be a legitimate patient. Research into how symptoms are recognised and classified by GPs is affected not only by medical discourse but also by the different social contexts in which the negotiation of the sick role takes place. For this reason, we employ a theoretical approach that takes into account the different contexts in which identity construction (Goffman 1959, Mead 1934) and negotiation of dominating categories take place.

The explanatory models provided by doctors become highly relevant when we take a symbolic interaction approach (Goffman 1990b, Jenkins 1996, Mead 1934) to the construction of a legitimate sick role. The symbolic interaction approach pays particular attention to the external validation of identities (see, for example, the 'generalised other' in Mead 1934: 151), in our case, the GPs' explanations of the legitimacy of the sick role. Doctors' external classifications are always related to issues of identification and identity (Jenkins 1996: 113; 120–1). Classification practices are here assumed to influence not only how GPs work and think but also how patients with MUS are perceived and valued in the doctor–patient relationship. The external categorisation performed by GPs gives information about the boundaries around what can be said to constitute a narrative of complaint and grounds the definition of the medical situation.

Shilling (2003) proposes the term 'the body project' to emphasise that there is a tendency in our modern, uncertain environment to see the body as 'in a process of becoming; a *project* which should be worked at and accomplished as part of an *indi*-

vidual's identity' (2003: 4; emphasis in original). It seems reasonable to suggest that, in regard to the categorisation of patients with MUS, the body can be seen as an important workable project. The bodies of patients with MUS are examples of 'malleable entities, which can be shaped and honed by the vigilance and hard work of their owners' (2003: 5). Patients with MUS must work laboriously to make their symptoms socially visible, real and physically present (Werner and Malterud 2003). These people, however, cannot do this work alone. Patients with MUS are in strong need of others (Mead 1934), such as GPs, to define their bodies as entities that need repair and, hence, qualify them for the sick role.

A sociological focus on the process in which deviant bodies are repaired in medicine is not new (see Parsons 1951, 1978). As classic work has shown, medicine has a normalising function (see Canguilhem 1989, Foucault 1977). Medicine restores defective organs to health and corrects bodily dysfunctions. Our question, however, is the following: what does medicine actually normalise for patients with MUS? According to the scholars just mentioned, medicine not only restores and repairs parts of the body that are malfunctioning (for example, a broken leg or an infected blood vessel); medicine also seeks to restore patients' abnormalities that go beyond pathological illness and seeks to deal with norms that define 'normal' health in the surrounding society and its institutions. Doctors can in this way be seen as 'moral entrepreneurs' (Becker 1997: 147) because they legitimise and label illness.

In our empirical analysis we show how this insight becomes relevant when the GPs, influenced by common sense psychology, repeatedly transmute patients with MUS into objects of inspection and discourse set outside a traditional biomedical vocabulary (Balint 1964, Rose 1985, 1999a). This common sense perspective includes ideas about responsible citizenship and ideal social values (Rose 1999b, Dean 2002) that recently were shown to exist in a Scandinavian context also (Michailakis and Schirmer 2010). The observation that patients are the targets of governing conduct is well known in the literature (see Armstrong 1983, Rose 1999a). However, what is

significant about our findings, as we demonstrate later, is how GPs provide these patients with legitimacy through an interest in their social and personal profiles.

Our primary concern with the 'making up' (Hacking 1986) of a legitimate patient in the eyes of GPs leads to a focus on the constructive element of identities in institutions. We try to understand how institutional selves (Gubrium and Holstein 2001, Holstein and Gubrium 2000) rely heavily on the social context from which the categories in question derive.

Data and methods 13

In order to gain a deeper insight in how GPs classify and recognise patients with MUS, 21 GPs were interviewed¹⁴. All interviews has been taped and transcribed. For ethical reasons the identities of the interviewees quoted in this article have been concealed. Participants in the study were made aware in advance that they would participate anonymously. In Denmark there are strict rules for processing and securing data. For example, we removed personal security numbers and last names from the interviews before sending them to transcription and we stored the data so only the research team could gain access to it. Besides these general rules for processing and storing the data and participant acceptance (based on descriptions of the research), no formal ethical approval to conduct a research project like this, is required in Denmark. We have, however, followed the guidelines given by the British Sociological Association on how to conduct research in an ethically responsible way.

12

¹³ This research is part of a larger study consisting of a national survey of welfare officers working in the area of sickness benefits in Denmark's 98 municipalities, 41 qualitative individual interviews with people with MUS, 15 group interviews with welfare officers (N = 52) and three individual interviews and eight group interviews with GPs (N = 21) (Mik-Meyer 2010, 2011).

¹⁴ We initially planned to conduct only group interviews with GPs. However, in three cases, only one GP was present (the others were either sick or too busy to participate as promised). We have, however, decided to include both the three individual interviews and our group interviews in our analysis.

Individual and group interviews with GPs (N = 21)

We¹⁵ contacted the participating GPs for this study by randomly selecting and calling GPs across Denmark during their daily consultancy hours, between 8 AM and 9 AM¹⁶. Approximately half the GPs contacted declined to participate in the study either because they were too busy or because they found the research project irrelevant. The participating GPs, however, decided to participate after being introduced to the study first on the telephone and later after having received written information on the study. The large number of GPs who declined to participate led us to reflect on our sample of participating GPs. During our initial telephone contact and later interviews, it became clear that one dominant reason for GPs to participate was that patients with MUS were perceived as extraordinary time consuming, medically challenging and often a demanding group to work with.

We could expect the GPs in our sample to have experience with MUS for two reasons. Firstly, citizens in Denmark can choose their own GP. Secondly, in an interview study with 41 patients with MUS we found that most change their GP if their current GP did not accommodate them as suffering individuals. Another common feature among the participating GPs was that they expressed the view that having MUS prevent individuals from fulfilling work-related obligations.

These reflections aside, conducting a research project based mainly on interview material produces certain possibilities and limitations for analysis. As such, the interview material should be seen as the result of active encounters between the interviewer, with her theoretically motivated research agenda, and the interviewees, confronted with this agenda (Holstein and Gubrium 1997, Järvinen 2000). The interviews were semi-structured and maintained a focus on the interviewees' perspective and

¹⁵ This research project's methodological design, data collection and processing were conducted solely by Nanna Mik-Meyer and a research assistant, but to harmonise the writing style in this article, we have decided to write 'we' throughout our discussions in this methodological section.

¹⁶ Unlike in other countries such as the UK, it is still possible in Denmark to get in personal phone contact with GPs during the daily consultancy times.

subjective experience. We paid close attention to their involvement in the process by asking open-ended questions. In the actual interviews, for example, we often listened to longer disquisitions on the (problematic) social background of individuals with MUS that was believed to have a negative effect on their ability to get well. As a consequence, more discussion was elicited on this issue than we had initially prepared for. Thus, our research design was also adjusted during the interview process to better reflect issues of importance to the GPs.

The interviews with the GPs consisted of eight group interviews with 18 GPs and three individual interviews. Just as individual interviews can be seen as active encounters between interviewers and interviewees (Holstein and Gubrium 1997), group interviews also give special importance to the social context and the interactions among the interviewed participants in the story that is produced (Kitzinger 1994). In this case, the GPs could spur each other on to discuss various matters during the interview; this would result in discussions of central themes and sometimes parting in disagreement. The analysis has been thoughtfully conducted and the quotes presented in this article carefully selected, that is, we have not presented quotes containing extreme opinions that are only shared by a few GPs. We have also been very attentive to shorter discussions among GPs that could be seen as examples of 'spurring each other on' and, as a result, have excluded these opinions from the analysis. In most cases, the group interviews with respectively three GPs (two groups) and two GPs (six groups) were conducted in a very similar fashion to the individual interviews with a question-and-answer structure. The GPs would give answers in turn and occasionally discuss the question posed among themselves. Systematic thematic readings of the data, however, did not reveal clear differences in the positions and attitudes of the GPs who were interviewed individually and those who participated in small groups. The three individual interviews conducted with three GPs (Peter, Martin and Monica) contained the same types of reflections as the groups in relation to the GPs'

explanation of the main problems of patients with MUS and the problem of labelling the patients' complaints.

The eight group interviews each lasted approximately 1–2 hours, and the three individual interviews each lasted approximately 1 hour. All interviews, except one that was conducted at the university department, were conducted at the GPs' workplaces. We began the analysis by reading the material in its entirety and listing the themes the interviewees talked about. We then systematically grouped the responses into themes and attitudes according to their relevance to our research questions. The following questions were used in our thematic reading: (i) what characterises this group of suffering people, according to the GPs? (ii) what labels and symbols are used to describe patients with MUS? (iii) what models of explanation are used to account for symptoms and complaints?

The sick role and social problems

As discussed earlier, the classical literature has particularly focused on systems of classification and labelling and the effects of these systems on patients (Becker 1997, Foucault 1977, Goffman 1959). This approach is important because it shows how any classification system, including that of medical diagnosis, can be seen as a social construct that reflects and is produced by the given social context. However, in our analysis we attempt to understand how GPs categorise symptoms and complaints and we examine the effects of GPs' classifications.

A patient whose arm is crushed in a rolling machine in a bookbinding factory does not automatically play a legitimate sick role. Brian, one of the GPs, makes the following observation about an encounter with a former pressman:

He got his arm stuck in a roll. The arm went all the way in. He crushed all his muscles and the arm swelled up to the size of a thigh, but he healed well; he didn't undergo surgery and he regained full mobility. Now he turns up claiming that eve-

rything is wrong, but, you see, he can manage everything with his arm. He has moved apartment seven times; he has built five houses, but he claims that he cannot do anything at all.

As this GP emphasises, the patient's arm was completely healed. However, there is a discrepancy between the objective findings of the GP and the subjective complaints expressed by the patient. Similarly, another case reveals that it is not enough to have been involved in a traffic accident to be classified as legitimately sick when no physical evidence of damage is revealed in a computed tomography scan. As one of the GPs mentions, a 'minor involvement in traffic accidents leads automatically to medical contact' (Mary) or as another GP (Martin) explains:

When somebody crashes her car accidentally into something, which causes complaints of neck pain, then all others symptoms easily follow . . . manually you only find a little stiffness in the neck of the patient . . . the rest is just a retelling of the person's own complaints . . . chronic pain, headache, concentration and memory problems, tinnitus, sexual difficulties and so on and so forth.

Instead, what the suffering individual must show to be recognised as a legitimate patient can depend on the GP's emphasis on certain social background parameters. With MUS it is important to understand the categories through which GPs interpret patients' complaints and how this classification proceeds from their perceptions of their patients' (problematic) social background. As we will show, MUS cannot be separated from two important dimensions of social context: the individuals' social problems and their (problematic) personality traits.

Despite differences in their responses, many GPs share the opinion that patients with MUS are 'not born with a silver spoon in the mouth', as one GP (Paul) explains. A violent family history, weak family ties and a lack of social resources can sometimes be enough to lend support to the sick role. Paul further elaborates that a young woman in her thirties who was married to a violent husband who 'beats her nearly to

death but looks like a dream for any mother-in-law but from whom she managed to escape; of course, she now has a condition'. This condition, Paul continues, lends the patient enough legitimacy to be declared 'dysfunctional'. Paul ends his description by stating that such a patient is 'somebody who, despite her young age, will never again return to the labour market . . . she just can't manage it'. In addition to a violent husband who has changed a seemingly healthy woman into a 'totally dysfunctional' patient with unexplained pain symptoms, Brian (GP) explains that other social background factors, such as growing up under the wing of a distressed mother who was prescribed Valium, can justify medical attention.

Most often, GPs mention socially defined problems when characterising patients with MUS. These patients have few (if any) 'resources' (Brian), they are 'non-educated' (Diana), react to 'problematic life circumstances . . . and have no motivation' (Paul). They 'don't function' (Michael), have 'unacknowledged conflicts in their private life' (Peter) and they are believed to have had a 'problematic childhood' (Susanne) in which they experienced 'violence' (Paul) and carried a 'heavy load of desertion and neglect' (Martin). A convincing picture emerges from our analysis of the interviews. As one of the GPs (Diana) explains, 'these patients accommodate something social-wise'. In other words, patients with MUS are pictured as having many social problems due to their problematic upbringing, current social situation and social capacities.

To provide a more coherent description of GPs' perceptions of patients with MUS, we quote Michael (GP), who gives the following description of a typical encounter with this type of patient:

In the beginning, [patients with MUS] are very focused on the somatic problems . . . and you begin to examine their complaints. And you start some treatment for a bit of rheumatic disease and you give them some medicine to cure their pains . . . and

you talk to them . . . and then it usually becomes apparent that they represent types of people that are disadvantaged.

Martin (GP) continues in the same line of thought:

They have experienced violent disturbances in their past. . . . All kinds of things which possibly can go wrong in life, have somehow also turned out wrong for this group of people. . . . At a moment in their life this social past is converted into somatic symptoms . . . pain, dizziness, headache, myalgia . . . classic somatic symptoms.

In our study, GPs appeared to construct a category of patients characterised by deviant social factors that differentiated them from 'normal' healthy people. The transformation of a suffering individual from a patient with somatic complaints into a patient with social problems, as shown in the above quotes, demonstrates a general pattern found in the interview material. Through this transformation, a legitimate institutional identity can be constructed despite the lack of physical evidence of any illness or physical disorder.

Broadly understood, to be a legitimate patient in the eyes of GPs is to be recognised as one who suffers (Cassell 2004). There is no doubt in our material that GPs see patients with MUS as individuals who suffer, but it is less obvious *what* exactly they suffer from. Is it a malfunctioning body, a hypersensitive nerve system or poor genes? Or do they actually suffer from social problems related to a problematic childhood, an abusive partner or a bad economic situation? And, perhaps more importantly, can the latter – social problems of whatever kind – be united with the sick role? In other words, can you – medically speaking – suffer from social problems? Or do you need to suffer from more than social problems to be a legitimate patient?

The sick role and problematic personality

Our analysis so far has shown that to be a legitimately sick patient, the complaints of a patient with MUS must be explained with reference to particular social aspects of that person's life. But as the question just posed suggests, social problems might not be enough of a burden to qualify. The legitimacy of a patient – or the 'permission to be ill' (Nettleton 2006: 1167) – only becomes truly manifest, as we will show below, if social background parameters are combined with the personality of patients with MUS in the classification process.

As Monica (GP) explains, this patient group's symptoms stem from a 'combination of physical things . . . and a personal shortcoming'. Many GPs have the opinion that patients with MUS suffer from some sort of physical pain, but few discuss pain using a medical model that focuses on, for example, what kind of new diseases the pain might reflect. Of course, GPs regularly use medical terminology in their descriptions. For examples, GPs say patients with MUS are 'symptoms producers' (Brian), are 'chronically tired' (Paul), have a 'pre-morbid psyche', have 'a sensitive nervous system' (Monica) and 'have a dysfunctional disorder' (Michelle). However, GPs typically relate these vaguely medical descriptions to the patients' problematic social backgrounds and deviant personalities regardless of their physical complaints.

In our data, GPs often provide explanatory models for 'somatisation' that are grounded not only in physical distress and social problems but also in the specific personality types of patients with MUS. This awareness of the problematic personality traits of patients with MUS may be another way that GPs shift their attention away from physical complaints that cannot be observed and towards other aspects of these patients' situations. Susanne (GP), for example, explains that

it is very often patients who have a certain type of personality. They don't necessarily have a low social status or just moderate abilities, but it is presumably a question of personality types.

And Brian (GP) adds that 'somebody just happens to be bowled over and just lies there . . . it has a lot to do with personality'.

As we read through the interviews, it became clear that, as Susanne (GP) points out, 'there are certain kinds of personality types who easily turn into this kind of a patient'. Despite the fact that a social group such as patients with MUS can be seen as an 'ill-defined, fuzzy, practical and symbolic construct' (Jenkins 2010: 13), these individuals transform into a clinical workable whole by GPs' ascriptions of problematic personalities. Here is another example illustrating this phenomenon, where two GPs discuss in some detail the personality of patients with MUS:

Brian: Even on days when I'm full of positive energy . . . in other words my

tank is totally full . . . then I enter the waiting room and there – Bang –

there she is, now she sits there again . . . it is heavy.

Ann: They are infectious.

Brian: Yes, they are indeed infectious, aren't they?

Ann: I used to say that one can feel when some of them show up at the

clinic, how their energy seems to be withdrawn and tugged out from their big toe and is spilled out on the floor, don't you think? One can

feel them down there; there is no energy present, no drive at all.

Ann's acknowledgement of the patients as infectious is supplemented by another female GP, Diana, who explains, 'When they fall off the treadmill, they can't get on again'. Peter (GP) has a similar observation concerning personality in relation to coping capabilities: 'They see half-empty glasses'. Or, as Brian (GP) says, 'They see problems instead of tasks to fulfil'. Lisa, a female GP, believes that 'a lot of them have a wandering personality, which gets them into all kinds of trouble'. What might, from an outsider's point of view, seem to be very offensive images of a particular patient group is, however, also supplemented by more emotionally loaded descriptions

and verbs. For example, Diana (GP) and others point out that patients with MUS are very 'sensitive', they are 'tired' (Michelle), 'they don't display any happiness; they are rather joyless' (Brian), 'they have a low threshold of frustration . . . a low threshold for adversity, stress and demands' (Susanne), their 'lives simply hurt' (Martin) and finally, Michael concludes that they are 'inept at living . . . whiners . . . pitiable people'.

Despite the variety of the metaphors used by the GPs ranging from lay to professional terminology, there is a common pattern in their utterances: the metaphors' ability to produce associations that characterise a certain type of personality. Together, the list of (problematic) social background parameters and the different (problematic) personality types make up a common pattern that patients with MUS can be fitted into. The GPs are then able to recognise a familiar pattern in the patients' complaints, a pattern that enables the GPs to proceed in the emerging diagnostic process and reach a sort of final point in the process that forms the basis for further intervention.

GPs' professional identities as doctors depend on their ability to construct patients out of people complaining about pain. They must provide a diagnostic trajectory, which may result in an improvement of the patients' situation. That is, not only do patients with MUS need to be conceptualised as legitimate in their complaints but also the GPs are in need of legitimate patients in the encounter to be judged as credible or infallible professionals (Jutel 2010). According to our findings, GPs discover in patients with MUS some kind of fundamental human weaknesses on both a social and personal level, which can constitute a useful pattern for further interventions. When GPs focus on the social background and personality traits of patients with MUS, they are able to treat individuals with MUS as legitimate patients.

Discussion

In this article we have explored the negotiation of a legitimate sick role for patients with MUS in primary care. We have analysed what it takes from the GPs' perspective

for a sick role to be seen as legitimate. Our findings show that the GPs' evaluation of the legitimacy of individuals with MUS, who are suffering and therefore unable to work and take on daily duties, relates first and foremost to an assessment of the social background and particular personality type of patients with MUS. When a problematic background combines with a problematic personality as a series of rather distinct elements, GPs can accept patients' medical legitimacy as suffering individuals and try to accommodate their particular complaints.

Other studies on MUS in medical practice have analysed patients' perspectives (Dumit 2006, Nettleton 2006, Werner and Malterud 2003), primarily examining how patients experience their encounters with GPs and how they work to be understood and taken seriously as patients. However, little research has been done on how the sick role of patients with MUS is negotiated from the doctors' point of view or on the explanatory models used by doctors in this process. This study attempts to shed light on these problems by focusing on the external side of the identity-formation process in which the sick role is negotiated; that is, how formal classification practices among doctors produce legitimate patients.

The lack of patient voices might appear to be a weakness in the article. Other studies, for example, show how patients do not trust doctors with discussions of emotional aspects of their problems and instead choose to hide those aspects behind somatic symptoms (Peters *et al.* 2008) or how doctors fail to respond to hints of the patients' desire for emotional support (Salmon *et al.* 2008). We have, however, deliberately left patients' voices out because our theoretical perspective favours 'the others' – the external validation – in the social process of identity work (Gubrium and Holstein 2001, Mead 1934). The role of a legitimate patient, in other words, is not a role a suffering person can just take – this sick role is a position that can be given to an individual by doctors only if the suffering individual's story and situation resemble, in GPs' eyes, the story and situation of paradigmatic patients with MUS. Legitimation is thus an institutionalised matter. Patients with MUS and their specific illnesses

are in this way assembled and legitimised as much in virtue of their own experience of illness as in virtue of GPs' perceptions of what it takes to be a legitimate patient.

The GPs' explanation of legitimacy in the medical encounter turns patients with MUS into objects of recognition (Foucault 1977). As noted by Atkinson (1995: 149), a general feature of modern medicine seems to be 'the dislocation of the case from the patient's bedside' and indeed from the patient's physical presence. Our analysis of how GPs classify and recognise patients with MUS demonstrates how the patient is transmuted into an object of careful inspection and discourse that is both set outside a traditional biomedical vocabulary and is physically dislocated from the patient. The lack of a traditional medical diagnosis of patients with MUS leads GPs to create a new kind of category, a social diagnosis, which resembles a clinical diagnosis in its function as an explanatory model for further intervention but is different from a clinical diagnosis in its lack of attention to pathological components. A possible effect of social diagnosis-making may be that GPs in some cases regard the suffering of patients with MUS as simply manifestations of the social.

GPs' discussions of the sick role show that a legitimate patient identity is closely tied to social responsibilities and to the performance of certain societal obligations. We now conclude by suggesting how the current sick role might be intimately connected to social obligations, as proposed in the work of Parsons (1951: 1978). The socio-political treatment of individuals by GPs is largely infected by their role as professional experts. When framing or establishing a legitimate patient, GPs not only stabilise their own professional identity as doctors; they might also take into account general societal norms about the obligation to participate in the labour market if one doesn't have a medically explainable physical or psychological defect. It becomes part of the GP's job to help patients with MUS be responsible citizens in relation to socioeconomic and political demands of labouring, such as obligations to wake up in the morning, to hold down a job and to manage the daily tasks in one's life. In other words, medicine could be seen as a profession that is engaged in translating and rear-

ticulating contemporary norms in society concerning what it takes to be a responsible citizen in modern western societies (Dean 1998, 2002; Michailakis and Schirmer 2010). In the words of one GP, 'The trick is to help these patients to be able to take on a job', a sentiment echoed by other GPs. The GPs' evaluation of a patient's legitimacy, in other words, relates, perhaps primarily, to an evaluation of the possibility that the individual could return to the labour market. This evaluation seems to be very focused on the social background and personality types of patients with MUS. The evaluation of legitimate patients, that is, individuals who are unable to support themself at a given moment, might be intimately connected to what is perceived to be the social obligations of citizens.

How to respond to patients suffering with MUS, however, is one of the fundamental dilemmas of contemporary medical practice in primary care (Wainwright *et al.* 2006), which leaves the GPs adrift in an uncertain domain. As Griffiths *et al.* (2005) have shown, GPs prefer problem formulations, such as diagnoses, that have simple solutions as a way to create order in the midst of the chaos and confusion that their patients present. On one hand, doctors might find themselves frustrated by their inability to come up with a clear-cut diagnosis in medical encounters with patients classified as MUS. On the other hand, this particular patient group is in need of a diagnosis to validate their diffuse symptoms, which pervade most aspects of their lives.

Our findings demonstrate that GPs are prepared to set aside the traditional search for objective findings to confirm the subjective complaints of patients with MUS. This happens by constructing and negotiating a sick role even when there is a lack of a clear-cut medical diagnosis and it is difficult to label a particular illness. As Parsons (1951) argued, this process give patients access to the sick role and give medicine its power to legitimise or construct illness. But this construction of the sick role also, perhaps unintentionally, determines that patients with MUS are individuals who suffer from a combination of social problems and problematic personality traits. To be a legitimate patient, then, does not come without side effects. GPs become co-

producers of novel sick roles that might have consequences for the everyday life of these people, including their social relations with their family, employers and the welfare state in general.

Acknowledgements

This article is equally co-authored. We would like to thank Monica Greco and Stewart Clegg for helpful comments on earlier drafts. We would also like to thank the anonymous referees and the journal's editorial board for their valuable comments.

Chapter 9: Conclusion

'One must recognise empathy as a modern combat weapon on human terrain'
(Human Terrain)¹⁷

Above quote refers to practices of mapping the combat terrain of modern warfare. According to the American frontline journalist, Sebastian Junger (2010), 'human terrain' is often mistakenly conceived as the opposite of 'physical terrain'. The latter refers to mountains, deserts, vehicles, weapons, maps, and so forth. Human terrain refers to the messy social aspects of war. This includes how professional soldiers engage in social relationships with other human beings to fulfil their mission in an efficient way. Even though military officials tend to divide their concerns into conceptual slices and deal with them separately, the physical terrain and human terrain interact in so complex ways that it is hardly possible to separate them in practice. Soldiers can occupy a hilltop in human terrain much like they can occupy it in 'hard' terrain. For instance, they may hire local people to work for them and thus protect them from attacks while exposing themselves to others. This is also the function of a strategic hilltop position, where the position itself functions as a kind of protective shield. However, there is an important difference between the two kinds of terrain: no matter how many physical hilltops the soldiers have occupied, they still cannot win the combat if they mess up in relation to the human terrain. For instance, if the steady presence of male soldiers in small villages means that local women cannot leave their houses, human terrain is lost. Another way to lose human terrain is by committing too many accidentally killings. The two terrains are depicted through extremely detailed data gathering. The human terrain is covered through genealogical data and flowcharts of

_

¹⁷ Human Terrain (2009). A documentary by James Der Derain, David Udris and Michael Udris

economic, social and cultural activity, and maps of religious relationships, language conditions and regional manners. Maps of the physical terrain are, on another hand, rendered from satellite data and show population centres, vegetation and landscape elevations. A precondition for occupying new territory is then to combine the two kinds of information and on this background develop an elaborated 'master plan' to provide the means to regulate action within both terrains.

This may seem a surprising way to begin a conclusion of a thesis focusing upon the relationship between emotions and professional conduct in a cancer clinic. However, there are a number of reasons for turning to 'concerns of terrain' in this conclusion. I began this thesis by presenting an observation from the cancer clinic of the presentation and understanding of empathy in clinical practice. The observation had to do with the question of how to label, regulate and measure emotional transactions which are taking place between health care employees and patients. It also pointed to the way employees are seeking to engage in practices ranging from displaying 'genuine' care and taking action upon patients' feelings, to training of communication style to evaluating and auditing in the service of the production of a tightly knitted cancer trajectory. The metaphor of 'mapping terrains' can be used to understand how current attempts to organise cancer treatment into accelerated, streamlined, measurable production flows also involve a systematic ordering and cultivation of human conduct. If we accept this analogy we see how 'knowledge of the whole terrain' (Elias, 2000: 398) becomes a precondition for not only social order, but also for effective clinical production. The object of 'empathy' becomes a tactic weapon to engage oneself with others both in- and outside work spaces. The complex of practices that make up 'empathy' or 'empathic behaviour' involves strategising, cunning, clever tricks and wicked manoeuvres, as well as war-like elements such as surveillance and quantification of particular sites. Tied to issues of quantification and calculability, the mapping practices of the terrains within health care are expanding. Like the mapping of terrain and combat zones in modern warfare, diverse programs

within health care try to transform health care workers' job into a terrain where activities and individuals are mapped in detail, enumerated, compared, evaluated and contained. Accelerated cancer care is in this light an example of the medico-administrative programs, which produce a 'regulated subjectivity' in the heart of medical practice, working through sophisticated emotion codes, standards and constraints (Rose, 1991; Rose and Miller, 1992).

Let me next follow up on these preliminary concerns by summing up how the thesis has explored the metrics and tactics of the 'human terrain' in accelerated medicine.

Theoretical intent and contribution

I set out to do three things in the thesis: 1) to review and discuss the way emotions have generally been addressed in the social sciences; 2) to discuss the way in which emotions have specifically been studied in relation to health care organisations and conducts of doctors therein; and 3) to empirically explore the role of emotions in 'accelerated medicine', more particularly, the introduction and use of accelerated, standardised treatment packages in a cancer clinic. I will now turn to the way these three tasks have been accomplished and how they may inform each other.

First, I have commented upon the concept of emotion as a slippery entity in social science studies. I have examined emotions in peoples own experiences and understandings in a cancer clinic and beyond, and the framing of emotions in policy documents, management discourses and training programs, as well as identifying some of the sociological underpinnings of 'questions of emotion'. Despite these exercises, emotions still remain somewhat elusive. This is, however, perhaps no surprise given to the fact that the role and expression of emotions vary across contexts. The complex interweaving of personal biographies, interpersonal relationships, power relations, institutional settings and socio-historical processes give meaning to emotional display. From that insight follows that emotions are of many types and have different intensities in human conduct. 'Questions of emotion' includes asking questions about

how individuals' activity and experience are interlinked with codes of conduct that are themselves historical and socio-cultural grounded. Thus the area of emotion research is an area contesting the lines of the classic sociological concern with the relationships between 'individual' and 'society'. In a wider perspective, and interesting for me at least in light of this thesis, are the ways we seek to identify emotions as operational within social life and how we try to make sense of – and express - our emotional experiences as part of a continuous project of improving human relationships in work and organisations.

Second, I have illustrated that how one chooses to approach emotions in the field of studying and understanding health care organisations relies upon one's theoretical grounding of emotions. I have explained how a kind of 'conceptual trapping' of emotions, tied to tropes of interiority and irrationality leads to all kinds of problems. This is typically brought forward in two types of arguments. The first type promotes the idea of 'colonised emotions': that emotions, intimate relations and production and economic activity are opposed to each other and that one can occupy the other. I previously touched upon the issue of inner/outer distinctions in Chapter 4 ('Theories on emotions'), in which selected emotion studies within health care were reviewed. The weaknesses of some of those studies could be addressed to their way of putting down inner feelings of individuals as something sacred to hostile attempts of polluting intimate relations – hence the idea of 'colonising emotions'.

The second group holds the idea of 'disappearing emotions': that emotions and intimate relations have been lost in bureaucratic processes of effectivity and efficiency, and now must be brought back into human services. This idea also leads to another dichotomous image, which is the image of private and public spheres: emotions are situated in the private 'untamed' sphere while reason or rationality is set in the public sphere as something collectively manipulated and controlled. In the hands of researchers within emotion studies who tie emotions to tropes of interiority,

we see how a trapping occurs when emotions are thought to be private and mysterious and hence rarely seen as systematic or structured. This thinking places emotions in contrast with an image of the 'rational' organisation or rationalised work procedures. From this follows that emotion are foremost seen as an adjunct to work and emotional expression is understood as rarely compatible with more instrumental processes and consequences of organising social life. These ideas are way different from the perspective employed throughout this thesis, theoretically informed by Weber and Elias, which approaches emotions and the constitution of an emotional person – such as the 'compassionate' doctor - as the direct outcome (opposite a precondition) of socially organised forms of training and practices.

Third, I have empirically investigated the management of emotions in a cancer clinic and a GP's practice. Due to the shifting state of these relationships, definitions of emotion and emotional expression have been demonstrated to be subjects for various modes of governance. This is manifest in particular organisational attempts of directing and regulating the conduct of doctors in a cancer clinic and beyond. More specifically, I have followed these attempts in relation to, what I called the 'framing', 'training' and 'performance' of emotions in accelerated medical relationships. I wanted to explore how the organisation of cancer illnesses and their treatment into accelerated, standardised treatment flows not only was directed at regulating technical aspects of medical activities but also included a cultivation of emotions and emotional transactions within these activities. I coined the phrase 'conduct of doctors', which is the trained manners in which doctors behave and manage their emotions in particular situations (Du Gay and McFall, 2008). Programs of accelerated cancer treatment was showed to frame how doctors are expected to behave in certain ways and take responsibility for managing their own and patients' emotions. I found that doctors are met with two strands of interconnected emotional injunctions: a) one which requests doctors to bring in compassion or feelings into clinical activities, encouraging an 'emancipation' of emotions in medical relationships,

and b) one which urge doctors to make themselves' emotionally available' in a particular way. However, I also showed that this was not simply a call for 'setting emotions free', but entailed a call for very particular forms of emotions The injunctions could be viewed as rather standardised forms of emotional engagement in medical relationships. Let me specify this chapter by chapter.

As I argued in Chapter 5, current trends of 'bringing back' compassion or feelings in health care workers public activities may be seen as a trend where emotional restraints are cast off in favour of more liberated emotional expressions of 'empathy', 'authenticity', 'engagement' and 'responsiveness'. However, instead of a relaxation in the control of emotions, the development is more accurately understood as a tendency of what Elias called 'controlled de-controlling of emotions' (Elias and Dunning, 1986: 44). More particularly, the complexity of medical activities in accelerated cancer pathways and discourses of 'compassionate' care and more 'human' engagement result in increasing emotional control rather than loosening ways of doctors' behavior in the public health care sector. Increasing emotional control can, for instance, be considered from the perspective of the central question posed in Chapter 6 – precisely how do we see that doctors are equipped or trained to act in a certain emotional way or as a certain kind of emotional available persons? Let me try to answer this question by once again paying attention to the analysis of the training workshop in empathic communication.

In Chapter 6, I explored how new reform initiatives within health care not only enroll doctors as the focus of standardised, measurable clinical activities. They also make claims to enroll doctors as engaged and responsible selves. This involves the emergence and development of technologies by which this can be achieved. A returning issue in both the accelerated cancer pathway descriptions (reform documents) and in the doctors own debates about possible improvement of their services in the cancer clinic, was the issue of communication. In generally, the assumption was that effective communication between cancer patients and doctors not only im-

proves patients' satisfaction within the provided services, but also facilitates the accelerated clinical procedures in the pathways. By claiming that problems in accelerated interactions between patients and doctors were merely technical in nature, I demonstrated how techniques in communication were diagnosed and employed to reskill the doctors and equip them to act in certain ways in patient interactions. The training of communicating emotionally through the use of a standardised language was showed to involve the management of disturbing emotions and uncertainties in the interactions. Thus communication was considered to be an emotional skill in medical practice through which doctors can engage in collaborations with patients through 'skills in instilling coordination and recognition' (Illouz, 2008). The idea of effective communication was mobilised at the medical training facility as means of 'cultivating on the conduct of the doctor' (May et al. 2006). Employment of effective communication skills may be a subtle way of health care workers to present themselves to others as compassionately engaged while they display emotions in extremely regulated and controlled ways.

This insight reached forward to a central finding of Chapter 7. Namely, that the doctors of the cancer clinic experienced it as continuous work to establish a proper presentation of 'doctor' in emotionally challenging situations. In order to be a detached observer and sustain affective neutrality in patient-doctor interactions, doctors need to carry out important micro-regulated tasks of emotion management to maintain composure in these interactions. In situations where there is very limited time to practise this kind of work, this become difficult and presents a number of problems. To sum up, the challenge of doctors is to regulate their conduct, not only by professional ideologies, clinical guidelines and standards, but also by a deliberate and strategic expression of 'emotions'. As these emotional expressions are further cultivated it may lead to increasingly more complicated and more standardised modes of 'empathic' self-presentation in medical relationships. As I have written previously,

a possible outcome of this could be a higher risk of job burnout or morale fatigueless within the profession of medicine.

The insights from Chapter 8 are connected to central findings of Chapter 7. As we saw earlier, this chapter was occupied with GPs in primary sector health care and it dealt likewise with the management of emotions in clinical 'core' situations. More specifically, the chapter explored how doctors try to make sense of emotions in negotiating meanings of patients' stories in everyday practice. Far from being an example of bio-medical reductionism, this negotiation became an example of how doctors respond emotionally in medical encounters, when they are faced with diffuse symptoms of unexplained illnesses. The chapter demonstrated how practices of emotion management are placed at the heart of medical diagnosis making. In the process of giving patients with MUS a patient position, doctors not only used objective biomedical diagnostic tools, they also diagnosed through feelings. They heavily drew on personal opinions and evaluations of the patients' particularities when deciding if a suffering person should be given credibility from the position of a patient or not. The presentation of doctors' experiences and understanding of various 'difficult' patients demonstrated that it takes work to constituting feelings and bringing these into being in response to awareness of the social norms underpinning the medical encounter about what the doctor should be feeling in particular situations and how she should be attending her suffering patients. On this background, I suggest that the subtleties of medical relationships are looked at more detailed to register what is actually happening and whether change in the management of emotions and the manners in which professionals meet patients or clients are supported from evidence from these relationships.

We are left with an important question: what can we take away with us from this investigation of questions of emotion? I have described some key answers to these questions and addressed four concerns at work:

- It seems that emotions and the management of emotions play a significant role
 in organisational structures that systematically both accelerate and reduce human interactions in public meetings, while intending to make these interactions
 more 'human'.
- The more 'hard' dimensions of medical services are standardised, the more we also see a standardisation of 'soft' dimensions of these services.
- These 'soft' dimensions such as the provision of empathy, responsiveness and engagement in interactions between doctors and patients are increasingly established through subtle 'metrics', in other words standards which are profoundly involved in the constitution of the entities they aim to measure.
- In clinical practice, the navigation between these 'hard' and 'soft' dimensions
 entails various challenges to doctors. Not least because the time frame of average consultations is very condensed and the amount of physical encounters between doctors and patients is reduced due to aims of acceleration and efficiency.

The described key points indicate, I suggest, a need for detailed studies of the kind of micro-organisational issues and challenges that are connected to contemporary forms of accelerated medical care. This would direct our attention to important questions of how 'hard' and 'soft' dimensions of care are organised in practice, and how health care workers on a daily basis, together with patients, try to coordinate and make sense of those dimensions. The crucial point here is that the task of future research, as explained through the thesis chapters, is not only to explore how emotions are getting excluded from production units, such as hospitals, because they hinder effective transactions. It is also necessary to explore how emotions are organised as part of formal organisation structures and how this affects the workers within these structures.

Practical concerns

The findings from this PhD project have implications for how we critically think about emotions in organisations and processes of organising. Different forms of organising entail different ways of putting emotions to rational use, which concurrently provide different challenges to the conduct of doctors. The development of accelerated cancer pathways can be viewed as a mode of organising that neglects to address the way the management of emotions are indispensable in skilful medical activities. Instead, the organisation of cancer illness and their treatment into standardised production flows builds upon the assumption that physical terrain and human terrains can be separated and compartmentalised into isolated activities or steps that subsequently can be distributed among different practitioners in different work settings. The very organisation of cancer illnesses and their treatment assumes that 'soft' relationships and emotional aspects can and should be separated from technical aspects of the patient pathways and that it leads to better patient care if each one individually is standardised and streamlined into separate entities. It remains unclear whether this increased fragmentation and 'work of purification' (Latour, 1993: 31) can be reconciled by anyone other than the patients and what costs this work might entail. The increased separation of rational transactions and psychological and social support elements provides new opportunities but also greater challenges to the construction and experience of continuity in the pathways. This form of organising medical services attests for a need for further exploration, because the divisions of terrains within specific services may both influence patients experiences of continuity and collaboration when they undergo treatment, and it may affect the motivation and job satisfaction of the health care workers involved.

One central feature of the rearranged cancer pathways is the reduction of the time available for concrete, physical interactions between medical professionals and patients. Every sequence of the pathway has been optimised and unnecessary slack has been cut off, including a reduction of face-to-face interactions and an acceleration of technical tasks. Accordingly, and as effect of the speed, patients are left with a lot less time to cope with their feelings, when they are faced with a diagnosis of cancer and the mutilating bodily interventions, which often accompany this kind of diagnosis. Also the time available for doctors to make themselves emotional available and engage with patients in the technical and psychological intimacies involved in medical relationships, and to manage the often strong emotional responses of patients, is strongly reduced.

Another central feature of the rearranged cancer pathways are that these 'intimacies' are increasingly focused upon as matters of importance and improvement. A doctor (Sarah) in the cancer clinic makes this reference to the changing organisational context and its possible effects upon her daily activities in an interview conducted for the thesis. She explains:

'The better we get at taking care of the patients and discharging them effectively from the clinic, the more we preclude them from our "softer" services, which we know that they find important – as demonstrated in various patient studies and surveys. Also social interventions, which we ourselves believe in are necessary for a good healing process, are now rendered mission impossible because of the developments of the treatment regime'.

The so-called 'softer' services, the doctor emphasises in the quote, involve the psychological and social dimensions of medical care. However, the point here is that the speed-ups of the cancer pathways influence the forms and expression of emotional engagement between medical professionals and patients.

In Denmark, at least, there has been little attempt to engage systematically with the questions of whether professionals' behaviour and everyday manners have been shaped by organisational changes, either specifically in relation to specific po-

litical or institutional changes or in the wider sense of being governed by modern forms of rules, such as elaborated technologies for the regulation of professional conduct. On the basis of this thesis I propose a move towards engaging directly with how the conduct of individuals are getting reframed in relation to the organising and managing impulses of contemporary society and what kind of problems this entails.

Future research

This thesis has of course its obvious limitations. However, instead of talking about what I should have done differently throughout my research, I will point to some further research suggestions, which can substantiate the analysis of the existing work.

This research was based on a general type of contemporary speculation. This means that I have not covered here historical work that links changes of the 'psychological make-up' of doctors with changes of the socio-economic health apparatus over time. There are well-documented studies that show how societal changes bring about changes of the identity of doctor, Armstrong (2002) for instance shows how the identity of doctor since the 1960's has changes from that of an observing doctor, subsuming the doctor's identity to that of her professional group, to that of a subjective doctor, who can be externally inspected in relation to her 'personality'. The psychological presence of the doctor involved that it became natural both within the medical profession and the political-administrative system to debate the doctor's personality traits, such as her psycho-social capabilities and her way of emotionally connecting to patients in the medical encounter. We see how it is in these historical shifts to this 'personal' doctor-patient relationship, that real problems emerge. Suddenly previous boundaries defined by professional codes of conduct and 'doctorcentred' concerns became diffuse and complex and the possibilities for involvement and detachment correspondingly multiple and ill defined. This existing research on the management of the emotions in accelerated medical relationships could positively benefit from historical accounts of 'what makes doctor' in order to capture the sociohistorical changeability of emotions and emotional display. A part of that account would also include paying attention to changes in the control or management of emotions and in the relative importance of specific emotions and emotional display in different periods.

Another important issue to be dealt with here is the question about 'whose emotion management'. I have throughout the thesis only focused on doctors' activities. However, doctors cannot handle cancer illnesses, their treatment and psycho-social dimensions of care all by themselves. They need others for it, such as nurses, patients, secretaries, administrators, relatives, IT devices and so on. Apropos the nursing profession, a general assumption is that doctors merely deals with technical aspects of patient trajectories and hence tend to be forgiven their poor emotion management skills as, in the gendered division of health care labour, nurses are tasked with picking up the emotional mess the doctors leave behind (see for example Theodosius, 2008). We might also recollect the presence of this assumption in the nurse's view upon doctors' work in the introductory observation from a staff meeting in the cancer clinic. Indeed, there is some truth to the argument. In the cancer clinic we see for instance how nurses follow up on doctor-patient conversation in order to make sure that the patients have understood the central content of this conversation. The individual nurses in the cancer clinic also have their own agenda in relation to how they present themselves in encounters with patients. An issue to be dealt with for future research is thus the emerging issue of 'conflict' and 'coordination' in cases where psycho-social dimensions of cancer illnesses and their treatment are organised across work units and professions.

This brings me to the final issue in this conclusion, namely the presence of emotions in our time. 'Presence' is perhaps the wrong term to suggest here, because I am not going to propose a diagnostic of the present. Empirical attention in this thesis has

been focused on standardised forms of organising emotional engagement in a cancer clinic. However, these reflections can also be transferred to sites outside the clinic.

The specific ways of organising the 'emotional' and the 'rational' are reproduced in the increased presence of so-called 'emotion programs', where social and psychological aspects of medical care are organised in separated entities outside the hospital organisation to better supplement the accelerated procedures inside the hospital corridors. A concrete example in relation to the treatment of women with breast cancer is the development of so-called 'emotion agencies', which are organisational entities that are created for the specific purpose to supplement the delivery of effective, standardised transactions in health care organisations by providing emotional support to cancer patients. The emergence of rehabilitation centres, gym classes, mindfulness meditation programs, illness coaching, therapeutic self-help support, relaxing kayak courses, and etcetera in both private and public domains is an example of these kinds of agencies.

Several important socio-historical trends concerning the management and expression of emotions are related to the contemporary valuation of emotions. One trend is the move towards the emancipation of emotions and 'informalisation of manners' (Wouters, 2011: 140). Emotions which have previously been denied or repressed now regain wider acceptance in social life. Another trend is the tendency of revealing and articulating emotions and one's emotional state to others. Accordingly we see the emergence of social arrangements which can meet this interest. What this suggests is that processes of standardisation and rationalisation are followed by an interrelated intensely specialised emotion culture, which plays a crucial role in the ways individuals act and understand themselves. If the study of emotions teaches us anything it might be that it offers us a lesson about how emotions work to constitute both subjects and social formations.

English summary

The introduction of 'cancer packages' in the Danish healthcare system has led to a reorganisation of cancer diseases and their treatment into accelerated, streamlined patient pathways. This development involves not only that technical work activity in the pathways, for example to take a biopsy, to book a patient for surgery or to prescribe preoperative medicine, increasingly becomes standardised. It also involves that political and managerial light is cast on the psychological and social activity involved in cancer care and that these aspects are made objectives to processes of standardisation and cultivation.

The thesis has as its point of departure the assumption that new forms of organising influence how intimate interactions between doctors and patients are structured and directed in the health care system. The assumption gives rise to examine more specifically the relationship between emotions in a cancer clinic and types of professional conduct therein. An ongoing interest in the thesis chapters is the interest in so-called 'questions of emotion'. The attention to these questions – or more specifically to behaviour, feelings, gestures, impressions, sentiments, manners, affects, expressions etc. and the codes, norms and values directing and regulating them – can help us understand how individuals emotionally express and experience themselves in specific historical moments, organisational contexts and social situations.

More precisely, I ask how emotion is 'framed', 'trained' and 'performed' by doctors in accelerated medical relationships. The connection between these questions is investigated through detailed empirical analyses of how notions of 'compassion' and 'authenticity' emerge in the heart of accelerated, standardised medical services, and moreover, how doctors get to know about and manage people's feelings and make themselves emotionally available to others in social interactions.

Traditionally, both in the sociology of emotions and in particular in studies of emotions in health care practice, there have been a tendency to relate the man-

agement of emotions to the work undertaken by the nursing profession. However, practices and techniques of emotion management cannot be monopolised to a single profession. Doctors too are involved in this kind of labour in their daily routine interactions with patients. Reconsider for instance the actual work doctors must take on to be perceived as empathic and responsive beings in their engagement with patients. Or the concrete medical situations doctors find themselves in when they have to disclose serious diagnoses of cancer to anxious and alarmed patients; inform about bodily mutilating treatment plans, or provide support to gravely ill younger women with metastatic cancer disease.

This thesis is based on a qualitative study of doctors' management of emotions in a cancer clinic in Denmark in which accelerated cancer treatment regimes have been a part of the formal structure since 2008. Through a focus on doctors' interactions with patients in these regimes, the thesis explores the role and function of emotions, and furthermore how emotions are met with intense political and institutional attention, evident in the development of various quality and improvement strategies. This implies that emotions, for instance empathy, compassion and responsiveness, are framed, trained and performed in increasingly standardised ways that subsequently can be measured and evaluated.

The questions of emotion are guiding the thesis and framing the individual chapters. The thesis is divided into two parts – a frame (Chapter 1-4 + 9) and four articles (Chapter 5-8). The frame begins with a description of the organisation of cancer illnesses and their treatment in the Danish health care system. Accordingly, Chapter 2 introduces the concept of 'accelerated medicine' and it explains the development of the so-called 'cancer treatment packages'. It further introduces the issue of emotions in accelerated cancer care and it pays attention to how questions of emotion arise from the empirical context.

Moving on from here, Chapter 3 deals with theories on emotions in organisations. In this chapter, both classic sociological literature and more recent theories on emotions, social order and organisations are addressed. In addition to this, the chapter contains a review of texts that focus especially on emotions in health care practice and discusses how some of these texts fall short in their approach to emotions because they tend to elevate emotions to the disregard of rationality and rationalistic activities. The chapter then moves towards a preliminary draft of my own theoretical approach, which is further developed in the remaining chapters of the thesis.

Chapter 4 explains a number of methodological challenges which accompany my qualitative study of emotions in a cancer clinic and its surroundings. It describes the situatedness of my PhD project and my access to the clinic and its occupants, and it explains how my object of inquiry developed over time – from foci to a developing focus on emotions. It also contains a relatively clean description of the conducted fieldwork; of the qualitative methods and the extent of materials used for this research, and it describes how each analysis in the thesis takes smaller parts of the empirical material as its point of departure. Finally, it discusses problems of 'involvement' and 'detachment' (Elias, 1987) in the study of emotions, and it pays attention to a kind of 'methodological vocation' which is described as the vocation of researchers to approach emotions without sentimentality.

Chapter 5 explores how exactly emotions are framed in medical relationships and it addresses how this framing equips doctors to act in certain ways in particular situations. Using the development of accelerated cancer pathways and a wide range of health care reform documents as an empirical case, the chapter explores how doctors are encouraged to become more emotionally available to patients. As the chapter demonstrates, this call for emotionality is accompanied by increasing demands on doctors to manage the emotions of others in recordable, measurable and standardised ways. To be a compassionate doctor involves an exhibition of a particular emotional behaviour in medical relationships, what Elias calls a 'controlled de-

controlling of emotions' (Elias and Dunning, 1986: 44), namely that of a 'sincere' or 'authentic' behaviour which doctors are required to enact and to be committed towards. What this points to, I argue, is a set-up where health care reform documents encourage room for personal and unique concerns in medical relationships while expecting maximum acceleration and standardisation of these relationships.

Chapter 6 continues the investigation of questions of emotions through a discussion of how empathy and responsiveness, as specific techniques of emotion management, arise as the outcome of well-structured forms of training and practice in the cancer clinic. The chapter focuses on a training workshop in 'empathic communication' through which doctors from the clinic learn to recognise and control the emotional frame of doctor-patient interactions. It addresses how the training and practice of communicating empathically in these interactions rely on standardised scripts, which in turn direct and cultivate the conduct of doctors. In line with the argument in the previous chapter, it concludes that humanistic values increasingly become the target for techniques of micro-management such as qualitative measurement and performance audit. In other words, the chapter argues that attempts to improve 'soft' dimensions of medical services entail a further standardisation of these aspects.

Chapter 7 addresses the question of how doctors relate emotions to their understanding of professionalism and principles of standardisation and speed in the treatment of cancer illnesses. In a present work environment of reforming and rationalisation drives, the chapter directly asks 'how doctors feel'. This question is explored through doctors' personal biographies of emotional experiences and challenges in relation to their routine and everyday situations. The chapter explains how emotions and emotional display are frequently performed in a rationalised way to help doctors ensure the progress and efficiency of the accelerated treatment regimes. However, the chapter scrutinises how the rearrangements of cancer illnesses and their treatment into 'accelerated packages', and attempts to separate social and psychological dimensions of medical services from technical dimensions of these services, com-

plicate doctors' ability to incorporate emotions into their practices in a 'proper' way. What this points to, is the need for researchers to investigate how different forms of organising technical tasks within health care entail different ways of organising emotions and emotional practices within this field, and how this generates organisational problems of various kinds for both patients and doctors.

Chapter 8 completes the shift from the 'framing' and 'training' of emotions to concrete 'performances' of emotions in medical relationships. It addresses questions of emotion through a study of how general practitioners (GPs) approach patients with medically unexplained symptoms (MUS) in primary care. Concerns about the role and function of emotions in the core of the medical enterprise come under closer scrutiny here because the chapter explores how GPs are diagnosing with feelings. Accordingly, the chapter explores how GPs diagnose MUS patients as legitimately sick patients. What makes this classification procedure difficult is that MUS patients fall inside a particular residual category in the eyes of the doctors. This defies the doctors' capacity to decipher the patients' intentions and motives, making it hard to empathise with them. To confirm the subjective complaints of patients, the GPs must rely on their personal opinions and evaluations of a patient's particular circumstances when deciding whether the patient is legitimately sick or not. These different strategies for managing the complaints of patients are connected to ways of dividing individuals into certain categories of persons characterised by deviant features such as unpleasant personalities and manipulative appearances. The chapter emphasises how the affective connection between the patient and the doctor becomes paramount to the outcome – the diagnosis – of the clinical encounter.

In the concluding chapter, the main arguments of the thesis is summed up and their implications for understanding the relationship between emotions in a cancer clinic and forms of professional conduct therein and, more broadly, the active, shaping influences between individuals and social, medico-scientific and political worlds, are indicated. It is demonstrated how a central feature of the rearranged can-

cer pathways is the reduction of the time available for concrete, physical interactions between medical professionals and patients. Every sequence of the pathway has been optimised and unnecessary slack has been cut off, including a reduction of face-toface interactions and an acceleration of technical medical tasks. Accordingly, and as an effect of the speed, patients are left with a lot less time to cope with their feelings, when they are faced with a diagnosis of cancer and the mutilating bodily interventions which often accompany this kind of diagnosis. Also the time available for doctors to make themselves emotionally available and engage with patients in the technical and psychological intimacies involved in medical relationships, and to manage the often strong emotional responses of patients, is strongly reduced. The point here is that the speed-ups of the cancer pathways influence the forms and expression of emotional engagement between medical professionals and patients. The conclusion points to the fact that a contemporary focus on 'effective' and 'standardised' supply of medical services is accompanied by an increasing focus on 'soft' and 'intimate' relationships. The latter includes an interest in how these relationships should be structured and directed. Unlike previous ways of organising cancer care and health care workers' management of emotions, the increased separation of 'production' and 'human' creates new opportunities but also increased challenges in the Danish health care system – not least in relation to the individual doctor and the ways in which he or she is called upon to act as a certain kind of 'doctor'.

Resume af afhandling

Introduktionen af 'kræftpakker' i det danske sundhedsvæsen har ført til en reorganisering af kræftsygdomme og deres behandling. Disse udføres nu i stigende grad i accelererede, strømlinede patientforløb. Denne effektiviseringsproces indbefatter ikke kun, at tekniske arbejdsopgaver i patientforløbene, fx det at tage en biopsi, at booke en patient til operation eller at ordinere præoperativ medicin, ensartes. Det indbefatter også, at psykologiske og sociale aktiviteter ensartes og standardiseres. I afhandlingen arbejdes med en antagelse om, at nye former for rationel og standardiseret organisering har betydning for, hvordan "bløde" mellemmenneskelige relationer defineres og varetages i sundhedsvæsnet. Antagelsen giver anledning til at undersøge mere specifikt, hvordan forholdet er mellem følelser i en kræftafdeling og forskellige former for sundhedsprofessionelle handlemåder. En gennemgående interesse i afhandlingens kapitler er således en interesse i såkaldte 'følelsesspørgsmål'. Mere præcist spørger jeg hvordan følelse dikteres, trænes og udføres af læger i accelererede kræftbehandlingsforløb. Disse spørgsmål løber som en rød tråd igennem afhandlingens kapitler.

Traditionelt set er der både i følelsessociologien samt i specifikke studier af følelser i sundhedsorganisationer en tendens til at koble følelsesarbejde sammen med det omsorgsarbejde, der udføres af sygeplejersker. Følelsesarbejde kan dog ikke monopoliseres til kun at omfatte den sygeplejefaglige profession. Også læger udfører i høj grad følelsesarbejde i deres rutinemæssige, daglige interaktioner med patienter. Tænk blot på det konkrete arbejde læger skal udføre for at fremstå empatiske, lydhøre og engagerede, når de modtager patienter i konsultationen. Eller de helt konkrete, medicinske situationer læger befinder sig i, når de skal overbringe en alvorlig kræftdiagnose til en oprevet patient, informere om en, oftest kropslig mutilerende, behandlingsplan eller yde støtte til en alvorligt syg, angstpræget yngre kvinde med metastaserende kræft.

Afhandlingen bygger på en kvalitativ undersøgelse af lægers følelsesarbejde på en brystkræftafdeling i Danmark, hvor accelererede kræftforløb blev indført i 2008. Ved at følge lægers interaktioner med patienter i disse forløb, undersøges den rolle og funktion følelser har og ydermere, hvordan følelser i forhold til tidligere bliver genstand for øget opmærksom gennem forskellige kvalitets- og forbedringsstrategier. Dette indebærer, at følelser – som eksempelvis empati, lydhørhed og engagement – iscenesættes, trænes og udføres på stadig mere standardiserede måder, der kan måles og evalueres på linje med andre kliniske og administrative aktiviteter.

Afhandlingen er struktureret i to dele: En ramme (Kapitel 2-4 samt Kapitel 9) og fire selvstændige artikler (Kapitel 5-8). Den første del indledes i Kapitel 2 med en beskrivelse af organiseringen af kræftsygdomme og deres behandling i det danske sundhedsvæsen. Kapitlet introducerer til konceptet 'accelereret medicin' og det forklarer udviklingen af de såkaldte 'kræftpakker'. Endvidere stiller kapitlet skarpt på, hvordan særlige 'følelsesspørgsmål' udgår fra den empiriske kontekst.

Kapitel 3 kobler sig direkte på disse spørgsmål og ved hjælp af en teoretisk diskussion af diverse følelsessociologiske teorier, giver kapitlet forslag til, hvordan man kan forstå og undersøge følelsesspørgsmål i moderne organisationer. Kapitlet adresserer både klassisk, sociologisk følelsesteori samt nyere teori om følelser, social orden og organisationer. I forlængelse af dette gennemgås studier, der særligt har beskæftiget sig med følelser i sundhedspraksisser. Kapitlet diskuterer, hvordan flere af disse studier kommer i problemer, fordi de antager at rationalitet, eller rationelle organisatoriske aktiviteter, analytisk kan adskilles fra følelser og mere 'autentiske', individuelle aktiviteter. Afslutningsvist gives et bud på en teoretisk forståelsesramme, som videreudvikles i løbet af afhandlingens kapitler.

Kapitel 4 forklarer, hvilke metodologiske udfordringer der har ledsaget afhandlingens studie af følelser i en kræftafdeling. Kapitlet beskriver vejen til ph.d.-projektet, adgangen til kræftafdelingen og det forklarer, hvordan forskningsobjektet tog form over tid. Det indeholder også et udførligt overblik over det udførte feltarbej-

de; det beskriver de metoder, som er taget i brug for at indsamle det empiriske materiale og det redegør for, hvordan hver enkelt artikel gør brug af udvalgte dele af materialet. Endelig diskuteres problemet om henholdsvis 'detachement' og 'involvement' (Elias, 1987) i studiet af følelser, og det bringer et metodologisk kald på banen, nemlig et kald til følelsesforskere om at studere følelser uden sentimentalitet.

Afhandlingens første analyse præsenteres i Kapitel 5. Kapitlet tager afsæt i udviklingen af accelererede kræftforløb og et bredt udsnit af sundhedspolitiske dokumenter som empirisk case og undersøger, hvordan læger opfordres til at stille sig følelsesmæssigt til rådighed for patienter. At være en medfølende læge indebærer således en fremførelse af en særlig form for 'ægte' eller 'autentisk' følelsesadfærd i patient-læge interaktioner, eller hvad Elias kalder 'kontrolleret de-kontrollering af følelser' (Elias and Dunning, 1986: 44, min oversættelse). Dette peger i retning af et foretagende, hvor sundhedspolitiske dokumenter på den ene side opfordrer til, at der i interaktioner mellem læger og patienter skabes plads til den enkelte patient, der har unikke behov, som lægen må tilgodese på den bedst mulige vis. På den anden side forventes, at disse interaktioner skal udføres med maksimal hastighed og at alle enkeltaktiviteter i processen skal reguleres og standardiseres.

Kapitel 6 fortsætter undersøgelsen af følelsesspørgsmål gennem en diskussion af, hvordan empati og lydhørhed – forstået som specifikke følelsesstyringsteknikker, opstår som en effekt af velstrukturerede former for træning og praksis i kræftafdelingen. Kapitlet fokuserer på en træningsworkshop i 'empatisk kommunikation', hvor læger i afdelingen gennemgår træning, der skal sætte dem i stand til at genkende og kontrollere den følelsesmæssige ramme for læge-patient interaktioner. Det adresserer, hvordan træning i at kommunikere empatisk bygger på det at være i stand til at efterleve et sæt standardiserede manuskripter. Disse virker som et indirekte middel til at kultivere lægers handlemåder. I forlængelse af argumentet fra Kapitel 5, konkluderer dette kapitel, at humanistiske værdier i medicinsk praksis i stigende grad bliver genstand for mikro-styrings-teknikker som eksempelvis kvalitetsmåling

og præstationsaudit. Med andre ord viser kapitlet, at forsøg på at forbedre 'bløde' dimensioner af medicinske ydelser involverer en yderligere standardisering af disse.

Kapitel 7 adresserer, hvordan læger relaterer følelser til deres forståelse af professionalisme og principper for standardisering og hastighed i deres daglige arbejde. Kapitlet spørger direkte 'læge, hvordan føler du' i en kontekst præget af kontinuerlige reform- og rationaliseringstiltag. Dette spørgsmål undersøges via lægers personlige beretninger om følelsesmæssige oplevelser og udfordringer i forbindelse med deres arbejdsrutiner i afdelingen. Kapitlet viser, hvordan følelser typisk mobiliseres af læger på en rational måde for at skabe fremdrift og effektivitet i de accelererede behandlingsforløb. Imidlertid gransker kapitlet, hvordan reorganiseringen af kræftsygdomme og deres behandling i 'accelererede pakker' og forsøg på at adskille sociale og psykologiske dimensioner fra tekniske dimensioner af medicinske ydelser, i stigende grad gør det vanskeligt for læger at indarbejde følelser i deres aktiviteter på en hensigtsmæssig måde. Dette peger i retning af, at der er brug for yderligere at undersøge, hvorledes forskellige måder at organisere tekniske opgaver inden for sundhedsvæsnet også nødvendigvis medfører måder, hvorpå man samtidig organiserer følelser og muligheder for at udføre følelsesarbejde. Dette har konsekvenser for modtageren af ydelserne (patienten) samt for afsenderen af ydelserne (lægen), som får sværere ved at udføre følelsesarbejde som en naturlig integreret del af det øvrige kliniske og administrative arbeide.

Kapitel 8 tager læseren med ud af kræftafdelingen og adresserer, hvordan praktiserende læger håndterer følelser i lægekonsultationen, når de møder patienter med 'diffuse lidelser'. Problemet med patienter i denne patientkategori er, at de falder uden for almindelige, klassificerbare sygdomskategorier. Kapitlet undersøger, hvordan læger forsøger at skabe mening i patienternes diffuse sygdomsbillede og give dem en diagnose. Dette arbejde besværliggøres af, at disse patienter oftest vækker en masse modstridende følelser i lægen, hvilket samtidig gør det svært at imødekomme patienternes forventninger og empatisere med deres motiver. Kapitlet demonstrerer,

hvordan lægerne udstyrer disse patienter med en sygdomsdiagnose ved at klassificere dem som mennesker, der lider af en problematisk personlighed og under en svag social baggrund. Endvidere understreges hvordan den følelsesmæssige relation mellem læge og patient bliver afgørende for resultatet af mødet mellem dem.

Afslutningsvist i Kapitel 9 oprulles afhandlingens hovedkonklusioner og der peges frem mod nye forskningsfelter. Det fremføres, at samtidens fokus på 'effektive' og 'standardiserede' leveringer af sundhedsydelser ledsages af et stigende fokus på 'bløde' relationer mellem mennesker og hvordan disse skal organiseres. Hvor tekniske og psykosociale aspekter af medicinsk praksis tidligere blev håndteret 'organisk', dvs. som dele der organisatorisk hang sammen i patientens indlæggelsesforløb, fører den stadig skarpere rendyrkelse af effektive transaktioner (fx at informere effektivt om behandlingsplanen; at foretage et kirurgisk indgreb skånsomt; at udskrive patienten hurtigt) til en øget opmærksomhed på og krav om at kvalitetssikre og standardisere intime relationer (fx at være anerkendende lyttende i patientmøder; at fremstå empatisk og medfølende). Til forskel fra tidligere måder at organisere kræftbehandling og sundhedsprofessionelles følelsesarbejde, giver den øgede adskillelse af 'produktion' og menneske' nye muligheder, men også øgede udfordringer i det danske sundhedsvæsen – ikke mindst i forhold til den individuelle læge og de måder, hvorpå han eller hun nu skal være læge på.

Appendix A: Interviews in the cancer clinic and its surroundings

Name	Interviews	Position	Semi-structured	Explorative inter-
			interviews, length:	views
			30 minutes	Length: 30 min. –
				1.30min.
Mary	1	Medical staff	X	
Lucy	1	Medical staff	X	
Paul	1	Medical staff	X	
Erving	1	Medical staff	X	
Deborah	1	Medical staff	X	
Susan	1	Medical staff	X	
Michael	1	Medical staff	X	
Eva	1	Medical staff	X	
David	3	Head of research	X	XX
Sarah	1	Medical staff	X	
John	1	Medical staff	X	
Monica	1	Medical staff	X	
Brian	1	Medical staff	X	
Jan	1	Retired professor		X
Stewart	1	Professor		X
Susan	2	Head of nursing		XX
Ann	1	Clinical teacher		X
Sylvia	1	Pathway coordi-		X
		nator		
Rachel	1	Researcher		X
Beth	1	Consultant		X

Appendix B: Observations in the cancer clinic and its surroundings

Observation type	Location	Frequency
Morning conferences	Staff meeting room, the cancer	8 conferences
(mono-disciplinary)	clinic (CC), The University hospi-	
	tal (UH)	
Morning conferences (in-	Staff meeting room, CC, UH	3 conferences
ter-disciplinary)		
Patient-doctor consulta-	Outpatient unit, CC, UH	15 days
tions		
Breast cancer surgery	Operation theatre, surgical unit,	4 operations
	UH	
Ward rounds	Inpatient ward, CC, UH	4 rounds
Radiology conferences	Department of Radiology, UH	5 conferences
Multidisciplinary confer-	Department of Pathology, UH	2 conferences
ences		
Management meetings	Administrative unit, UH	1 meeting per
		week in three
		years
Training workshop	Education unit, UH	1 day

References

Abbott, A. (1988) *The system of professions: An essay on the division of expert labour.* Chicago: The University of Chicago Press.

Adler, N. E. and Ann, E. K. (2008) *Cancer care for the whole patient: Meeting psychosocial health needs*. Washington, D.C.: The National Academies Press.

Albrow, M. (1994) Accounting for Organizational Feeling, in Ray, L. J. and Reed, M. (eds) *Organizing Modernity: New Weberian Perspectives on Work, Organization and Society*. London: Routledge, pp. 98-122.

Ankowitsch, C. (2002) Generation Emotion. Berlin: BTV.

Athanasion, A., Hantzaroula, P. and Yannakopoulos, K. (2008). Towards a new epistemology: The affective turn, *Historein*, 8: 5-16.

Armstrong, D. (1983) *The political anatomy of the body*. New York: Cambridge University Press.

Armstrong, D. (2002) A new history of identity. New York: Palgrave

Ashforth, B. E. and Humphrey, R. H. (1995) Emotions in the workplace, *Human Relations*, 48(2): 97-125.

Atkinson, A. (1995) Medical talk and medical work. London: Sage.

Balint, M. (1964) *The doctor, his patient and the illness*. London: Pitman Books Unlimited.

Baszanger, I. and Dodier, N. (2004) Ethnography: relating the part to the whole, in Silverman, D. (ed.) *Qualitative Research - theory, method and practice*, Second edition. London: Sage, pp. 9-34.

Baum, M. (2007) *The philosophical surgeon: in defense of evidence based medicine*, Karl Popper Memorial Lecture, London School of Economics, November 2007.

Becker, H. S. (1997) *Outsiders: Studies in the sociology of deviance*. New York: The Free Press.

Bell, D. (1973) The coming of post-industrial society. New York: Basic Books.

Bendelow, G. (2009) Health, emotion and the body. Cambridge: Polity Press.

Bennett, M. J. (2001) *The empathic healer: An Endangered Species?* London: Academic Press.

Berger, P. L. and Luckmann, T. (1967) *The social contruction of reality: A treatise in the sociology of knowledge*. New York: Doubleday

Blichert-Toft, M., Christiansen, P. and Mouridsen, H. T. (2008) Danish Breast Cancer Cooperative Group – DBCG: History, organization, and status of scientific achievements at 30-year anniversary, *Acta Ontologica*, 47(4): 497-506.

Boltanski, L. and Chiapello, E. (2005) The New Spirit of Capitalism. London: Verso.

Bolton, S. C. (2001) Changing faces: nurses as emotional jugglers, *Sociology of Health and Illness*. 23(1): 85-100.

Bolton, S.C. (2005) *Emotion management in the workplace*. Basingstoke, Palgrave Macmillan.

Bolton, S. C. and Boyd, C. (2003) Trolley dolly or skilled emotion manager? Moving on from Hochschild's managed heart, *Work Employment and Society*, 17(2): 289-308.

Bone, D. (2002) Dilemmas of emotion work in nursing under market-driven health care, *The International Journal of Public Sector Management*, 15(2): 140-150.

Bowker, G.C. and Starr, S.L. (1999) Sorting things out: Classification and its consequences. Cambridge, Massachusetts: The MIT Press.

Brody, H. (1998) Foreword, in Greenhalgh and Hurwitz (eds) *Narrative based medicine*. London: BMJ Books, pp. xxiii-xv.

Brookes-Howell, L.C. (2006) Living without labels: The interactional management of diagnostic uncertainty in the genetic counselling clinic, *Social Science and Medicine*, 63(12): 3080-91.

Bub, B. (2004) The patient's lament: hidden key to effective communication: how to recognise and transform, *Medical Humanities*, 30(2): 63-63.

Buchanan, D.A. and Fitzgerald, L. (2010) 'New lock, new stock, new barrel, same gun: The accessorized Bureaucracy of health care', in Clegg, S., Harris, M. and

Höpfl, H. (eds) *Managing Modernity – Beyond Bureaucracy*. Oxford: Oxford University Press, pp. 56-80.

Bury, M. (1991) Illness narratives: fact of fiction?, *Sociology of Health and Illness*, 23(3): 263-85.

Canguilhem, G. (1989) The normal and the pathological. New York: Zone Books.

Cassell, E. (2004) *The nature of suffering and the goals of medicine*. New York: Oxford University Press.

Carver, R. (1995) What we talk about when we talk about love, in Carver, R., *Where I'm calling from.* London: Harvil Press, pp. 138-151.

Charon, R. (2001) Narrative medicine: A model for empathy, reflection, profession and trust, *The Journal of the American Medical Association*, 286(15): 1897-1902.

Clarke, A.E. (1991) Social worlds/arenas theory as organizational theory, in Maines,

D (ed) Social organization and social process: Essays in honor of Anselm Strauss.

New York: Aldine de Gruyter, pp. 119-158.

Clegg, S. (ed.) (2002) *Central currents in organizational theory*, Vol. 1-4. London: Sage.

Clough, P. T. and Halley, J. (eds) (2007) *The affective turn: Theorizing the social*. Durham: Duke University Press.

Cooke, H. (2007) Out there you're on stage: complaints and the management of emotion work in nursing, *International Journal of Work Organisation and Emotion*, 2(2): 145-160.

Cooren, F. (2004) Textual agency: How texts do things in organizational settings, *Organization*, 11(3): 373-393.

Corbin, J. (1991) Anselm Strauss: An intellectual biography, in Maines, D. (ed) *Social organization and social process: Essays in honor of Anselm Strauss*. New York: Aldine de Gruyter, pp. 17-42.

Crossley, N. (1998) Emotion and communicative action: Habermas, linguistic philosophy and existentialism, in Bendelow, G. and Williams, S. J. (eds) *Emotions in*

social life: Critical themes and contemporary issues. New York: Routledge, pp. 17-38.

Dagbladet Information (2011) Kvinder skal gå hjemme, 5-6. november, p. 15.

Dansk Sundhedsinstitut (DSI) (2008) Ledelse af cancerforløb.

(http://www.dsi.dk)

Dean, M. (1998) Administering ascetism – reworking the life of the unemployed citizen, in Dean, M. and Hindess, B. (eds) *Governing Australia – studies of contemporary govern-*

ment. Melbourne: Cambridge University Press, pp. 87–107.

Dean, M. (1999) *Governmentality: Power and rule in modern society*. London: Sage. Dean, M. (2002) Liberal government and authoritarianism, *Economy and Society*, 31(1): 37–61.

Dent, M. (1993) Professionalism, educated labour and the state: hospital medicine and the new managerialism, *The Sociological Review*, 41(2): 244-273.

Derain, J. D., Udris, D. and Udris, M. (2009) *Human Terrain*, The Watson Institute's Global Media Project.

Doolin, B. (2002) Enterprise culture, professional identity and the organizational control of hospital clinicians, *Organization Studies*, 23(3): 369-390.

Douglas, M. and Ney, S. (1998) *Missing persons: A critique of personhood in the social sciences*. Berkeley, California: University of California Press,

Du Gay, P. (2000a) In praise of bureaucracy. London: Sage.

Du Gay, P. (2000b) Enterprise and its futures: A response to Fournier and Grey, *Organization*, 7(1): 165-183.

Du Gay, P. (2008) 'Without affection or enthusiasm': Problems of involvement and attachment in responsive public management', *Organization*, 15(3): 335-353.

Du Gay, P. and McFall, L. (2008) Introduction, in McFall, L., Du Gay, P. and Carter, S. (eds) *Conduct: Sociology and social worlds*, Manchester: Manchester University Press, pp. 1-21.

Dumit, J. (2006) Illnesses you have to fight to get: Facts as forces in uncertain, emergent illnesses, *Social Science and Medicine*, 62(3): 577-90.

Elias, N. (1987) Involvement and detachment. Oxford: Basil Blackwell.

Elias, N. (2000 [1939]) *The Civilizing Process*. Revised edition, Oxford: Blackwell Publishing.

Elias, N. and Dunning, E. (1986) *Quest for excitement – Sport and leisure in the civilizing process*. Oxford, UK: Blackwell.

Fair, B. (2010) Morgellons: contested illness, diagnostic compromise and medicalisation, *Sociology of Health and Illness*, 32(4): 597-612.

Fineman, S. (1993) Organizations as Emotional Arenas, in Fineman, S. (ed.) *Emotion in Organizations*. London: Sage, pp. 9-36.

Fineman, S. (2006) Emotion and Organizing, in Clegg, S., Hardy, C., Lawrence, T. and Nord, W. *The Sage Handbook of Organization Studies*, Second edition, London: Sage, pp- 675-700.

Fineman, S. (2008) 'Whither Emotion?', in D. Barry and H. Hansen (eds.) *The Sage Handbook of New Approaches in Management and Organisation*. London: Sage, pp. 239-240.

Fitzgerald, L. and Ferlie, E. (2000) Professionals: Back to the future, *Human Relations*, 53(5): 713-739.

Fontana, A. and Frey, J.H. (2002) Interviewing: The art of science, in Denzin, N.K. and Lincoln, Y.S. (eds) *Handbook of Qualitative Research*, Second Edition. London: Sage, pp. 361-376.

Foucault, F. (1973) *The birth of the clinic*. London: Tavistock Publications.

Foucault, M. (1977) *Discipline and punish: The birth of a prison*. London: Penguin Books.

Game, A. and Metcalfe, A. (1996) Passionate Sociology. London: Sage.

Garfinkel, H. (1967) *Studies in etnomethodology*. Englewood Cliffs, N.J.: Prentice Hall.

Gaub, J. (2010) Pakkeforloeb, Journal of Danish Medical Association, 172(4): 273.

Giese- Davis, J. and Spiegel, D. (2003) Emotional expression and cancer progression, in Davidson, R., Scherer, K. R. and Goldsmith, H. H. (eds) *Handbook of affective sciences*, Oxford: Oxford University Press, pp. 1053-1083.

Gilbert, P. (2009) The compassionate mind: A new approach to life's challenges.

Vancouver: Raincoast Books.

Gilham, B. (2005) *Research Interviewing: The range of techniques*. Berkshire: Open University Press.

Gill, V.T., Pomerantz, A. and Denvir, P. (2010) Pre-emptive resistance: patients' participation in diagnostic sense-making activities, *Sociology of Health and Illness*, 32(1): 1-20.

Goffman, E. (1956) Embarrassment and social organization, *The American Journal of Sociology*, 62(3): 264-271.

Goffman, E. (1959) *The presentation of self in everyday life*. London: Penguin Books.

Goffman, E. (1967) Interaction ritual. New York: Pantheon Books.

Goffman, E. (1974). Frame Analysis. New York: Harper and Row.

Goffman, E. (1989) On fieldwork, *Journal of Contemporary Ethnography*, 18(2): 123-132.

Goffman, E. (1990b) Stigma. London: Penguin Books.

Greco, M. (2008) On the art of life: A vitalist reading of medical humanities, *The Sociological Review*, 56(Issue supplement s2): 23-45.

Greco, M. and Stenner, P. (2008) Introduction, in Greco, M. and Stenner, P. (Eds.) *Emotions: A social science reader*. New York: Routlegde, pp. 1-23.

Gubrium, J.F. and Holstein, J.A. (2001) Introduction: Trying Times, Troubled Selves, in Gubrium, J.F. and Holstein, J.A. (eds) *Institutional selves. Troubled identities in a postmodern world*. Oxford: Oxford University Press, pp. 1-23.

Gärtner, R. (2010) Early recovery and chronic sequelae after treatment for breast cancer. PhD thesis, Copenhagen: Faculty of Health Sciences, University of Copenhagen.

Hacking, I. (1986) Making Up People, in Heller, T.C., Sosna, M. and Wellby, D.E. (eds) *Reconstructing individualism*. California: Stanford University Press, pp. 222-36.

Halpern, J. (2001) From detached concern to Empathy: Humanzing medical practice. Oxford: Oxford University Press.

Harris M, Clegg S and Höpfl H (2011) Introduction: Managing Modernity: Beyond Bureaucracy, in Harris M, Clegg S and Höpfl H (eds) *Managing modernity: Beyond bureaucracy*. Oxford: Oxford University Press, pp. 1-11.

Harrison, S. (2002) New labour, modernisation and the medical labour process, *Journal of Social Policy*, 31(3): 465-485.

Harrison, S. and MacDonald, R. (2008) *The politics of health care in Britain*. London: Sage.

Hennis, W. (2000) *Max Weber's science of man: New studies for a biography of work*. Newbury, Berks: Threshold Press.

Hochschild, A. (1979) Emotion work, feeling rules, and social structure, *The American Journal of Sociology*, 85(3): 551-575.

Hochschild, A. (1983) The managed heart: Commercialization of human feeling.

London: University of California Press.

Hochschild, A. (1993) Preface, in Fineman, S. (ed.) *Emotion in organization*. London: Sage Publications, pp. ix-xiii.

Hochschild, A.R. (1998) The sociology of emotion as a way of seeing, in Williams, S. and Bendelow, G. (eds), *Emotions in social life*. London: Routledge, pp. 3-15.

Holstein, J.A. and Gubrium, J.F. (1997) Active Interviewing, in Silverman, D. (ed.) *Qualitative research: theory, method and practice*. London: Sage, pp. 140-62.

Holstein, J. A. and Gubrium, J. F. (2000) *The self we live by: Narrative identity in a postmodern world.* Oxford: Oxford University Press.

Holstein, J. A. and Gubrium, J. F. (2002) 'From the individual interview to the interview society'. In Holstein, J. A. and Gubrium, J. F. (eds) *Handbook of interview research, context and method.* Thousand Oaks, California: Sage, pp. 3-31.

Hunter, I. (2006) The History of Theory, Critical Inquiry, 33(4): 78-112.

Illouz, E. (2007) *Cold Intimacies: The making of emotional capitalism*. Cambridge: Polity Press.

Illouz, E. (2008) Saving the modern soul: therapy, emotions and the culture of self-help. Berkeley, California: University of California Press.

IKAS (Institut for Kvalitet og Akkreditering I Sundhedsvæsnet) (2009) *Den danske kvalitetsmodel: Akkrediteringsstandarder for sygehuse*, 1. version.

Jaggar, A. M. (1989) Love and knowledge: Emotion in Feminist Epistemology, *Inquiry*, 32(2): 151-176.

James, N. (1989), Emotional labour: skill and work in the social regulation of feelings, *Sociological Review*, 37(1): 15-42.

James, N. (1993) Divisions of Emotional Labour: Disclosure and Cancer, in Fineman, S. (ed.) *Emotion in Organizations*. London: Sage, pp. 94-118.

Jenkins, R. (1996) Social identity. London: Routledge.

Jenkins, R. (2000) Categorization: Identity, social process and epistemology, Current Sociology. *The Sociological Conundrum of the Category*, 48(3): 7-25.

Junge, A.G., Risoer, M.B., Toustrup, K. and Grau, C. (2010) Hoved-hals-kraeft-patienters oplevelse af accelererede patientforloeb, *Journal of Danish Medical Association*, 172(4): 274-278.

Junger, S. (2010) War. London: Fourth Estate.

Jutel, A. (2009) Sociology of a diagnosis: a preliminary review, *Sociology of Health and Illness*, 31(2): 278-99.

Jutel, A. (2010) Medical unexplained symptoms and the disease label. *Social Theory* and *Health*, 8(3): 229-45.

Järvinen, M. (2000) The Biographical illusion. Constructing meaning in qualitative interviews, *Qualitative Inquiry*, 6(3):370-91.

Kehlet, H. & Dahl J. B. (2003) Anaesthesia, surgery and challenges in postoperative recovery, *Lancet*, 362(9399):1921-1928

Kehlet, H. and Hoejgaard, L. (2004) Det sengeløse hospital. *Ugeskrift for laeger*, 166(51): 4706-4707.

Kehlet, H. and Wilmore, D.W. (2002) Multi-modal strategies to improve surgical outcome, *The American Journal of Surgery*, 183(6): 630-641

Kitzinger, J. (1994) The Methodology of focus groups: The importance of interaction between research participant, *Sociology of Health and Illness*, 16(1): 103-21.

Kirmayer, L.J. (1988) Mind and body as metaphors: hidden values in biomedicine, in Lock, M. and Gordon, D. (eds) *Biomedicine examined*. Dordrecht: Academic Publishers Kluwer, pp. 57-92.

Kuzmics, H. (1987) Civilization, state and bourgeois society: the theoretical contribution of Norbert Elias, *Theory, Culture and Society*, 4(2-3): 515-531.

Larson, E. B. and Yao, X. (2005) Clinical empathy as emotional labor in the patient-physician relationship, *The Journal of the American Medical Association*, 293(9): 1100-1106.

Latour, B. (1993) We have never been Modern. Cambridge, MA: Harvard University Press.

Lowe, J., Delbridge, R. and Oliver, N. (1997) High-performance manufacturing: evidence from the automotive components industry, *Organisation Studies*, 18(5): 783-798.

Lupton, D. (1997), Doctors on the medical profession, *Sociology of Health and Illness*, 19(4): 480-497.

Lupton, D. (1998) The emotional self. London: Sage.

Lupton, D. (2003a) Medicine as Culture. Second edition, London: Sage.

Lupton, D. (2003b) Your life in their hands: Trust in the medical encounter, in James, V. and Gabe, J. (eds) *Health and the sociology of emotions*. Oxford: Blackwell, pp. 157-172.

Madden, S. and Sim, J. (2006) Creating meaning in fibromyalgia syndrome, *Social Science and Medicine*, 63(11): 2962-73.

Mann, S. (2005) A health-care model of emotional labour: An evaluation of the literature and development of a model, *Journal of Health Organization and Management*, 19(4/5): 304-317.

Manning, P. (2008) Goffman on Organizations, *Organization Studies*, 29(5): 677-99. Mark, A. (2005) Organizing emotions in health care, *Journal of Health Organization and Management*, 19(4/5): 277-89.

May, C., Rapley, T., Moreira, T., Finch, T. and Heaven, B. (2006) Technogovernance: Evidence, subjectivity, and the clinical encounter in primary care medicine, *Social Science and Medicine*, 62(4): 1022-1030.

May, C. et al. (2004) Framing the doctor-patient relationship in chronic illness: a comparative study of general practitioners' accounts, *Sociology of Health and Illness*, 26(2): 135-158.

McPherson, S. and Armstrong, D. (2009) Negotiating 'depression' in primary care: A qualitative study, *Social Science and Medicine*, 69(8): 1137-43.

Mead, G. H. (1959) *Mind, Self and Society from the Standpoint of a Social Behaviorist*. Chicago: University of Chicago Press.

Mead, N. and Bower, P. (2000) Patient-centredness: A conceptual framework and review of the empirical literature, *Social Science and Medicine*, 51(7): 1087-1110. Mead, N. (2002) Producing patient-centred health care: Patient perspectives about health and illness and the physician-patient relationship, *Social Science and Medicine*, 55(2): 349-350.

Menzies, I.E.P. (1960) A Case-Study in the functioning of social systems as a defence against anxiety: A report on a study of the nursing service of a general hospital, *Human Relations*, 13(2): 95-121.

Mertz, B. G., Nielsen, J. Ø., Williams, H. (2009) Opereret for brystkræft og hjem efter halvanden dag, *Sygeplejersken*, 9: 45-49.

Mestrovic, S. G. (1997) Postemotional Society. London: Sage.

Michailakis, D. and Schirmer, W. (2010) Agents of their health? How the Swedish welfare state introduces expectations of individual responsibility, *Sociology of Health and Illness*, 32(6): 930-47.

Mik-Meyer, N. (2010) An illness of one's own: power and the negotiation of identity among social workers, doctors, and patients without a bio-medical diagnosis, *Journal of Power*, 3(2): 171-87.

Mik-Meyer, N. (2011) On being credible ill: Class and gender in illness stories among welfare officers and clients with medically unexplained symptoms, *Health Sociology Review*, 20(1): 28–40.

Mik-Meyer, N. and Roelsgaard Obling, A. (2012), "The negotiation of a sick role: General practitioners' classification of patients with medical unexplained symptoms", *Sociology of Health and Illness*, 34(8)

Miller, P. and Rose, N. (2008) *Governing the present*. Cambridge: Polity Press.

Mol, A. (2002) *The body multiple: ontology in practice*. Durham: Duke University Press.

More, E.S. and Milligan, M.A. (1994) *The Empathic practitioner: Empathy, Gender and Medicine*. New Brunswick, NJ: Rutgers University Press.

Montgomery Hunter, K. (2006) *How doctors think: Clinical judgment and the practice of medicine*. Oxford: Oxford University Press.

Montgomery Hunter, K. (2001) *Doctors' stories: The narrative structure of medical knowledge*. Princeton, N. J.: Princeton University Press

Morgan, G. (1986) *Images of organizations*. Thousand Oaks, California: Sage.

Nettleton, S. (2006) I just want permission to be ill: Towards a sociology of medically unexplained symptoms, *Social Science and Medicine*, 62(5): 1167-78.

Nettleton, S., Burrows, R. and Watt, I, (2008) How do you feel doctor? An analysis of emotional aspects of routine medical work, *Social Theory and Health*, 6(1): 18-36. Neumann, M. et al. (2009) Analyzing the 'nature' and specific effectiveness of clinical empathy: a theoretical overview and contribution towards a theory-based research agenda. *Patient Education and Counseling*, 74(3): 339-346.

Newton, T. (2001) Organization: The relevance of the limitations of Elias, *Organization*, 8(3): 467-495.

Olesen, V. and Bone, D. (1998) Emotions in rationalizing organizations: Conceptual notes from professional nursing in the USA, in Bendelow, G. and Williams, S. J. (eds) *Emotions in social life: Critical themes and contemporary issues*. London: Routledge, pp. 313-329.

Osborne, T. (1993) On liberalism, neo-liberalism and the liberal profession of medicine, *Economy*

and Society, 22(3): 345-356.

Overgaard, J. (2008) DBCG: The Danish Breast Cancer Cooperative Group – a 30-year struggle for better breast cancer treatment in Denmark, *Acta Oncologica*, 47(4): 491-496.

Parsons, T. (1951) The Social System. London: Routledge.

Parsons, T. (1978) Action theory and the human condition. New York: Free Press.

Peters, S., Rogers, A., Salmon, P., Gask, L., Dowrick, C., Towey, M., Clifford, R. and Moriss, R. (2008) What do patients choose to tell their doctors? Qualitative analysis of potential barriers to reattributing medically unexplained symptoms, *Journal of General Internal Medicine*, 24(4): 443-49.

Power, M. (1997) Audit Society. Oxford: Oxford University Press.

Prior, L. (2003) Using documents in social research. London: Sage

Putnam, L. L. and Mumby, D. K. (1993) Organizations, emotion and the myth of rationality, in Fineman, S. (ed.) *Emotion in Organizations*. London: Sage, pp. 36-58.

Region Hovedstaden (2008) *Landsdækkende undersøgelse af kræftpatienters oplevelser*, Enheden for Brugerundersøgelser på vegne af Ministeriet for Sundhed of Forebyggelse.

(http://www.patientoplevelser.dk/kraeft/rapport.pdf)

Rigshospitalet, Copenhagen University Hospital (2010) *Rigshospitalets Kvalitetsplan* 2009-2011, Monitoreringsplan, Copenhagen.

Ring, A., Dowrick, C.F., Humphris, G.M., Davies, J. and Salmon, P. (2005) The somatising effect of clinical consultation: what patients and doctors say and do not say when patients present medically unexplained physical symptoms, *Social Science of Medicine*, 61(1): 1501-15.

Roelsgaard Obling, A. (2010) Pakkede patientforløb og styringsambitioner i det danske sundhedsvæsen, in Kjaer, P. and Reff Pedersen, A. (eds) *Patienten i centrum*.

Copenhagen: Copenhagen Business School Press, pp. 101-129.

Rorty, A.O. (1986) Cartesian Passions and the Union of Mind and Body, in Rorty, A. O. (ed) *Essays on Descartes meditations*. Berkeley, California: University of California Press, pp. 513-34.

Rose, N. (1985) *The Psychological complex: Psychology, politics and society in England 1869-1939*. London: Routledge.

Rose, N. (1991) Governing by numbers: Figuring out democracy, *Accounting, Organizations and Society*, 16(7): 673-692.

Rose, N. and Miller, P. (1992) Political power beyond the state: problematic of government, *British Journal of Sociology*, 43(2): 173-205.

Rose, N. (1999a) Governing the soul: the shaping of the private self. London: Free Association Books.

Rose, N. (1999b) *Powers of Freedom. Reframing political thought.* Cambridge: Cambridge University Press.

Rose, N. (2007) *The politics of life itself – Biomedicine, power, and subjectivity in the Twenty-First Century*. Princeton, NJ: Princeton University Press.

Sabo, B., Joffres, M. and Williams, T. (2000) How to deal with medically unknown symptoms, *Western Journal of Medicine*, 172(2): 128-30.

Salmon, P., Ring, A., Humphris, G.M., Davies, J.C. and Dowrick, C.F. (2008) Primary care consultations about medically unexplained symptoms: how do patients indicate what they want?, *Journal of General Internal Medicine*, 24(4): 450-56.

Schafer, A. (2009) Medical humanities: demarcations, dilemmas and delights. *Medical Humanities*, 35(1): 3-4.

Schanz, H.J. (1999) Selvfølgeligheder: Aspekter ved moderniteten og metafysik. Århus: Modtryk.

Shilling, C. (2003) *The body and social theory*. Second edition. London: Sage Publications.

Smith, A.C., III and Kleinman, S. (1989) Managing emotion in medical school: Students' contact with the living and the death, *Social Psychology Quarterly*, 52(1): Special Issue: Sentiments, Affect and Emotion, pp. 56-69.

Smith, P. and Gray, D. (2000) *The emotional labour of nursing: how students and qualified nurses learn to care*. London: South Bank University.

Speed, E. (2011) Applying soft bureaucracy to rhetorics of choice: The UK NHS 1983-2007, in Clegg, S., Harris, M. and Höpfl, H. (eds) *Managing modernity – Be- yond bureaucracy*. Oxford: Oxford University Press, pp. 81-104.

Star, S. L. (1991) The sociology of the invisible: The primacy of work in the writings of Anselm Strauss, in Maines, D. (ed) *Social organization and social process: Essays in honor of Anselm Strauss*. Hawthorne, NY: Aldine de Gruyter, pp. 265-283.

Star, S. L. and Strauss, A. (1999) Layers of silence, arenas of voice: The ecology of visible and invisible work, *Computer Supported Cooperative Work*, 8(1-2): 9-30.

Strauss, A., Fagerhaugh, S., Suczek, B. and Wiener, C. (1982) Sentimental work in the technologized hospital, *Sociology of Health and Illness*, 4(3): 254-278.

Strauss, A., Fagerhaugh, S., Suczek, B. and Wiener, C. (1997) *Social organization of medical work*. Chicago: The University of Chicago Press.

Taylor, F. W. (1967 [1911]) *The principles of scientific management*. New York: Norton Library.

Teherani, A., Hauer, K. E. and O'Sullivan, P. (2008) Can simulations measure empathy? Considerations on how to assess behavioral empathy via simulations, *Patient education and counseling*, 71(2): 148-152.

Terpe, S. and Paierl, S. (2010) From bureaucratic agencies to modern service providers: the emotional consequences of the reformation of labour administration in Germany, in Sieben, B. and Wettergren, A. A. (eds) *Emotionalizing Organisations and Organizing Emotions*. Hampshire: Palgrave MacMillan, pp. 209-230.

The Cancer Clinic, Rigshospitalet (2009) *Model program: tumor mammae*, intern document, Copenhagen.

The Capital Region of Denmark (2010) Kvalitet først, Copenhagen.

(http://www.regionh.dk/menu/sundhedOghospitaler/Politikker+planer+og+strategier/Kvalitetspolitik+og+handlingsplan.htm).

The Danish Cancer Society (2006) *Kraeftpatientens verden*, Forskningsenheden, Bispebjerg Hospital

(http://www.cancer.dk/NR/rdonlyres/60C1A9F7-8B1A-400A-8455-

547180CB0189/0/behovsundersogkort2006.pdf)

The Danish Government (2009) Sundhedspakke 2009, Copenhagen.

(http://www.sum.dk/Aktuelt/Publikationer/~/media/Filer%20-

% 20 dokumenter/Sundhedspakken 09/Sundhedspakke 09. ashx)

The Danish Ministry of Finance (2010) *Aftaler om Finansloven for 2011*, Copenhagen.

(http://www.fm.dk/Publikationer.aspx)

The Danish Ministry of Interior and Health (2003) *Patientens møde med sundheds-væsnet*, Arhus: Højbjerg Bogtryk.

The Danish Survey of Patient Experiences, The Capital Region of Denmark, 2010. (http://www.patientoplevelser.dk)

The Ministry of Health and Interior (2007) *Bedre velfærd og større arbejdsglæde - Regeringens strategi for høj kvalitet i den offentlige service.*

(http://www.kvalitetsreform.dk/page.dsp?page=428)

The National Board of Health, Denmark (2000) National Kraeft Plan I

(http://www.sst.dk/publ/Publ2000/Kraeft/National_kraeftplan_1.pdf)

The National Board of Health, Denmark (2005a) National Kraeft Plan II

(http://www.sst.dk/publ/publ2005/plan/kraeftplan2/kraeftplan2.pdf)

The National Board of Health, Denmark (2005b) *Opfoelgning paa Kraeft Plan II* (http://www.sst.dk/publ/publ2005/plan/kraeftplan2/kraeftplan2.pdf)

The National Board of Health, Denmark (2008a) Sundhedsfaglige elementer som grundlag for pakkeforløb for brystkræft, Copenhagen

(http://www.dsam.dk/files/30/b07_brystkraeft.pdf)

The National Board of Health, Denmark (2008b) Akut behandling og klar besked:

Generelle rammer for indførelse af pakkeforløb for kræftpatienter, Copenhagen.

(http://www.sst.dk/~/media/Planlaegning%20og%20kvalitet/Kraeftbehandling/Om%20pakkeforloeb/akut%20handling%20og%20klar%20besked_generelle%20rammer%20for%20indfoerelse%20af%20pakkeforloeb%20for%20kraeftpatienter.ashx)

The The National Board of Health, Denmark (2008c) *Notat: Generelle indsatser ved*rørende psykosocial omsorg, pleje og symptomlindring, rehabilitering og palliation i relation til sundhedsfaglige elementer i pakkeforløb for kræft.

(http://www.sst.dk/~/media/Planlaegning%20og%20kvalitet/Kraeftbehandling/Om%20pakkeforloeb/notat%20om%20psykosocial%20indsats%20pleje%20og%20sympto mlindring%20rehabilitering%20og%20palliation%20%2021%2004%2008%20final.a shx)

The National Board of Health, Denmark (2009a) *Pakkeforloeb for Brystkraeft*, Copenhagen

(http://www.sst.dk/publ/Publ2009/SUPL/Pakke_kraeft/Kraeft_bryst_sep09.pdf)

The National Board of Health, Denmark (2009b) Sammenhaeng mellem kliniske retningslinijer og patientforloebsbeskrivelser, Copenhagen.

(https://www.sst.dk/~/media/Behandligsforloeb%20og%20rettigheder/Kliniske%20re tningslini-

er/Sammenh%C3%A6ng%20mellem%20kliniske%20retningslinjer%20og%20patien tforl%C3%B8bsbeskrivelser%20final.ashx)

The National Board of Health, Denmark (2010a) *Monitorering af kræftområdet: tid* fra henvisning modtaget på sygehus til start af behandling.

(http://www.sst.dk/publ/Publ2010/EMM/Kraeft/Monitorering kraeft rapport.pdf)

The National Board of Health, Denmark (2010b) *Notat: Monitorering af pakkeforløb* for kræftpatienter.

 $(http://www.sst.dk/\sim/media/Planlaegning\%20og\%20kvalitet/Maaling\%20af\%20kvalitet/pakkeforloeb\%20for\%20kraeftpatienter/Dokumenter\%20relateret\%20til\%20Monitoreringsinformationssyste-\\$

met%20MIS%20Kraeft/Monitorering%20af%20pakkeforl%C3%B8b%20for%20kr%C3%A6ftpatienter.ashx)

The National Board of Health, Denmark (2011) *Metodebeskrivelse: Monitorering af kræftområdet*,

(http://www.sst.dk/~/media/Planlaegning%20og%20kvalitet/Maaling%20af%20kvalitet/pakkeforloeb%20for%20kraeftpatienter/Monitorering%20af%20kr%C3%A6ftomr%C3%A5det%202011%200111.ashx)

Theodosius, C. (2006) Recovering Emotion from Emotion Management, *Sociology*, 40(5): 893-910.

Theodosius, C. (2008) Emotional labour in health care. Abingdon: Routledge.

The Secretary of Ministers (2007) Kvalitet i den offentlige sektor: Fakta om accelererede forløb. (http://www.servicereform.dk/multimedia/Faktabilag_accelererede_patientforl_b.pdf) Townley, B. (2008) *Reason's neglect: Rationality and Organizing*. Oxford: Oxford University Press.

Updike, J. (1995) Introduction, in Shem, S. *The House of God*. London: Black Swan, pp. 6-8.

Van Iterson, A. V. (2009) Norbert Elias's impact on Organization Studies, in Adler, P. (ed.) *The Oxford Handbook of Sociology and Organization Studies, Classical Foundations*. Oxford: Oxford University Press, pp. 327-348.

Van Iterson, A., Mastenbroek, W. and Soeters, J. (2001) Civilizing and informalizing: Organizations in an Eliasian context, *Organization*, 8(3): 497-514.

Van Krieken, R. (1990) The organisation of the soul: Elias and Foucault on discipline and the self, *Archieves Europeénes du Sociologie*, 31(2): 353-71.

Van Krieken, R. (1996) Proto-governmentalization and the historical formation of organizational subjectivity, *Economy and Society*, 25(2): 195-221.

Van Vree, W. (2011) Meetings: the frontline of civilization, *The Sociological Review*, 59(Issue Supplement s1): 241-262.

Wainwright, D., Calnan, M., O'Neil, C., Winterbottom, A. and Watkins, C. (2006) When pain in the arm is 'all in the head': the management of medically unexplained suffering in primary care, *Health, Risk and Society*, 8(1): 71-88.

Waring, J., Dixon-Woods, M. and Young, K. (2010) Modernising medical regulation: Where are we now, *Journal of Health Organization and Management*, 24(6): 540-555.

Watson, S. (1999) Policing the Affective Society: beyond governmentality in the theory of social control, *Social and Legal Studies*, 8(2): 227-251.

Weber, M. (1978) *Economy and Society, vol. I-II*. Berkeley, California: University of Berkeley Press.

Weick, K. (2007), Drop your Tools: On refiguring management education, *Journal of Management Education*, 31(1): 5-16.

Werner, A. and Malterud, K. (2003) It is hard work behaving as a credible patient: encounters between women with chronic pain and their doctors, *Social Science and Medicine*, 57(8): 1409-19.

Wilemana, L., Mayb, C. and Chew-Grahama, C. (2002) Medically unexplained symptoms and the problem of power in the primary consultation: a qualitative study, *The Journal of Family Practice*, 19(2): 178-82.

Williams, S. (2001) Emotion and social theory. London: Sage.

Williams, S. and Bendelow, G. (1998) Introduction, in Williams, S. and Bendelow, G. (eds) *Emotions in social life*. Routledge, London, pp. xv-xxi.

Wittgenstein, L. (1999 [1957]) Philosophical Investigations. Oxford: Blackwell.

Wouters, C. (1986) Formalization and informalization: Changing tension balances in civilizing processes, *Theory, Culture and Society*, 3(2): 1-18.

Wouters, C. (1989) The sociology of emotions and flight attendants: Hochschild's managed heart, *Theory, Culture and Society*, 6(1): 95-123.

Wouters, C. (1999) Changing patterns of social controls and self-controls, *British Journal of Criminology*, 39(3): 416-432.

Wouters, C. (2009) The civilizing of emotions: Formalization and informalization, in Hopkins, D., Kuzmics, H. and Flam, H. (eds) *Theorizing emotions*. New York: Campus Verlag, pp. 169-193.

Wright Mills, C. (1959) *The sociological imagination*. Middlesex, UK: Penguin Books.

Wouters, C. (2011) How civilizing processes continued: towards an informalization of manners and a third nature personality, *The Sociological Review*, 59(Issue supplement, s1): 140-159.

Wulff, H. and Goetzshe, P. (2000) *Rational diagnosis and treatment: Evidence based clinical decision making*. London: Wiley-Blackwell.

TITLER I PH.D.SERIEN:

2004

- Martin Grieger
 Internet-based Electronic Marketplaces
 and Supply Chain Management
- 2. Thomas Basbøll LIKENESS A Philosophical Investigation
- 3. Morten Knudsen
 Beslutningens vaklen
 En systemteoretisk analyse of moderniseringen af et amtskommunalt
 sundhedsvæsen 1980-2000
- 4. Lars Bo Jeppesen
 Organizing Consumer Innovation
 A product development strategy that
 is based on online communities and
 allows some firms to benefit from a
 distributed process of innovation by
 consumers
- 5. Barbara Dragsted
 SEGMENTATION IN TRANSLATION
 AND TRANSLATION MEMORY
 SYSTEMS
 An empirical investigation of cognitive
 segmentation and effects of integrating a TM system into the translation
 process
- 6. Jeanet Hardis
 Sociale partnerskaber
 Et socialkonstruktivistisk casestudie
 af partnerskabsaktørers virkelighedsopfattelse mellem identitet og
 legitimitet
- 7. Henriette Hallberg Thygesen System Dynamics in Action
- 8. Carsten Mejer Plath Strategisk Økonomistyring
- 9. Annemette Kjærgaard Knowledge Management as Internal Corporate Venturing

- a Field Study of the Rise and Fall of a Bottom-Up Process
- Knut Arne Hovdal
 De profesjonelle i endring
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur
- Søren Jeppesen
 Environmental Practices and Greening
 Strategies in Small Manufacturing
 Enterprises in South Africa
 A Critical Realist Approach
- Lars Frode Frederiksen
 Industriel forskningsledelse
 – på sporet af mønstre og samarbejde
 i danske forskningsintensive virksomheder
- 13. Martin Jes Iversen
 The Governance of GN Great Nordic
 in an age of strategic and structural
 transitions 1939-1988
- 14. Lars Pynt Andersen
 The Rhetorical Strategies of Danish TV
 Advertising
 A study of the first fifteen years with
 special emphasis on genre and irony
- 15. Jakob Rasmussen *Business Perspectives on E-learning*
- Sof Thrane
 The Social and Economic Dynamics of Networks
 - a Weberian Analysis of Three Formalised Horizontal Networks
- 17. Lene Nielsen
 Engaging Personas and Narrative
 Scenarios a study on how a usercentered approach influenced the
 perception of the design process in
 the e-business group at AstraZeneca
- 18. S.J Valstad

 Organisationsidentitet

 Norsk ph.d., ej til salg gennem
 Samfundslitteratur

- 19. Thomas Lyse Hansen
 Six Essays on Pricing and Weather risk
 in Energy Markets
- 20. Sabine Madsen
 Emerging Methods An Interpretive
 Study of ISD Methods in Practice
- 21. Evis Sinani
 The Impact of Foreign Direct Investment on Efficiency, Productivity
 Growth and Trade: An Empirical Investigation
- 22. Bent Meier Sørensen Making Events Work Or, How to Multiply Your Crisis
- 23. Pernille Schnoor
 Brand Ethos
 Om troværdige brand- og
 virksomhedsidentiteter i et retorisk og
 diskursteoretisk perspektiv
- 24. Sidsel Fabech
 Von welchem Österreich ist hier die
 Rede?
 Diskursive forhandlinger og magtkampe mellem rivaliserende nationale
 identitetskonstruktioner i østrigske
 pressediskurser
- 25. Klavs Odgaard Christensen Sprogpolitik og identitetsdannelse i flersprogede forbundsstater Et komparativt studie af Schweiz og Canada
- 26. Dana B. Minbaeva Human Resource Practices and Knowledge Transfer in Multinational Corporations
- 27. Holger Højlund

 Markedets politiske fornuft

 Et studie af velfærdens organisering i
 perioden 1990-2003
- 28. Christine Mølgaard Frandsen
 A.s erfaring
 Om mellemværendets praktik i en

- transformation af mennesket og subjektiviteten
- Sine Nørholm Just
 The Constitution of Meaning
 – A Meaningful Constitution?
 Legitimacy, identity, and public opinion in the debate on the future of Europe

- Claus J. Varnes
 Managing product innovation through
 rules The role of formal and structured methods in product development
- Helle Hedegaard Hein
 Mellem konflikt og konsensus

 Dialogudvikling på hospitalsklinikker
- 3. Axel Rosenø

 Customer Value Driven Product Innovation A Study of Market Learning in New Product Development
- Søren Buhl Pedersen
 Making space
 An outline of place branding
- 5. Camilla Funck Ellehave Differences that Matter An analysis of practices of gender and organizing in contemporary workplaces
- 6. Rigmor Madeleine Lond Styring af kommunale forvaltninger
- 7. Mette Aagaard Andreassen Supply Chain versus Supply Chain Benchmarking as a Means to Managing Supply Chains
- 8. Caroline Aggestam-Pontoppidan
 From an idea to a standard
 The UN and the global governance of
 accountants' competence
- 9. Norsk ph.d.
- 10. Vivienne Heng Ker-ni
 An Experimental Field Study on the

Effectiveness of Grocer Media Advertising Measuring Ad Recall and Recognition, Purchase Intentions and Short-Term Sales

- 11. Allan Mortensen
 Essays on the Pricing of Corporate
 Bonds and Credit Derivatives
- 12. Remo Stefano Chiari
 Figure che fanno conoscere
 Itinerario sull'idea del valore cognitivo
 e espressivo della metafora e di altri
 tropi da Aristotele e da Vico fino al
 cognitivismo contemporaneo
- Anders McIlquham-Schmidt Strategic Planning and Corporate Performance An integrative research review and a meta-analysis of the strategic planning and corporate performance literature from 1956 to 2003
- 14. Jens Geersbro
 The TDF PMI Case
 Making Sense of the Dynamics of
 Business Relationships and Networks
- 15 Mette Andersen
 Corporate Social Responsibility in
 Global Supply Chains
 Understanding the uniqueness of firm
 behaviour
- 16. Eva Boxenbaum Institutional Genesis: Micro – Dynamic Foundations of Institutional Change
- 17. Peter Lund-Thomsen

 Capacity Development, Environmental

 Justice NGOs, and Governance: The

 Case of South Africa
- 18. Signe Jarlov

 Konstruktioner af offentlig ledelse
- Lars Stæhr Jensen
 Vocabulary Knowledge and Listening
 Comprehension in English as a Foreign
 Language

- An empirical study employing data elicited from Danish EFL learners
- Christian Nielsen
 Essays on Business Reporting
 Production and consumption of
 strategic information in the market for
 information
- 21. Marianne Thejls Fischer
 Egos and Ethics of Management
 Consultants
- 22. Annie Bekke Kjær
 Performance management i Procesinnovation
 belyst i et social-konstruktivistisk
 perspektiv
- 23. Suzanne Dee Pedersen
 GENTAGELSENS METAMORFOSE
 Om organisering af den kreative gøren
 i den kunstneriske arbejdspraksis
- 24. Benedikte Dorte Rosenbrink
 Revenue Management
 Økonomiske, konkurrencemæssige &
 organisatoriske konsekvenser
- 25. Thomas Riise Johansen
 Written Accounts and Verbal Accounts
 The Danish Case of Accounting and
 Accountability to Employees
- 26. Ann Fogelgren-Pedersen The Mobile Internet: Pioneering Users' Adoption Decisions
- 27. Birgitte Rasmussen
 Ledelse i fællesskab de tillidsvalgtes
 fornyende rolle
- 28. Gitte Thit Nielsen
 Remerger
 skabende ledelseskræfter i fusion og
 opkøb
- 29. Carmine Gioia
 A MICROECONOMETRIC ANALYSIS OF
 MERGERS AND ACQUISITIONS

- 30. Ole Hinz
 Den effektive forandringsleder: pilot,
 pædagog eller politiker?
 Et studie i arbejdslederes meningstilskrivninger i forbindelse med vellykket
 gennemførelse af ledelsesinitierede
 forandringsprojekter
- 31. Kjell-Åge Gotvassli
 Et praksisbasert perspektiv på dynamiske
 læringsnettverk i toppidretten
 Norsk ph.d., ej til salg gennem
 Samfundslitteratur
- 32. Henriette Langstrup Nielsen
 Linking Healthcare
 An inquiry into the changing performances of web-based technology for asthma monitoring
- 33. Karin Tweddell Levinsen Virtuel Uddannelsespraksis Master i IKT og Læring – et casestudie i hvordan proaktiv proceshåndtering kan forbedre praksis i virtuelle læringsmiljøer
- 34. Anika Liversage Finding a Path Labour Market Life Stories of Immigrant Professionals
- 35. Kasper Elmquist Jørgensen Studier i samspillet mellem stat og erhvervsliv i Danmark under 1. verdenskrig
- 36. Finn Janning
 A DIFFERENT STORY
 Seduction, Conquest and Discovery
- 37. Patricia Ann Plackett
 Strategic Management of the Radical
 Innovation Process
 Leveraging Social Capital for Market
 Uncertainty Management

Christian Vintergaard
 Early Phases of Corporate Venturing

- 2. Niels Rom-Poulsen Essays in Computational Finance
- 3. Tina Brandt Husman
 Organisational Capabilities,
 Competitive Advantage & ProjectBased Organisations
 The Case of Advertising and Creative
 Good Production
- Mette Rosenkrands Johansen
 Practice at the top
 – how top managers mobilise and use non-financial performance measures
- Eva Parum
 Corporate governance som strategisk kommunikations- og ledelsesværktøj
- 6. Susan Aagaard Petersen
 Culture's Influence on Performance
 Management: The Case of a Danish
 Company in China
- 7. Thomas Nicolai Pedersen
 The Discursive Constitution of Organizational Governance Between unity
 and differentiation
 The Case of the governance of
 environmental risks by World Bank
 environmental staff
- 8. Cynthia Selin Volatile Visions: Transactons in Anticipatory Knowledge
- 9. Jesper Banghøj Financial Accounting Information and Compensation in Danish Companies
- 10. Mikkel Lucas Overby Strategic Alliances in Emerging High-Tech Markets: What's the Difference and does it Matter?
- 11. Tine Aage
 External Information Acquisition of
 Industrial Districts and the Impact of
 Different Knowledge Creation Dimensions

A case study of the Fashion and Design Branch of the Industrial District of Montebelluna, NE Italy

- 12. Mikkel Flyverbom
 Making the Global Information Society
 Governable
 On the Governmentality of MultiStakeholder Networks
- 13. Anette Grønning
 Personen bag
 Tilstedevær i e-mail som interaktionsform mellem kunde og medarbejder i dansk forsikringskontekst
- 14. Jørn Helder
 One Company One Language?
 The NN-case
- 15. Lars Bjerregaard Mikkelsen
 Differing perceptions of customer
 value
 Development and application of a tool
 for mapping perceptions of customer
 value at both ends of customer-supplier dyads in industrial markets
- 16. Lise Granerud
 Exploring Learning
 Technological learning within small
 manufacturers in South Africa
- 17. Esben Rahbek Pedersen
 Between Hopes and Realities:
 Reflections on the Promises and
 Practices of Corporate Social
 Responsibility (CSR)
- 18. Ramona Samson
 The Cultural Integration Model and
 European Transformation.
 The Case of Romania

2007

Jakob Vestergaard
 Discipline in The Global Economy
 Panopticism and the Post-Washington
 Consensus

- Heidi Lund Hansen
 Spaces for learning and working
 A qualitative study of change of work,
 management, vehicles of power and
 social practices in open offices
- 3. Sudhanshu Rai
 Exploring the internal dynamics of
 software development teams during
 user analysis
 A tension enabled Institutionalization
 Model; "Where process becomes the
 objective"
- 4. Norsk ph.d. Ej til salg gennem Samfundslitteratur
- 5. Serden Ozcan

 EXPLORING HETEROGENEITY IN

 ORGANIZATIONAL ACTIONS AND

 OUTCOMES

 A Behavioural Perspective
- Kim Sundtoft Hald
 Inter-organizational Performance
 Measurement and Management in
 Action
 — An Ethnography on the Construction of Management, Identity and
 Relationships
- 7. Tobias Lindeberg
 Evaluative Technologies
 Quality and the Multiplicity of
 Performance
- 8. Merete Wedell-Wedellsborg
 Den globale soldat
 Identitetsdannelse og identitetsledelse
 i multinationale militære organisationer
- Lars Frederiksen
 Open Innovation Business Models
 Innovation in firm-hosted online user
 communities and inter-firm project
 ventures in the music industry
 A collection of essays
- 10. Jonas Gabrielsen
 Retorisk toposlære fra statisk 'sted'
 til persuasiv aktivitet

- Christian Moldt-Jørgensen
 Fra meningsløs til meningsfuld
 evaluering.
 Anvendelsen af studentertilfredsheds målinger på de korte og mellemlange
 videregående uddannelser set fra et
 psykodynamisk systemperspektiv
- 12. Ping Gao
 Extending the application of
 actor-network theory
 Cases of innovation in the telecommunications industry
- 13. Peter Mejlby
 Frihed og fængsel, en del af den
 samme drøm?
 Et phronetisk baseret casestudie af
 frigørelsens og kontrollens sameksistens i værdibaseret ledelse!
- 14. Kristina Birch Statistical Modelling in Marketing
- 15. Signe Poulsen Sense and sensibility: The language of emotional appeals in insurance marketing
- 16. Anders Bjerre Trolle

 Essays on derivatives pricing and dynamic asset allocation
- 17. Peter Feldhütter
 Empirical Studies of Bond and Credit
 Markets
- 18. Jens Henrik Eggert Christensen
 Default and Recovery Risk Modeling
 and Estimation
- 19. Maria Theresa Larsen Academic Enterprise: A New Mission for Universities or a Contradiction in Terms? Four papers on the long-term implications of increasing industry involvement and commercialization in academia

- Morten Wellendorf
 Postimplementering af teknologi i den offentlige forvaltning
 Analyser af en organisations kontinuerlige arbejde med informationsteknologi
- 21. Ekaterina Mhaanna Concept Relations for Terminological Process Analysis
- 22. Stefan Ring Thorbjørnsen
 Forsvaret i forandring
 Et studie i officerers kapabiliteter under påvirkning af omverdenens forandringspres mod øget styring og læring
- 23. Christa Breum Amhøj

 Det selvskabte medlemskab om managementstaten, dens styringsteknologier og indbyggere
- 24. Karoline Bromose
 Between Technological Turbulence and
 Operational Stability
 – An empirical case study of corporate
 venturing in TDC
- Susanne Justesen
 Navigating the Paradoxes of Diversity in Innovation Practice
 – A Longitudinal study of six very different innovation processes in practice
- Luise Noring Henler
 Conceptualising successful supply
 chain partnerships
 – Viewing supply chain partnerships
 from an organisational culture perspective
- 27. Mark Mau
 Kampen om telefonen
 Det danske telefonvæsen under den
 tyske besættelse 1940-45
- 28. Jakob Halskov

 The semiautomatic expansion of
 existing terminological ontologies
 using knowledge patterns discovered

- on the WWW an implementation and evaluation
- 29. Gergana Koleva European Policy Instruments Beyond Networks and Structure: The Innovative Medicines Initiative
- 30. Christian Geisler Asmussen Global Strategy and International Diversity: A Double-Edged Sword?
- 31. Christina Holm-Petersen Stolthed og fordom Kultur- og identitetsarbejde ved skabelsen af en ny sengeafdeling gennem fusion
- 32. Hans Peter Olsen
 Hybrid Governance of Standardized
 States
 Causes and Contours of the Global
 Regulation of Government Auditing
- 33. Lars Bøge Sørensen Risk Management in the Supply Chain
- 34. Peter Aagaard
 Det unikkes dynamikker
 De institutionelle mulighedsbetingelser bag den individuelle udforskning i
 professionelt og frivilligt arbejde
- 35. Yun Mi Antorini

 Brand Community Innovation

 An Intrinsic Case Study of the Adult
 Fans of LEGO Community
- 36. Joachim Lynggaard Boll Labor Related Corporate Social Performance in Denmark Organizational and Institutional Perspectives

- 1. Frederik Christian Vinten Essays on Private Equity
- 2. Jesper Clement Visual Influence of Packaging Design on In-Store Buying Decisions

- Marius Brostrøm Kousgaard Tid til kvalitetsmåling?

 Studier af indrulleringsprocesser i forbindelse med introduktionen af kliniske kvalitetsdatabaser i speciallægepraksissektoren
- 4. Irene Skovgaard Smith

 Management Consulting in Action
 Value creation and ambiguity in
 client-consultant relations
- 5. Anders Rom
 Management accounting and integrated information systems
 How to exploit the potential for management accounting of information technology
- Marina Candi
 Aesthetic Design as an Element of
 Service Innovation in New Technologybased Firms
- Morten Schnack
 Teknologi og tværfaglighed – en analyse af diskussionen omkring
 indførelse af EPJ på en hospitalsafde ling
- 8. Helene Balslev Clausen Juntos pero no revueltos – un estudio sobre emigrantes norteamericanos en un pueblo mexicano
- 9. Lise Justesen

 Kunsten at skrive revisionsrapporter.

 En beretning om forvaltningsrevisionens beretninger
- Michael E. Hansen
 The politics of corporate responsibility:
 CSR and the governance of child labor and core labor rights in the 1990s
- 11. Anne Roepstorff

 Holdning for handling en etnologisk

 undersøgelse af Virksomheders Sociale

 Ansvar/CSR

- 12. Claus Bajlum
 Essays on Credit Risk and
 Credit Derivatives
- Anders Bojesen
 The Performative Power of Competence an Inquiry into Subjectivity and Social Technologies at Work
- 14. Satu Reijonen
 Green and Fragile
 A Study on Markets and the Natural
 Environment
- 15. Ilduara Busta Corporate Governance in Banking A European Study
- 16. Kristian Anders Hvass
 A Boolean Analysis Predicting Industry
 Change: Innovation, Imitation & Business Models
 The Winning Hybrid: A case study of isomorphism in the airline industry
- 17. Trine Paludan
 De uvidende og de udviklingsparate
 Identitet som mulighed og restriktion
 blandt fabriksarbejdere på det aftayloriserede fabriksgulv
- 18. Kristian Jakobsen
 Foreign market entry in transition economies: Entry timing and mode choice
- 19. Jakob Elming
 Syntactic reordering in statistical machine translation
- 20. Lars Brømsøe Termansen
 Regional Computable General Equilibrium Models for Denmark
 Three papers laying the foundation for
 regional CGE models with agglomeration characteristics
- 21. Mia Reinholt The Motivational Foundations of Knowledge Sharing

- 23. Peter D. Ørberg Jensen
 OFFSHORING OF ADVANCED AND
 HIGH-VALUE TECHNICAL SERVICES:
 ANTECEDENTS, PROCESS DYNAMICS
 AND FIRMLEVEL IMPACTS
- 24. Pham Thi Song Hanh
 Functional Upgrading, Relational
 Capability and Export Performance of
 Vietnamese Wood Furniture Producers
- 25. Mads Vangkilde Why wait? An Exploration of first-mover advantages among Danish e-grocers through a resource perspective
- 26. Hubert Buch-Hansen
 Rethinking the History of European
 Level Merger Control
 A Critical Political Economy Perspective

- Vivian Lindhardsen
 From Independent Ratings to Communal Ratings: A Study of CWA Raters'
 Decision-Making Behaviours
- Guðrið Weihe Public-Private Partnerships: Meaning and Practice
- 3. Chris Nøkkentved
 Enabling Supply Networks with Collaborative Information Infrastructures
 An Empirical Investigation of Business
 Model Innovation in Supplier Relationship Management
- 4. Sara Louise Muhr Wound, Interrupted – On the Vulnerability of Diversity Management

- 5. Christine Sestoft
 Forbrugeradfærd i et Stats- og Livsformsteoretisk perspektiv
- 6. Michael Pedersen
 Tune in, Breakdown, and Reboot: On
 the production of the stress-fit selfmanaging employee
- 7. Salla Lutz
 Position and Reposition in Networks

 Exemplified by the Transformation of
 the Danish Pine Furniture Manufacturers
- 8. Jens Forssbæck
 Essays on market discipline in
 commercial and central banking
- 9. Tine Murphy
 Sense from Silence A Basis for Organised Action
 How do Sensemaking Processes with
 Minimal Sharing Relate to the Reproduction of Organised Action?
- 10. Sara Malou Strandvad
 Inspirations for a new sociology of art:
 A sociomaterial study of development
 processes in the Danish film industry
- Nicolaas Mouton
 On the evolution of social scientific metaphors:
 A cognitive-historical enquiry into the divergent trajectories of the idea that collective entities states and societies, cities and corporations are biological organisms.
- 12. Lars Andreas Knutsen Mobile Data Services: Shaping of user engagements
- 13. Nikolaos Theodoros Korfiatis Information Exchange and Behavior A Multi-method Inquiry on Online Communities

- 14. Jens Albæk
 Forestillinger om kvalitet og tværfaglighed på sygehuse
 skabelse af forestillinger i læge- og
 plejegrupperne angående relevans af
 nye idéer om kvalitetsudvikling gennem tolkningsprocesser
 - Maja Lotz
 The Business of Co-Creation and the Co-Creation of Business
 - 16. Gitte P. Jakobsen
 Narrative Construction of Leader Identity in a Leader Development Program
 Context
- 17. Dorte Hermansen
 "Living the brand" som en brandorienteret dialogisk praxis:
 Om udvikling af medarbejdernes
 brandorienterede dømmekraft
- 18. Aseem Kinra
 Supply Chain (logistics) Environmental
 Complexity
- 19. Michael Nørager How to manage SMEs through the transformation from non innovative to innovative?
- 20. Kristin Wallevik Corporate Governance in Family Firms The Norwegian Maritime Sector
- 21. Bo Hansen Hansen
 Beyond the Process
 Enriching Software Process Improvement with Knowledge Management
- 22. Annemette Skot-Hansen
 Franske adjektivisk afledte adverbier,
 der tager præpositionssyntagmer indledt med præpositionen à som argumenter
 En valensgrammatisk undersøgelse
- 23. Line Gry Knudsen

 Collaborative R&D Capabilities

 In Search of Micro-Foundations

- 24. Christian Scheuer
 Employers meet employees
 Essays on sorting and globalization
- 25. Rasmus Johnsen
 The Great Health of Melancholy
 A Study of the Pathologies of Performativity
- 26. Ha Thi Van Pham
 Internationalization, Competitiveness
 Enhancement and Export Performance
 of Emerging Market Firms:
 Evidence from Vietnam
- 27. Henriette Balieu Kontrolbegrebets betydning for kausativalternationen i spansk En kognitiv-typologisk analyse

- Yen Tran
 Organizing Innovationin Turbulent
 Fashion Market
 Four papers on how fashion firms create and appropriate innovation value
- Anders Raastrup Kristensen
 Metaphysical Labour
 Flexibility, Performance and Commitment in Work-Life Management
- Margrét Sigrún Sigurdardottir
 Dependently independent
 Co-existence of institutional logics in the recorded music industry
- Ásta Dis Óladóttir Internationalization from a small domestic base: An empirical analysis of Economics and Management
- 5. Christine Secher
 E-deltagelse i praksis politikernes og
 forvaltningens medkonstruktion og
 konsekvenserne heraf
- Marianne Stang Våland What we talk about when we talk about space:

- End User Participation between Processes of Organizational and Architectural Design
- 7. Rex Degnegaard
 Strategic Change Management
 Change Management Challenges in
 the Danish Police Reform
- 8. Ulrik Schultz Brix
 Værdi i rekruttering den sikre beslutning
 En pragmatisk analyse af perception
 og synliggørelse af værdi i rekrutterings- og udvælgelsesarbejdet
- Jan Ole Similä
 Kontraktsledelse
 Relasjonen mellom virksomhetsledelse
 og kontraktshåndtering, belyst via fire
 norske virksomheter
- 10. Susanne Boch Waldorff

 Emerging Organizations: In between
 local translation, institutional logics
 and discourse
- 11. Brian Kane
 Performance Talk
 Next Generation Management of
 Organizational Performance
- Lars Ohnemus
 Brand Thrust: Strategic Branding and Shareholder Value
 An Empirical Reconciliation of two Critical Concepts
- 13. Jesper Schlamovitz Håndtering af usikkerhed i film- og byggeprojekter
- 14. Tommy Moesby-Jensen Det faktiske livs forbindtlighed Førsokratisk informeret, ny-aristotelisk ἡθος-tænkning hos Martin Heidegger
- 15. Christian Fich
 Two Nations Divided by Common
 Values
 French National Habitus and the
 Rejection of American Power

- 16. Peter Beyer
 Processer, sammenhængskraft
 og fleksibilitet
 Et empirisk casestudie af omstillingsforløb i fire virksomheder
- 17. Adam Buchhorn
 Markets of Good Intentions
 Constructing and Organizing
 Biogas Markets Amid Fragility
 and Controversy
- 18. Cecilie K. Moesby-Jensen Social læring og fælles praksis Et mixed method studie, der belyser læringskonsekvenser af et lederkursus for et praksisfællesskab af offentlige mellemledere
- Heidi Boye
 Fødevarer og sundhed i sen modernismen
 En indsigt i hyggefænomenet og
 de relaterede fødevarepraksisser
- Kristine Munkgård Pedersen
 Flygtige forbindelser og midlertidige
 mobiliseringer
 Om kulturel produktion på Roskilde
 Festival
- 21. Oliver Jacob Weber Causes of Intercompany Harmony in Business Markets – An Empirical Investigation from a Dyad Perspective
- 22. Susanne Ekman
 Authority and Autonomy
 Paradoxes of Modern Knowledge
 Work
- 23. Anette Frey Larsen
 Kvalitetsledelse på danske hospitaler
 Ledelsernes indflydelse på introduktion og vedligeholdelse af kvalitetsstrategier i det danske sundhedsvæsen
- 24. Toyoko Sato

 Performativity and Discourse: Japanese

 Advertisements on the Aesthetic Education of Desire

- 25. Kenneth Brinch Jensen
 Identifying the Last Planner System
 Lean management in the construction
 industry
- 26. Javier Busquets
 Orchestrating Network Behavior
 for Innovation
- 27. Luke Patey
 The Power of Resistance: India's National Oil Company and International
 Activism in Sudan
- 28. Mette Vedel
 Value Creation in Triadic Business Relationships. Interaction, Interconnection
 and Position
- 29. Kristian Tørning
 Knowledge Management Systems in
 Practice A Work Place Study
- 30. Qingxin Shi An Empirical Study of Thinking Aloud Usability Testing from a Cultural Perspective
- 31. Tanja Juul Christiansen Corporate blogging: Medarbejderes kommunikative handlekraft
- 32. Malgorzata Ciesielska Hybrid Organisations. A study of the Open Source – business setting
- 33. Jens Dick-Nielsen Three Essays on Corporate Bond Market Liquidity
- 34. Sabrina Speiermann
 Modstandens Politik
 Kampagnestyring i Velfærdsstaten.
 En diskussion af trafikkampagners styringspotentiale
- 35. Julie Uldam
 Fickle Commitment. Fostering political
 engagement in 'the flighty world of
 online activism'

- 36. Annegrete Juul Nielsen
 Traveling technologies and
 transformations in health care
- 37. Athur Mühlen-Schulte
 Organising Development
 Power and Organisational Reform in
 the United Nations Development
 Programme
- 38. Louise Rygaard Jonas

 Branding på butiksgulvet

 Et case-studie af kultur- og identitetsarbejdet i Kvickly

- Stefan Fraenkel
 Key Success Factors for Sales Force
 Readiness during New Product Launch
 A Study of Product Launches in the
 Swedish Pharmaceutical Industry
- Christian Plesner Rossing
 International Transfer Pricing in Theory
 and Practice
- 4. Kim Pettersson
 Essays on Audit Quality, Auditor Choice, and Equity Valuation
- 5. Henrik Merkelsen The expert-lay controversy in risk research and management. Effects of institutional distances. Studies of risk definitions, perceptions, management and communication
- 6. Simon S. Torp
 Employee Stock Ownership:
 Effect on Strategic Management and
 Performance
- 7. Mie Harder Internal Antecedents of Management Innovation

- 8. Ole Helby Petersen
 Public-Private Partnerships: Policy and
 Regulation With Comparative and
 Multi-level Case Studies from Denmark
 and Ireland
- Morten Krogh Petersen 'Good' Outcomes. Handling Multiplicity in Government Communication
- 10. Kristian Tangsgaard Hvelplund Allocation of cognitive resources in translation - an eye-tracking and keylogging study
- 11. Moshe Yonatany
 The Internationalization Process of
 Digital Service Providers
- 12. Anne Vestergaard
 Distance and Suffering
 Humanitarian Discourse in the age of
 Mediatization
- 13. Thorsten Mikkelsen Personligsheds indflydelse på forretningsrelationer
- Jane Thostrup Jagd
 Hvorfor fortsætter fusionsbølgen udover "the tipping point"?
 – en empirisk analyse af information og kognitioner om fusioner
- 15. Gregory Gimpel
 Value-driven Adoption and Consumption of Technology: Understanding
 Technology Decision Making
- Thomas Stengade Sønderskov
 Den nye mulighed Social innovation i en forretningsmæssig kontekst
- 17. Jeppe Christoffersen Donor supported strategic alliances in developing countries
- 18. Vibeke Vad Baunsgaard

 Dominant Ideological Modes of
 Rationality: Cross functional

- integration in the process of product innovation
- 19. Throstur Olaf Sigurjonsson Governance Failure and Icelands's Financial Collapse
- 20. Allan Sall Tang Andersen
 Essays on the modeling of risks in
 interest-rate and inflation markets
- 21. Heidi Tscherning

 Mobile Devices in Social Contexts
- 22. Birgitte Gorm Hansen
 Adapting in the Knowledge Economy
 Lateral Strategies for Scientists and
 Those Who Study Them
- 23. Kristina Vaarst Andersen
 Optimal Levels of Embeddedness
 The Contingent Value of Networked
 Collaboration
- 24. Justine Grønbæk Pors Noisy Management A History of Danish School Governing from 1970-2010
- 25. Stefan Linder
 Micro-foundations of Strategic
 Entrepreneurship
 Essays on Autonomous Strategic Action 4.
- 26. Xin Li
 Toward an Integrative Framework of
 National Competitiveness
 An application to China
- 27. Rune Thorbjørn Clausen

 Værdifuld arkitektur

 Et eksplorativt studie af bygningers

 rolle i virksomheders værdiskabelse
- 28. Monica Viken Markedsundersøkelser som bevis i varemerke- og markedsføringsrett
- 29. Christian Wymann
 Tattooing
 The Economic and Artistic Constitution
 of a Social Phenomenon

- 30. Sanne Frandsen
 Productive Incoherence
 A Case Study of Branding and
 Identity Struggles in a Low-Prestige
 Organization
- 31. Mads Stenbo Nielsen
 Essays on Correlation Modelling
- 32. Ivan Häuser
 Følelse og sprog
 Etablering af en ekspressiv kategori,
 eksemplificeret på russisk
- 33. Sebastian Schwenen
 Security of Supply in Electricity Markets

- Peter Holm Andreasen
 The Dynamics of Procurement
 Management
 A Complexity Approach
- Martin Haulrich
 Data-Driven Bitext Dependency
 Parsing and Alignment
- 3. Line Kirkegaard Konsulenten i den anden nat En undersøgelse af det intense arbejdsliv
- 4. Tonny Stenheim

 Decision usefulness of goodwill

 under IFRS
- 5. Morten Lind Larsen Produktivitet, vækst og velfærd Industrirådet og efterkrigstidens Danmark 1945 - 1958
- 6. Petter Berg
 Cartel Damages and Cost Asymmetries
- 7. Lynn Kahle
 Experiential Discourse in Marketing
 A methodical inquiry into practice
 and theory
- 8. Anne Roelsgaard Obling

 Management of Emotions
 in Accelerated Medical Relationships

TITLER I ATV PH.D.-SERIEN

1992

 Niels Kornum
 Servicesamkørsel – organisation, økonomi og planlægningsmetode

1995

Verner Worm
 Nordiske virksomheder i Kina
 Kulturspecifikke interaktionsrelationer
 ved nordiske virksomhedsetableringer i
 Kina

1999

3. Mogens Bjerre
Key Account Management of Complex
Strategic Relationships
An Empirical Study of the Fast Moving
Consumer Goods Industry

2000

Lotte Darsø
 Innovation in the Making
 Interaction Research with heterogeneous Groups of Knowledge Workers creating new Knowledge and new Leads

2001

5. Peter Hobolt Jensen

Managing Strategic Design Identities
The case of the Lego Developer Network

2002

- 6. Peter Lohmann
 The Deleuzian Other of Organizational
 Change Moving Perspectives of the
 Human
- 7. Anne Marie Jess Hansen
 To lead from a distance: The dynamic
 interplay between strategy and strategizing A case study of the strategic
 management process

2003

- Lotte Henriksen
 Videndeling
 om organisatoriske og ledelsesmæssige udfordringer ved videndeling i
 praksis
- 9. Niels Christian Nickelsen
 Arrangements of Knowing: Coordinating Procedures Tools and Bodies in
 Industrial Production a case study of
 the collective making of new products

2005

10. Carsten Ørts Hansen
Konstruktion af ledelsesteknologier og
effektivitet

TITLER I DBA PH.D.-SERIEN

2007

Peter Kastrup-Misir
 Endeavoring to Understand Market
 Orientation – and the concomitant
 co-mutation of the researched, the
 re searcher, the research itself and the
 truth

2009

- Torkild Leo Thellefsen
 Fundamental Signs and Significance
 effects
 A Semeiotic outline of Fundamental
 Signs, Significance-effects, Knowledge
 Profiling and their use in Knowledge
 Organization and Branding
- Daniel Ronzani
 When Bits Learn to Walk Don't Make
 Them Trip. Technological Innovation
 and the Role of Regulation by Law
 in Information Systems Research: the
 Case of Radio Frequency Identification
 (RFID)

2010

Alexander Carnera
 Magten over livet og livet som magt
 Studier i den biopolitiske ambivalens